

Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter

Document Version: 2012-06-29



This guide is a primer for adding remotely accessible PC or servers into your NETLAB+ equipment pods using the <u>VMware</u> ESXi and vCenter virtualization products.

This guide covers features available in NETLAB+ version **2011.R2** and later. The details of this guide are specific to **VMware ESXi versions 4.01** and **4.1 U2 with vCenter.**

Documentation for interfacing with other versions of VMware virtualization products can be found in their respective *Remote PC Guide for VMware Implementation* guides.

Copyright © Network Development Group, Inc. www.netdevgroup.com

NETLAB Academy Edition, NETLAB Professional Edition, and NETLAB+ are registered trademarks of Network Development Group, Inc.

VMware is a registered trademark of VMware, Inc. Cisco, IOS, Cisco IOS, Networking Academy, CCNA, and CNP are registered trademarks of Cisco Systems, Inc.



1	Background	6
	1.1 What should I know before proceeding?	6
	1.2 What is a Remote PC?	7
	1.3 What can users do with a remote PC?	8
	1.4 What is a Virtual Machine?	10
	1.5 How do NETLAB+, VMware vCenter and VMware ESXi work together?	11
2	Planning	14
	2.1 VMware Product Comparison	14
	2.2 NETLAB+ Feature Support	16
	2.3 Virtual Machine Software Licenses	16
	2.4 Obtaining VMware vSphere Software and Licenses for the NETLAB+	
	Infrastructure	17
	2.5 VMware ESXi	17
	2.5.1 VMware ESXi Host Requirements	18
	2.5.2 Special ESXi Host Requirements for VMware IT Academy	20
	2.5.3 Obtaining VMware ESXi and Licenses	20
	2.6 VMware vCenter	23
	2.6.1 VMware vCenter Server Requirements	23
	2.6.2 Obtaining VMware vCenter Server and Licenses	25
	2.7 Networking Models	30
	2.7.1 Single-Homed Networking	31
	2.7.1.1 Single-Homed Setup Tasks	32
	2.7.2 Dual-Homed Networking	33
	2.7.2.1 Dual-Homed Setup Tasks	34
	2.7.3 Secure+ Networking	35
	2.7.3.1 Secure+ Networking Setup Tasks	36
	2.8 Storage Area Networks	37
3	VMware ESXi Server Setup	38
•	3.1 Preparing the ESXi Server	39
	3.1.1 DELL R710 BIOS System changes	39
	3.2 DELL R710 RAID Configuration	43
	3.2.1 3x2TB HDD Configuration	44
	3.2.2 3x1TB HDD Configuration	46
	3.3 Installing FSXi on Host Server	48
	3.4 Configure Root Password	49
	3.5 Network Configuration	50
4	VMware vCenter Server Setup	56
•	4.1 vCenter Configuration Options	56
	4.1.1 vCenter Configuration Option 1	57
	4.1.2 vCenter Configuration Option 2	58
	4.1.3 vCenter Configuration Option 3	59
	4.1.4 vCenter Configuration Option 4	60
	4.2 Networking Overview for vCenter Server	61
	4.2.1 Virtualized vCenter Server Networking Ontions	61
	4.2.1.1 Virtualized vCenter Server with Single Homed Networking	61
		01



	4.2.1.2 Virtualized vCenter Server with Dual Homed Networking	62
	4.2.1.3 Virtualized vCenter with Secure+ Networking	63
	4.2.2 Bare Metal vCenter Server Networking Options	65
	4.2.2.1 Bare Metal vCenter Server with Single Homed Networking	65
	4.2.2.2 Bare Metal vCenter Server with Dual Homed Networking	66
	4.2.2.3 Bare Metal vCenter Server with Secure+ Networking	67
	4.3 Configure Management Server	68
	4.4 Install Windows Server 2008 R2 64-bit	68
	4.5 Configuring TCP/IP on vCenter Server Network Adapters	69
	4.5.1 Outside Interface	69
	4.5.2 Inside Interface	71
	4.6 Installing VMware vCenter and Related Software	73
	4.6.1 Installing vCenter with Microsoft SQL Server 2008 R2 (Option 1 a	nd 3) 73
	4.6.1.1 Configure Hostname and Create User Account	74
	4.6.1.2 Install Microsoft SQL Server 2008 R2	78
	4.6.1.3 Create vCenter Database and ODBC drivers	83
	4.6.1.4 Install vCenter with SQL Server 2008 R2 database	88
	4.6.2 Installing vCenter with Microsoft SQL Express (Option 2 and 4)	
	4.7 Install the vSphere Client on the vCenter Server System	
	4.8 Install vCenter Converter	
	4.8.1 Install and Enable the vCenter Converter Plug-in	
	4.9 Creating a Virtual Datacenter in vCenter	
	4.10 Create Windows User Account and vCenter Role for NETLAB+	
	4.11 Registering a Virtual Datacenter in NETLAB+	105
	4.12 Setting the Database Retention Policy	106
5	Adding ESXi Hosts to vCenter and NETLAB+	108
	5.1 Adding ESXi hosts to vCenter	108
	5.2 ESXi Host Virtual Switches	110
	5.3 Verifying vSwitch0 Configuration	111
	5.4 Inside Network Configuration	112
	5.4.1 Creating vSwitch1 and Binding to Physical NIC	115
	5.4.2 Configuring Control Switch 802.1q Trunk Ports	120
	5.4.3 Connecting Virtual Machines to Real Equipment Pods	122
	5.4.3.1 Creating a Real Equipment Pod	123
	5.4.3.2 Determining the Base VLAN and VLAN Pool	123
	5.4.3.3 Creating Port Groups for Pod VLANs on the Inside Network	123
	5.4.3.4 Increasing the Inside vSwitch Port Count	124
	5.5 Creating a Safe Staging Network	127
	5.6 Adding ESXi hosts in NETLAB+	130
	5.7 Proactive Resource Awareness	132
6	vCenter Update Manager	134
	6.1 Installing vCenter Update Manager with Microsoft SQL Server 2008 R	2 (Options
	1 and 3)	135
	6.2 Installing vCenter Update Manager with Microsoft SQL Server Express	; (Options
	2 and 4)	143



	6.2	loct	all the vCenter Undete Manager plug in	1 1 1
	6.3	Inst	an the vCenter Opdate Manager plug-in	144
_	6.4	Peri	forming updates using the vCenter Update Manager plug-in	146
7	Bui	ding	s Virtual Machines	151
	7.1	Usir	ng NDG Template Virtual Machines and 3rd Party Virtual Appliances	153
	7.2	Crea	ating Virtual Machines from Scratch	155
	7.2.	1	Providing a Name for Your Virtual Machine	156
	7.2.	2	Selecting a Datastore	156
	7.2.	3	Select the Virtual Machine Hardware Version	157
	7.2.	4	Selecting the Guest Operating System	157
	7.2.	5	Selecting the Number of Processors	158
	7.2.	6	Configuring the Memory Size	159
	7.2.	7	Choosing Network Connections	160
	7.2	8	Selecting the Disk Controller	162
	7.2	9	Creating a Virtual Hard Disk	163
	72	10	Specifying Advanced Ontions	164
	7.2	11	Verifying the Settings	165
	73	Inst	alling a Guest Operating System	166
	74	Fdit	ing the Virtual CD/DVD Device	166
	75	Fsse	ential Virtual Machine Performance Ontimizations	168
	75	1	Installing VMware Tools	168
	75	2	Disabling the Deskton Background	170
	7.5	2	Setting the Virtual Machine Display Properties	170
	7.5	<u>л</u>	Adjusting Visual Effocts	171 172
	7.5	4 • 4 d	Aujusting visual Effects	172
	7.0 7.7	Virt	ual Machine Spanchetc	173
	/./ 77	1	How NETLAR - Lloss Spanshots	175
	/./. 	1 2	How NETLAD+ Uses Shapshots	175
	7.7.	2	Tables a New Consider	174
	/./.	3	Taking a New Snapshot	175
_	/./.	4	Managing Snapshots	176
8	NET	LAB	+ Virtual Machine Inventory	177
	8.1	Virt	ual Machine Roles	177
	8.2	Hov	v Virtual Machines Become Part of the NETLAB+ Inventory	178
	8.3	Imp	orting VMs into the Virtual Machine Inventory	179
	8.4	Virt	ual Machine Cloning	181
	8.4	1	Golden Masters and Golden Snapshots	182
	8.4	2	Using NETLAB+ to Clone a Single Virtual Machine	182
	8.5	Assi	gning Virtual Machines to Pods	186
9	Clo	ning	Virtual Machine Pods	193
	9.1	Gol	den Masters and Golden Snapshots	195
	9.2	Crea	ating a Master Pod	195
	9.3	Clor	ning a Virtual Machine Pod Using Linked Clones	196
	9.4	Tasl	ks to Perform After Pod Cloning	199
	9.5	Savi	ing Time on Subsequent Pod Cloning	199
	9.6	Crea	ating a Full Clone of a Virtual Machine Equipment Pod	200
	9.7	Crea	ating Pods that Run on a Multiple VMware ESXi Hosts	200



10	/irtual Machine Operations	03
10	1 Delete All Virtual Machines in a Pod2	04
10	2 Deleting Individual Virtual Machines 2	05
10	Changing the Name of a Virtual Machine 2	07
10	Changing the Role of a Virtual Machine 2	08
10	5 Migrating a Virtual Machine to a Different ESXi Host	09
11	Ising NDG Automated Pods 2	11
11	1 NDG Virtual Machine Topologies 2	11
11	2 NDG Real Equipment Topologies2	12
11	Setting the Local System ID When Using Multiple NETLAB+ Systems	13
Appe	ndix A Manual Network Setup and Troubleshooting	14
Ар	endix A.1 Creating Inside VLANs that Connect to Real Equipment	14
	ppendix A.1.1 Creating a Real Equipment Pod	15
	ppendix A.1.2 Determining the Base VLAN and VLAN Pool	16
	ppendix A.1.3 Creating Port Groups for Pod VLANs on the Inside Network 2	18
Ар	endix A.2 Verifying Connectivity Between Virtual Machines and Lab Gear 2	21





1 Background

NETLAB+ remote PCs and servers in a pod can be implemented using virtual machines running on VMware vSphere 4. The flexibility and broad selection of operating systems and configurations that may be provisioned on a virtual machine offer great potential to support IT training in a wide range of disciplines using NETLAB+.

This guide provides information on NETLAB+ remote PC VMware implementation, using ESXi versions 4.01 and 4.1 U2 with vCenter.

In the subsections below, we will begin by bringing to your attention the prerequisite knowledge recommended, along with building a fundamental understanding of how remote PCs, virtualization and NETLAB+ work together.

Objectives

- What should I know before proceeding?
- What is a remote PC?
- What can users do with a remote PC?
- What is a virtual machine?
- How do NETLAB+, VMware vCenter and VMware ESXi work together?

1.1 What should I know before proceeding?

NETLAB+ communicates with VMware vSphere to perform automated tasks and virtual machine management. Users of this guide should have a working knowledge of VMware vCenter, the vSphere Client and VMware ESXi.

NETLAB+ administrators using this guide should be comfortable with the process of creating and configuring a virtual machine "from scratch" (using the vSphere Client).

The <u>VMware vSphere Virtual Machine Administration Guide</u> provides detailed guidance on provisioning virtual machines.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

1.2 What is a Remote PC?

A *remote PC* is a personal computer or server that can be remotely accessed from another desktop. *Remote access* allows a user to have full access to the keyboard, video, and mouse of the remote PC. NETLAB+ provides built-in client software for remote access, which is loaded automatically via the user's web browser.





1.3 What can users do with a remote PC?

Users can remotely access the keyboard, video, and mouse of a remote PC. NETLAB+ also provides special features such as shared simultaneous access, interfacing with real lab equipment (routers, switches, and firewalls), remotely powering a PC on or off, and restoring the PC to a clean state. This offers a wide range of possibilities. Here are a few scenarios that are being used today.

- **Online Lab Delivery.** Provide students with self-paced, scheduled access to real operating systems and application software, without distributing software or licenses.
- **Distance Learning.** Provide remote instructor-led training by allowing simultaneous shared access to remote PCs and remote servers. Several users can connect to and share the remote PC's graphical user interface at the same time. Using NETLAB+, students can observe what the instructor is doing on the remote PC, and vice-versa.
- **Resource Scheduling.** Provide scheduled usage to limited physical lab equipment and virtual machine host servers. Proactive Resource Awareness allows you to timeshare virtualization servers using the NETLAB+ scheduler.
- **Online Network Training.** Provides online delivery of network training. Remote PCs can be interface with real lab equipment, such as routers, switches, and firewalls, all of which can be accessed remotely using NETLAB+.
- Online General IT Training. Provide on-line access to real operating systems and real application software. Using NETLAB+, remote PCs can be completely isolated from production networks, providing a safe environment for instructors and students to do things that are not typically allowed on production networks. Students can safely experience administrative privileges in complex computing environments. You can now provide labs that are not practical for students to set up at home, or scenarios that would be too difficult to set up by new IT students. NETLAB+ includes 25 virtual topologies that can be used to teach a variety of courses, including Linux, Microsoft or Cyber Security. Pods using these topologies can be created very quickly using NETLAB+'s pod cloning and automated network features.
- Online Security Training. Provides online delivery of security training. Using NETLAB+, remote PCs can be completely isolated from production networks, providing a safe environment for instructors and students to do things that are not typically allowed on production networks. This might include configuring PCs and lab devices using administrator privileges, installing new software, capturing network traffic, experimenting with firewalls and VPNs, dealing with live viruses



and malware, and scanning networks. At the end of the lab reservation, NETLAB+ will undo any changes.

• VMware vSphere ICM Course. The VMware vSphere ICM course prepares your students for the VMware Certified Professional exam. NDG has partnered with VMware to prepare a series of labs for the NETLAB+ environment. Using the virtualization and pod assignment capabilities of NETLAB+, each student has access to their own set of virtual equipment, which they may maintain exclusive use of throughout the course. Student pods can be created very quickly using NETLAB+'s pod cloning and automated network features. NETLAB+'s use of virtualized lab components results in a significant cost reduction by allowing several pods to run simultaneously on one physical server. For more information, please visit <u>https://www.vmware.com/partners/programs/vap/</u>.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

1.4 What is a Virtual Machine?

In NETLAB+, a *virtual machine* is a remote PC or remote server that runs on virtualized hardware. Although the hardware is virtualized, real operating systems and real application software can still be used; virtual hardware appears to be real as far as the software is concerned. In fact, the software running on a virtual machine is allowed to execute instructions directly on the real CPU. This provides relatively good performance, comparable to actual hardware in most cases. A special process known as the *hypervisor* manages workload among virtual machines (VMs) to ensure that each application has time to execute.



Virtualization allows you to host real operating systems and real application software with fewer hardware resources. To implement VMs, the NETLAB+ software interfaces with third party virtualization products that run on separate servers (not on the NETLAB+ server). This guide is specific to ESXi 4.01 and 4.1 U2 with vCenter, from VMware Inc.

Each NETLAB+ remote PC or remote server runs inside of a virtual machine. VMware ESXi provides virtual CPU, virtual memory, virtual disk drives, virtual networking interface cards, and other virtual hardware for each virtual machine. ESXi also provides the concept of a virtual networking switch. Virtual switches can be connected to real networks via host network adapters, allowing VMs to connect to real networks.



1.5 How do NETLAB+, VMware vCenter and VMware ESXi work together?



The following diagram depicts four major components that make up a typical NETLAB+ system setup using VMware vCenter and VMware ESXi 4.x.

- 1. The NETLAB+ server provides the user interface for student and instructor access, an interface to manage VMs, and software features to automate virtual machine pod creation. This document assumes you have already setup your NETLAB+ server.
- 2. VMware vCenter is used to manage your physical VMware ESXi servers, to create VMs, and to take snapshots of virtual machines. NETLAB+ communicates with vCenter to perform automated tasks and virtual machine management.
- 3. Physical VMware ESXi servers host the virtual machines in your virtual machine pods. In the example environment shown here, there are two host servers. Each NETLAB+ remote PC or remote server runs inside of a virtual machine
- 4. Pods consisting of virtual machines reside on your physical ESXi host server disks. Optionally, these VMs can reside on a Storage Area Network (SAN).

Virtualization using ESXi is performed on separate physical servers, not included with NETLAB+. You can interface with multiple ESXi servers if necessary.

Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

Here is list of features and benefits provided by NETLAB+, working in conjunction with VMware vCenter and VMware ESXi.

- **Remote Access.** The keyboard, video and mouse of each virtual machine can be accessed without a "backdoor" network or interface on the virtual machine. Access to a virtual machine is proxied through NETLAB+ and the virtualization host system, similar to KVM-over-IP hardware solutions. No special client software (other than Java) is required on the user's computer. NETLAB+ will download its remote PC access application to the client whenever the user clicks on a PC.
- **Sharing.** Multiple users can share access to a virtual machine simultaneously.
- **Connection Proxy.** NETLAB+ *multiplexes* virtual machine traffic using a single IP address and two TCP ports. It also provides a front-end to the virtual machine environment, so that virtualization servers and VMs do not have to be placed on production networks. This significantly increases security and eases firewall administration. If the user has a valid lab reservation, NETLAB+ will proxy client access to the keyboard, video and mouse of the virtual machine. This access is terminated when the lab reservation completes, ensuring that users of different reservations do not interfere with each other.
- Automated Operations. Users may power on, power off, and revert to clean state (scrub) from the NETLAB+ web interface.
- **Snapshots.** NETLAB+ supports *revert to snapshot*. Changes to a virtual machine can be discarded at the end of a lab reservation, returning the PC to a clean state.
- **Progressive Labs.** Using NETLAB+ Pod Assigner, students may be assigned their own pod and personal VMs for an entire course. These VMs will retain their state between lab reservations.
- Linked Virtual Machines. NETLAB+ cloning operations support linked virtual machines. A linked virtual machine shares virtual disks with a master virtual machine in an ongoing manner. This conserves disk space and allows multiple virtual machines to use the same software installation. Linked virtual machines can be created very quickly because most of the disk is shared with the master VM.
- **Pod Cloning.** Pods containing only virtual machines can be cloned in a single operation. This is called pod cloning. The NETLAB+ administrator can clone a master pod very quickly. Using linked virtual machines, a typical pod can be cloned in one minute or less.
- Automatic Networking. Many NDG provided pods templates support automatic networking. When a pod starts, NETLAB+ will create all the necessary virtual

switches and/or port groups on the VM's host server, and bind each virtual machine network adapter to the correct port group. At the end of the lab reservation, NETLAB+ will delete the virtual switches and port groups used by the pod to free networking resources.

- Automatic Remote Display Setup. NETLAB+ now supports automatic setup of remote display parameters on virtual machines. This feature combined with pod cloning and automatic networking allows pods to be deployed very quickly from a set of master VMs without any manual setup.
- **Proactive Resource Awareness.** Proactive Resource Awareness allows you to time-share virtualization servers using the NETLAB+ scheduler. If scheduling a particular pod would exceed the virtualization host server limits in a 30-minute time slot, the pod cannot be scheduled at that time and will be clearly indicated on the scheduler. This feature allows you to increase trainees with fewer servers while providing a good lab experience.
- VM Deletion. Virtual machines can be removed from the inventory and/or completely from the disk directly from NETLAB+. When deleting a pod, NETLAB+ provides the option to delete all virtual machines in one operation.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.ne

2 Planning

2.1 VMware Product Comparison

The following table compares NETLAB+ support for selected VMware hosting products. This guide is specific to **VMware vSphere ESXi with vCenter.**

Do not use VMware vSphere 5. NETLAB+ does not support vSphere 5 at this time due to several <u>known issues</u> (regarding ESXi and vCenter). See the <u>NETLAB+ Remote PC</u> <u>Support</u> page for the most current information. The diagram below illustrates the supported options.



- VMware ESXi 4.1 U2 is currently recommended for **physical** host servers.
- VMware vCenter 4.1 is currently recommended for NETLAB+ VM management.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

		THIS		
VMware Product	VMware ESXi vCenter	VMware ESXi vCenter	VMware ESXi vCenter	VMware ESXi Standalone
VMware Major Version(s)	5.0	4.1 U2	4.01	4.01
NETLAB+ Support	Not Supported	Recommended	Supported	Supported
Minimum NETLAB+ Version		2011.R2	2011.R2	2009.R1
Architecture		Bare Metal Hypervisor	Bare Metal Hypervisor	Bare Metal Hypervisor
VMware Versions Tested by NDG		4.1 U2	4.01	4.01
Host Operating System Required		No	No	No
vCenter Required		Yes	Yes	No
Support ICM 4.1 Pods		Yes	No	No
Support ICM 5.0 Pods		Yes	No	No

The documentation and screen-shot examples in this guide illustrate the use of ESXi 4.1 U2 for host servers. NETLAB+ system software will work with ESXi 4.01 servers, but be aware that if you plan to use your NETLAB+ system to teach the VMware IT Academy Program Install, Configure, Manage (ICM) 4.1 or 5.0 course, ESXi 4.1 U2 is required for the ICM course and ICM pod deployment.

If your system is currently using ESXi 4.01, upgrading to ESXi 4.1 U2 is recommended.



2.2 NETLAB+ Feature Support

The following table compares NETLAB+ features that are supported with various VMware virtualization products.

	THIS GUIDE	
NETLAB+ Features	VMware ESXi vCenter	VMware ESXi Standalone
VMware Major Version(s)	4.01 4.1 U2	4.01
Remote PC Viewer	Yes	Yes
Power On / Off VM	Yes	Yes
Revert to Snapshot	Yes	Yes
Integration with VMware vCenter	Yes	No
VM Inventory	Yes	No
Linked Clones	Yes	No
Clone VM Pods and Individual VMs	Yes	No
Automatic Networking (1)	Yes	No
Automatic Remote Display Setup (2)	Yes	No
Delete Pod VMs and Individual VMs	Yes	No
VMware vSphere ICM Course Support (3)	Yes	No

(1) NETLAB+ will automatically setup networking on NDG standard pods.

(2) NETLAB+ will automatically program Remote Display parameters.

(3) VMware vSphere ICM course requires ESXi 4.1 U2 and vCenter 4.1.

2.3 Virtual Machine Software Licenses

For the purpose of software licensing, each virtual machine is treated as an individual PC or server. Please refer to the specific vendor license agreements (and educational discount programs, if applicable) to determine licensing requirements for your virtual machines' software, server software, operating systems, and applications.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

2.4 Obtaining VMware vSphere Software and Licenses for the NETLAB+ Infrastructure

The <u>VMware Academic Program</u> enables member organizations worldwide to gain easy access to cutting-edge virtualization technology and resources at no charge. Eligible faculty within educational institutions may easily access and download selected VMware software at no charge.

For further information, please visit: <u>http://www.vmware.com/partners/programs/education/e-academy.html</u>

Renewable licenses for your physical ESXi Host servers and vCenter Server are available from the VMware e-academy.

Server Software		Source
ESXi Hosts	VMware vSphere ESXi Server	VMware e-academy website
	VMware vCenter Server Standard	VMware e-academy website
vCenter Server	Windows Server 2008 R2 64-bit	MSDN-AA or Retail
	Microsoft SQL Server 2008 R2 64-bit	MSDN-AA or Retail

2.5 VMware ESXi

Physical VMware ESXi servers host the virtual machines in your pods. Virtualization using ESXi is performed on separate physical servers, not included with NETLAB+. You can interface with multiple ESXi servers if necessary.

NETLAB+ is compatible with VMware ESXi versions 4.01 and 4.1 U2. For new installations, version 4.1 U2 is recommended. The hardware you use for your ESXi server(s) must be compatible with the version of ESXi you select.

VMware ESXi 4.1 U2 is a required component if you plan to use your NETLAB+ system to teach the VMware IT Academy Program Install, Configure, Manage (ICM) course. For more details on NETLAB+ support of the ICM course: http://www.netdevgroup.com/content/vmita/



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

2.5.1 VMware ESXi Host Requirements

The following table shows the specifications for the VMware host machine used by NDG as the 2012 test platform. We recommend using these specifications as a reference when planning your own system configuration. Your specific CPU, memory, and disk requirements will vary depending on the number of active virtual machines and their respective configurations.

Components	Recommended Minimum / Features
Server Model	Dell R710
Operating System	Specify NO operating system on order.
Hypervisor (installed by you)	VMware ESXi 4.1 U2
Physical CPUs	Two (2) x Intel Xeon E5620 Quad Core @ 2.4GHz ^{1,2}
Hardware Assisted Virtualization Support	Intel-VT and Intel-EPT ³
Total CPU Cores/Threads	8 cores, 16 threads
Total System Memory	Memory requirements vary based on server role or curriculum. Management Server: 32GB or higher recommended. Cisco NetAcad only: 64GB or higher recommended. VMware ICM 4.1 course: 72GB minimum. VMware ICM 5.0 course: 128GB minimum. Use 16GB quad-ranked DIMMs for maximum expansion. (Kingston part number KTD-PE310Q/16G)
Chassis Hard Drive Configuration	6 x 3.5"
Storage Configuration Options	 1.5TB Internal Direct Attached Storage Option⁴ H700 RAID Controller, 512MB Cache RAID 5 3 X 1TB, 3.5 SATA, 7200 RPM 3.0GB/s Western Digital RE4 WD1003FBYX Recommended 1 VMware VMFS Datastore 3.5TB Internal Direct Attached Storage Option ⁴ H700 RAID Controller, 512MB Cache RAID 5 3 X 2TB, 3.5 SATA, 7200RPM, 3.0GB/s Western Digital RE4 WD2003FYYS Recommended 2 VMware VMFS Datastores Dell PERC H700 Controllers with 512MB cache are recommended. Dell PERC H200 Controllers are NOT recommended.

(Table continued on next page)



(Table continued from previous page)

Components	Recommended Minimum / Features
NIC	Dual Two-Port Embedded Broadcom NetXtreme II 5709 Gigabit Ethernet
Riser Card	Riser with 2 PCIe x8 + 2 PCIe x4 Slot
BIOS Setting	Performance BIOS Setting
Power supplies	High Output Power Supply, Redundant, 870W
Embedded Management	DRAC6 Express
Optical Drive	DVD ROM, SATA, Internal

¹Minimum recommended processor is Intel E5620 @ 2.4Ghz (4 cores, 8 threads per CPU). VMware ICM pods have not been tested on AMD based systems by NDG and are not supported.

²Two (2) physical CPUs per server (i.e. dual-socket) are required for VMware ICM pod support (8 cores, 16 threads per host).

³64-bit processors with hardware-assisted virtualization (Intel-VT/EPT) are required for good virtual machine performance and to support VMware ICM course offerings.

⁴Internal Direct Attached Storage consists of hard drives that reside on the ESXi server and are connected directly to the host system via a RAID controller.

Please search the VMware Compatibility guide to ensure your ESXi host hardware is compatible with the VMware version you wish to use.

http://www.vmware.com/resources/compatibility/search.php

NDG Equipment Selection Disclaimer

NDG offers no warranties (expressed or implied) or performance guarantees (current or future) for 3rd party products, including those products NDG recommends. Due to the dynamic nature of the IT industry, our recommended specifications are subject to change at any time.

NDG recommended equipment specifications are based on actual testing performed by NDG. To achieve comparable compatibility and performance, we strongly encourage you to utilize the same equipment, exactly as specified and configure the equipment as directed in our setup documentation. Choosing other hardware with similar specifications may or may not result in the same compatibility and performance. The customer is responsible for compatibility testing and performance validation of any hardware that deviates from NDG recommendations. NDG has no obligation to provide support for any hardware that deviates from our recommendations, or for configurations that deviate from our standard setup documentation.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter

2.5.2 Special ESXi Host Requirements for VMware IT Academy

Hardware Assisted Virtualization (Intel VT-x) is **REQUIRED** on any host you use for the VMware IT Academy Install, Configure, Manage (ICM) course.

The VMware IT Academy labs have not been tested on AMD processors and are not supported on AMD processors.

The *NETLAB+ VMware ICM Pod Installation and Configuration Guide* provides detailed instructions for delivering the ICM course using your NETLAB+ system. You may obtain this guide through the NDG Lab Resource Center for the VMware IT Academy:

http://www.netdevgroup.com/content/vmita/resource/

2.5.3 Obtaining VMware ESXi and Licenses

The following procedure assumes you are a registered member of the VMware Academic Program (VMAP). Non-members can obtain evaluation copies of VMware vCenter and ESXi software from http://www.vmware.com and purchase through retail partners.

When downloading ESXi it is important to select a version that is compatible with NETLAB+.

- NETLAB+ is compatible with VMware ESXi versions 4.01 and 4.1 U2.
- For new installations, version 4.1 U2 is recommended.
- VMware ESXi 4.1 U2 is a required component if you plan to use your NETLAB+ system to teach the VMware IT Academy Program Install, Configure, Manage (ICM) course.

Keep in mind that VMware ESXi version 5.0 is not supported. Do not select VMware ESXi 5.0.

The current link from VMware Academic Alliance Program is for VMware ESXi 4.1. You will need to upgrade to 4.1 U2 as recommended.

- 1. Follow the link provided to you by VMware when you registered in the VMware Academic Alliance Program. This will take you to the academic software store.
- 2. Click on **Faculty/Staff** at the top to see the available downloads. You must be a registered Faculty/Staff user. For more information, contact the VMware Academy contact at your school.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.

					Sign In	English	1
(١
	vm ware) .					
							1
	Tweet 0	📫 Like 📑 Be the firs	t of your friends to like this.				
	Students	Faculty/Staff					
	VMwaro						Ĺ
	vmware						

3. Click on **VMware vSphere ESXi Server**, which is marked with a red box in the picture below. This is the first of two software items you will download from the e-academy website (see section 2.6.2). These items are marked with "Yes" in the picture below.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

4. Click on Add to Cart.

/Mware vSphe	re ESXi Server - Download	
Vitiware	Manufacturer: VMware, Inc.	Free
VMware ESXi 4	Delivery Type: Download	Quantity: 1 restricted
	Available to: Faculty/Staff	🛒 Add To Cart
	In Stock	<u>Are you eligible?</u>
Tweet 0	Recommend I Sign IIn to see what your friends recommend	<u>Are you eligible?</u>

5. Sign in with your registered login.

6. Click on **Check Out** to continue.

Your Cart			
Name VMware vSphere ESXi Server - Download Date Added:	Quantity	Unit Price Free	Price Free Remove
	Update Cart	Subtotal:	\$0.00
	* (heck Out	👚 Continue Shopping

- 7. Read and accept the **EULA**.
- 8. On the confirmation page, click on **Proceed With Order.**



9. On the receipt page, record the serial number found under **Items**. You will need this serial number for the installation later.



There is a limit of one download of ESXi Server per user account. If your system will be running more than 2 Dell R710 ESXi Host Servers (see section 3 for discussion), you will need one or more additional licenses. In order to circumvent the limitation, you may use additional registered faculty/staff accounts to obtain additional licensed downloads.

2.6 VMware vCenter

VMware vCenter Server enables you to manage the resources of multiple ESXi hosts and allows you to monitor and manage your physical and virtual infrastructure. Starting with software version 2011.R2, NETLAB+ integrates with VMware vCenter Server to assist the administrator with installing, replicating and configuring virtual machine pods.

2.6.1 VMware vCenter Server Requirements

A separate server running a **64-bit** Windows Server operating system is required for vCenter Server Standard. This server can be a physical server (bare metal) or virtual machine running on a VMware ESXi 4.1 U2 host. In either case, the physical server on which vCenter resides should be a dedicated "management server" to provide ample compute power.

NDG does not support configurations where vCenter is running on a heavily loaded ESXi host and/or an ESXi host that is also used to host virtual machines for NETLAB+ pods. Such configurations have exhibited poor performance, API timeouts, and sporadic errors in NETLAB+ operations.



This table outlines several configurations supported by VMware and recommended for NETLAB+.

Option	Host	Operating System	vCenter Version	Database	Host/VM Limit
1	VMware ESXi 4.1 U2	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Server	
2	VMware ESXi 4.1 U2	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Express (built-in database)	5 / 50
3	Bare Metal	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Server	
4	Bare Metal	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Express (built-in database)	5 / 50

Option 1 is recommended for any vSphere infrastructure of 50 or more virtual machines.

Option 2 is recommended for small vSphere 4.1 U2 deployments that will not exceed 50 virtual machines.

Options 3 and 4 are bare metal Windows 2008 server deployments and do not leverage virtualization. Only one instance of vCenter can be run on the server. Running vCenter in a virtual machine instance is recommended as it provides additional benefits:

- The vCenter VM can be backed up using vSphere backup utilities.
- The vCenter VM can be easily migrated to another host.
- The vCenter VM can failover to another host.
- You may stand up new versions of vCenter on the management server, in parallel with the production instance, and slowly upgrade components and VMs to newer versions of VMware vSphere as they become available.

The vCenter server must have network access to your ESXi servers and to the Internet for updates and patches. You will use the VMware vSphere Client to access vCenter Server.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

2.6.2 Obtaining VMware vCenter Server and Licenses

The following procedure assumes you are a registered member of the VMware Academic Program (VMAP). Non-members can obtain evaluation copies of VMware vCenter and ESXi software from http://www.vmware.com and purchase through retail partners.

The installer for vCenter Server is available through the VMware e-academy website. If you are installing vCenter Server on a physical machine, it is recommended you perform these steps on that computer to avoid large file transfers. If you are installing on a virtual machine, you will need to upload this file to the ESXi datastore.

- 1. Follow the link provided to you by VMware when you registered the academy. This will take you to the academic license software store.
- 2. Click on **Faculty/Staff** at the top to see the available downloads. You must be a registered Faculty/Staff user. For more information, contact your VMware academy contact at your school.

	Home Your Account Help Sign In
vm ware [.]	
Students Faculty/Staff	
VMware	
English (United States) I US Dollars Pro >Tweet I C Like I Be the first of your friends to like this.	oduct Search Go



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

10. Click on **VMware vCenter Server Standard,** which is marked with a red box in the picture below. This is the second of two software items you will download from the e-academy website (see section 2.6.2). These items are marked with "Yes" in the picture below.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

3. Click on Add To Cart.

		Но	me You	r Account Help Sigr
vm ware [.]				
Students Faculty,	Staff			
lish (United States) 📟 U	'S Dollars	Product Search		G
Mware vCenter Se	rver Standard - Downl	oad		
	Manufacturer: VMware	e, Inc.		
Vinware	Payment: PayPal, Express	Visa, MasterCard, American		Quantity: 1
	Delivery Type: Downloa	ad	Free	Who is Eligible?
	In Stock			Faculty/Staff (more) Quantity restricted
Tweet 0	ommend 📑 Be the first of your fri	ends to recommend this.		
ou will be able to place a he license you will receiv	n order for this product again in a with this offering is valid 12 r	n 12 months after the initial order. nonths starting with the 1st of the r	month the	offering was ordered.

4. Sign in with your registered login.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

5. Click on **Check Out** to continue.

Home Your Account	🦙 Shopping Cart	Help Sign Out
vm ware [.]		
Students Faculty/Staff		
English (United States) 🧮 US Dollars	Product S	Gearch
Your Cart		
Name <u>VMware vCenter Server Standard - Download</u> Date Added:	Quantity	Unit Price Price Free Free Remove
	Update Cart	Subtotal: \$0.00
·	4	Check Out Continue Shopping
Privacy Policy	Safe Shopping	

- 6. Read and accept the **EULA**.
- 7. On the confirmation page, click on **Proceed With Order.**
- 8. On the receipt page, record the serial number found under **Items**. You will need this serial number for installation later.

Items All prices are in US Dollars			
Name	Quantity	Unit Cost	Amount
1. VMware vCenter Server Standard - Download Download Options VMware EULA	1	\$0.00	\$0.00
Serial Number - 4M4066 D-1 -89U5M			
You will be able to place an order for this product again in 12 months after the initial order. The license you will receive with this offering is valid 12 months starting with the 1st of the month the offering was ordered.			
		Subtotal:	\$0.00
		Taxes: Total:	<u>\$0.00</u> \$0.00



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

9. Click on **Download Options** to get the link for downloads.

Items All prices are in US Dollars			
Name 1. VMware vCenter Server Standard - Download Download Options VMware EULA Serial Number - 4M4066 D89U5M You will be able to place an order for this product again in 12 months after the initial order. The license you will receive with this offering is valid 12 months starting with the 1st of	Quantity 1	Unit Cost / \$0.00	Amount \$0.00
		Subtotal: Taxes: Total:	\$0.00 <u>\$0.00</u> \$0.00

- 10. Click on VMware vCenter Server 4.1.0 to start the download. Save the file to your desktop for installation.
- 11. Sign out and close the e-academy website.





2.7 Networking Models

The NETLAB+ server, vCenter server and ESXi hosts may be *single-homed* or *dual-homed* depending on your requirements. This guide documents three common networking configurations:

- Single-Homed Networking
- Dual-Homed Networking
- Secure+ Networking

Please review each configuration to determine which one best suits your needs. Choose only one of these configurations, then perform the corresponding setup tasks .

The NETLAB+ server, vCenter server and ESXi hosts should be geographically co-located. If Single-Homed (outside) networking is used, all components should be connected by 100Mb/s or higher speed LAN. NDG does not support split server configurations that traverse a WAN and/or configurations that place firewalls between these servers.

The following table describes the symbols and connection types denoted in the networking diagrams throughout this section.

Symbol	Connection Type
нттр	HTTP Connection
РС	Remote PC Display User Connection
РСВ	Remote PC Display Back Connection
MGT	VMware vSphere Management Connection
REQ	Real Equipment Traffic on VLANs
RDP	Remote Desktop Protocol



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

2.7.1 Single-Homed Networking

Single-homed networking (SH) connects one NIC from each server to a routable network on your campus. SH can be used if your pods contain only virtual machines (i.e. you are not hosting only real lab equipment). All traffic flows across the campus LAN. Consider the Secure+ network model (described later) if you do not want the VMware infrastructure components or traffic on the campus LAN.





SH does not require NETLAB+ control switches. A 1Gb/second switch port is highly recommended for each server connection. This will provide optimal bandwidth for remote display connections and virtual machine cloning operations between hosts.

2.7.1.1 Single-Homed Setup Tasks

If you have chosen single-homed networking, the following setup tasks can be performed now.

- 1. Obtain an IP addresses, subnet mask, and default gateway for each outside server interface connected to your campus LAN.
- 2. If you are using a physical server for vCenter, connect it to a Gigabit Ethernet port on your campus LAN.
- 3. Connect the first Ethernet port of each ESXi host to a Gigabit Ethernet port on your campus LAN.

Later in the setup process, you will configure IP parameters for these interfaces. The following table shows key configurations parameters that will be implemented for this networking model.

	NETL	NETLAB+		vCenter		ESXi Host 1		ESXi Host 2	
	Outside	Inside	Outside	Inside	Outside	Inside	Outside	Inside	
IP Address	Campus		Campus		Campus		Campus		
Subnet Mask	Campus		Campus		Campus		Campus		
Gateway	Campus		Campus		Campus		Campus		
vSwitch					vSwitch0		vSwitch0		
Management Path					*		*		



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

2.7.2 Dual-Homed Networking

Dual-homed networking (DH) can be used if you are hosting both virtual machines and real lab equipment in your NETLAB+ pods. In this environment, VLANs on NETLAB+ control switches serve as an "inside" communication path for virtual machine traffic between your ESXi host servers and real lab equipment.





2.7.2.1 Dual-Homed Setup Tasks

If you have chosen dual-homed as your networking model, the following setup tasks can be performed now.

- 1. Obtain an IP addresses, subnet mask, and default gateway for each outside server interface connected to your campus LAN.
- 2. If you are using a physical server for vCenter:
 - a. Connect the first Ethernet interface to a port on your campus LAN.
 - b. Connect the second Ethernet Interface to a control switch reserved port; a Gigabit uplink port can be used if available.
- 3. Connect your ESXi host servers:
 - a. Connect the first Ethernet port of each ESXi host to a Gigabit Ethernet port on our campus LAN.
 - b. Connect the second Ethernet port on your ESXi to a designated reserved port on a control switch, or a Gigabit Ethernet uplink port (if one is available). A Gigabit port will provide the highest bandwidth.
- 4. Console into the control switch and configure each inside ESXi host switchport as a trunk (NETLAB+ does not do this automatically).

```
interface FastEthernet0/n
  description inside connection to ESXi host #
  switchport mode trunk
  switchport mode nonegotiate
```

Later in the setup process, you will configure virtual switches, VM kernel ports, and IP addresses for interfaces. The following table shows key configuration parameters that will be implemented for this networking model.

	NETLAB+		vCenter		ESXi Host 1		ESXi Host 2	
	Outside	Inside	Outside	Inside	Outside	Inside	Outside	Inside
IP Address	Campus	169.254.0.254	Campus	169.254.0.239	Campus	169.254.0.241	Campus	169.254.0.242
Subnet Mask	Campus	255.255.255.0	Campus	255.255.255.0	Campus	255.255.255.0	Campus	255.255.255.0
Gateway	Campus	not set	Campus	not set	Campus	not set	Campus	not set
vSwitch					vSwitch0	vSwitch1	vSwitch0	vSwitch1
Management Path					*		*	



2.7.3 Secure+ Networking

This model focuses on security by moving all virtual machine traffic and management to the inside. This is a good model if you implement vCenter on a physical server and do not mind accessing vCenter from the server console. There is no possibility that virtual machines can access the campus LAN or Internet and vice-versa. Because of this, virtual machine creation requires software installation from media and/or ISO images.



You may add an optional outside connection for the vCenter server. This allows indirect access to vCenter using Remote Desktop Protocol (RDP). An instance of the vSphere client may be installed and run on the vCenter server thereby allowing you to manage the vSphere infrastructure over the inside path, indirectly using RDP from the outside path.

Technically, the vSphere client running on an outside desktop can be used to access vCenter. However, attempts to open the console of a virtual machine will fail, as this requires a routable network path between the vSphere client desktop and ESXi host. This is why RDP is used to access an instance of the vSphere client running on the vCenter server itself.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.ne

2.7.3.1 Secure+ Networking Setup Tasks

If you have chosen the Secure+ networking model, the following setup tasks can be performed now.

- 1. If you are using a physical server for vCenter:
 - a. Optionally connect an Ethernet interface to a port on your campus LAN if you desire RDP management access. You will require a Campus IP address, subnet mask, and gateway setting for this interface.
 - b. Connect an Ethernet Interface to a control switch reserved port; a Gigabit uplink port can be used if available.
- 2. Connect the first Ethernet port on your ESXi to a designated reserved port on a control switch, or a Gigabit Ethernet uplink port (if one is available). A Gigabit port will provide the highest bandwidth.
- 3. Console into the control switch and configure each inside ESXi host switchport as a trunk (NETLAB+ does not do this automatically).

```
interface FastEthernet0/n
  description inside connection to ESXi host #
  switchport mode trunk
  switchport mode nonegotiate
```

Later in the setup process, you will configure virtual switches, VM kernel ports, and IP addresses for interfaces. The following table shows key configuration parameters that will be implemented for this networking model.

	NETLAB+		vCenter		ESXi Host 1		ESXi Host 2	
	Outside	Inside	Outside	Inside	Outside	Inside	Outside	Inside
IP Address	Campus	169.254.0.254	Campus+	169.254.0.253		169.254.0.241		169.254.0.242
Subnet Mask	Campus	255.255.255.0	Campus+	255.255.255.0		255.255.255.0		255.255.255.0
Gateway	Campus		Campus+	not set		not set		not set
vSwitch						vSwitch0		vSwitch0
Connection Path						*		*
			+ optional					

for RDP




2.8 Storage Area Networks

A storage area network (SAN) provides centralized shared storage for virtual machines and data. Sharing storage usually simplifies storage administration and adds flexibility since virtual machines can be migrated from one ESXi host to another without copying large files.

NDG performs all testing on servers with Internal Direct Attached Storage (i.e. RAID arrays and RAID controllers directly attached to each). This is the configuration that most academic institutions are likely to find affordable and adopt.

A Storage Area Network (SAN) is a dedicated network that provides access to consolidated, block level data storage that can be used for disk storage in a VMware vSphere environment.

Currently NDG does not provide benchmarks, guidance or troubleshooting for SAN configurations. Our documentation may show an optional SAN in the environment; however, this is not a recommendation or requirement to deploy a SAN.

NDG benchmarks and capacity planning guidance do not account for the additional latencies introduced by SAN.

- When compared to Direct Attached Storage, a SAN may introduce additional I/O latency between ESXi server and disk. Therefore, a SAN may reduce the number of active VMs you can run on an ESXi host.
- If you deploy a SAN, you should perform your own benchmarks and determine the number of active VMs you can host on your ESXi server. Your mileage may vary.
- Always configure NETLAB+ Proactive Resource Awareness to ensure that the number of VMs that can be activated will remain within your predetermined performance limits.

Caution. Deployment of a SAN requires skill and planning. Performance of SAN solutions can vary greatly. NDG performance benchmarks are based on Direct Attached Storage (local disks connected to each ESXi hosts). Free SAN solutions may be attractive, but test results show I/O rates that are half or less than directly attached SATA or SAS drives attached to the ESXi hosts.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

3 VMware ESXi Server Setup

This section describes the initial BIOS setup and software installation on a VMware ESXi host server.

The following table shows the minimum server configurations recommended for various NDG supported courseware. These configurations are based on the Dell R710 specification and vary only by memory and active VMs supported. You do not need separate host servers for each curriculum and may run VMs for Cisco, General IT, and Cybersecurity on the same servers. We recommend no more than 40 active VMs per server with 128GB of memory. Always configure NETLAB+ Proactive Resource Awareness to limit the number of scheduled VMs at any one given time and to prevent oversubscription of the host resources.

Server Role/Courses	Server Type	Processor(s)	Memory	Cores/Threads	Active VMs
Cisco Only Setup (8 Active MAP pods)	Dell R710	2 X Intel E5620	64GB	8/16	24
VMware ICM Course (8 Active ICM Pods)	Dell R710	2 X Intel E5620	128GB	8/16	40
General IT / Cybersecurity	Dell R710	2 X Intel E5620	128GB	8/16	40

All tasks in this section are performed on **separate dedicated physical servers** that you provide. Do not perform any of the tasks in this section on the NETLAB+ server appliance, as this will delete the NETLAB+ software, requiring you to return it to NDG for re-installation.

Setup of the VMware ESXi host server is illustrated in the sub-sections below using a Dell server model Dell R710. This is the recommended server model at this writing (QTR 4, 2011). Please also refer to the NDG website for the latest server recommendations. If you are using a recommended server other than the Dell R710, be aware that the process and screen images shown below will vary from your system.

Be sure that **hardware assisted virtualization support** (Intel-VT) is **enabled** in the BIOS. Some vendors disable this technology by default.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

3.1 **Preparing the ESXi Server**

If you are using the recommended server from section 2.5, there are several BIOS settings and RAID configurations that you will need to make sure are set correctly.

You must perform the steps detailed in the subsection below for every ESXi host server on your NETLAB+ system.

It is highly recommended that you read this section completely, prior to making changes to your system.

3.1.1 DELL R710 BIOS System changes

Please verify that your system has the latest BIOS installed. You may obtain the latest drivers and downloads for the Dell R710 from Dell's website.

Instructions for changing BIOS settings:

- 1. Turn on or restart your system.
- 2. Press <F2> to enter System Startup at the BIOS startup screen.



Note: BIOS Revision at the time of this document was 6.1.0.

3. Use the arrow keys to select **Memory Settings -> Memory Operating Mode** and make sure it is set to **Optimizer Mode**.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

Service Tag: Asset Tag:
System Time 17:39:54 System Date System Memory Size 128.0 GB Memory Setting Processor Sett System Memory Type ECC DDR3 SaTA Settings System Memory Ultage 1.350 Boot Settings Justem Memory Testing B MB Justem Memory Operating Mode Disabled Memory Operating Voltage Disabled Memory Operating Voltage Auto Serial Communication (ENTER) Serial Communication (ENTER)
Power Management

4. Press **ESC** to return to the main menu.



5. Use the arrow keys to select **Processor Settings - > Virtualization Technology** and make sure it is set to **Enabled.** Also, check **Execute Disable** is set to **Enabled**.



- 6. Press **ESC** to return to the main menu.
- 7. Use the arrow keys to select **Report Keyboard Errors** and make sure it is set to **Do Not Report**.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

Dell Inc. (www. BIO	deil.com) - PowerEdge #710 JS Version 6.1.0
Service Tag:	Asset Tag:
SATA Settings	(ENTER)
Doot Settings	(ENTER)
Integrated Devices PCI IRQ Assignment	CENTEB>
Serial Communication Embedded Server Management	(ENTEB) (ENTER)
Power Management System Security	CENTER> CENTER>
Keyboard NumLock Report Rephoerd Reserve	On
Up,Down Arrow to select SPAC	E,+,- to change ESC to exit F1=Heip

8. Use the arrow keys to select **F1/F2 Prompt on Error** and make sure it is set to **Disabled.**

	Dell I	Inc. (www.dell. BIOS Ver	com) - PowerEdg sion 6.1.0	e 8710	
Service	Tag :		Asset Tag:		
SATA Set Boot Set Integra PCI IR Serial Embedd Power System Keyboar Benort	ttings ttings ted Devices WARNING: After will not pause errors will be Log. rd MumLock Keuboard Errors	setting this f if an error oc displayed and	ield to Disable curs during POS logged to the S	(ENTER) (ENTER) (ENTER) d, the system T. Any critical ystem Event On On	ł
Up, Down	n Arrow to selec	t SPACE,+,-	to change	ESC to exit	F1=Help

9. Press ESC and select Save Changes and Exit.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

3.2 DELL R710 RAID Configuration

Redundant Array of Independent Disks (RAID) is designed to give the server redundancy and increased performance depending on the RAID type selected. The recommended and tested configuration is a RAID 5. The H700 RAID controller with 512MB cache is recommended and supported. **The H200 is NOT supported**. The H200 has exhibited poor performance and does not support RAID 5. Please see section 2.5.1 for details on hardware requirements.

RAID 5 distributes error-correcting bits (parity) along with the data and requires all drives but one to be present to operate; the array is not destroyed by a single drive failure. Upon drive failure, any subsequent reads can be calculated from the distributed parity such that the drive failure is masked from the end user. However, a single drive failure results in reduced performance of the entire array until the failed drive has been replaced and the associated data rebuilt.

RAID 5 arrays may take several hours to initialize as the controller creates parity on the drives. The 3x2TB drive configuration will take approximately **4** hours.

Please verify that your system has the latest H700 RAID Controller Firmware installed. You may obtain the <u>latest drivers and downloads</u> for the Dell R710 from Dell's website.

- 1. Turn on or restart your system.
- During the boot process, you will be prompted to, Press <Ctrl><R> to Run Configuration Utility.



Note: The H700 FW package at the time of this document was 12.10.2-0004.

3. You should start with a clean configuration. If not, you will need to clear any existing configuration.



3.2.1 3x2TB HDD Configuration

This section is for the 3x2TB HDD Configuration for RAID 5. When installing ESXi 4.1 U2, you must create two virtual disks when using this HDD configuration. One is a 2TB datastore, and the other is the remaining space in another datastore.

1. Use the arrow keys to select the PERC H700 controller and press enter.



- 2. Select RAID Level 5, RAID-5.
- 3. Select all your physical disks on the left.
- 4. Edit the VD Size to **2048.00 GB**. This will be the first datastore.
- 5. Edit the VD Name to **VD-DAS1**.

Uiri RAID Level : RAID-5 PD per Span :N/A Physical Disks Disk ID Size # IX100:00:00 1862.50 GB 00 IX100:00:01 1862.50 GB 01 IX100:00:02 1862.50 GB 02	Create New UD Basic Settings UD Size: 2048.00 GB UD Mame: UD-DAS1 L 1 Advanced Settings Strip Element Size: 64KB Read Policy : Adaptive R CANCEL
Secure VD: No 1-Help F12-Ctlr	Write Policy: Write Back [] Force WB with no battery [] Initialize [] Configure HotSpare



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

- 6. Select **OK** on the right. Note the message indicating we need to initialize the Virtual Disk after it has been configured. We will do that in a later step.
- 7. Use your arrow keys to highlight Free Capacity and press Enter.
- 8. Edit the VD Name to **VD-DAS2.** Select **OK**. Note the message indicating we need to initialize the Virtual Disk after it has been configured. We will do that in a later step.
- Use your arrow keys to highlight Virtual Disk 0. Press F2 and select initialization

 Start Init. Select Yes to confirm Initialization of the HDDs. This process can take several hours.

Urtual Disk Hanagement [-] PERC H760 Integrated (Bus 0x03, Dev 0x00) LI-1 Disk Group: 0. Raid 5 I-1 Uirtual Disks Uirtual Disk: 0. UD Uirtual Disk: 1. UD Start Init. I+1 Thysical Disks I+1 Total Free Capacity: Hot spares Hot spares	Virtual Disk 0: State: Optimal Initialization Consistency Check Belete VD Properties
	Expand VD size

10. You can press **Enter** on the Virtual Disk to see the estimated time remaining.

Uirtual Di	tual Disk Management sk: 0, VD-DAS1, 2048.00 GB	
RAID Level : RAID-5	UD Size:2048.00 GB	Operation: Init.
RAID State: Optimal Physical Disks Disk ID Size	I Advanced Settings	Progress : ^{0%}
00:00:00 1862.50 GB 00 00:00:01 1862.50 GB 01 00:00:02 1862.50 GB 02	Element Size: 64KB	Time Left : 02:11:50

- 11. When initialization is completed, select **OK** to confirm.
- 12. Repeat steps 8-10 for Virtual Disk 1.
- 13. Press **ESC** to close out of the RAID Configuration Utility. Select **OK** to confirm exiting. You will be prompted to reboot your system.



3.2.2 3x1TB HDD Configuration

This section is for the 3x1TB HDD Configuration for RAID 5. When installing ESXi 4.1 U2, you create only one datastore in this HDD configuration.

1. Use the arrow keys to select the PERC H700 controller and press enter.



- 2. Select RAID Level 5, RAID-5.
- 3. Select all your physical disks on the left.
- 4. Edit the VD Name to **VD-DAS1**.

RAID Level : RAID-5		Create New UD Basic Settings UD Size:1862.00 GB	
PD per Span : N/A Physical Disks Disk ID Size [X100:00:00 931.00 GB	# 00	UD Name: UD-DAS1 I Advanced Settings Strip Element Size: 64KB	OK
[X]00:00:02 931.00 GB	02	Read Policy : Adaptive R	CANCEL



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.

- 5. Select **OK** on the right. Note the message indicating we need to initialize the Virtual Disk after it has been configured. We will do that in a later step.
- Use your arrow keys to highlight Virtual Disk 0. Press F2 and select initialization

 Start Init. Select Yes to confirm Initialization of the HDDs. This process can take several hours.



7. You can press **Enter** on the Virtual Disk to see the estimated time remaining.

VII I	caar Disk nanayement	
Virtual Dis	sk: 0. UD-DAS1. 1862.00 GB 🚃	
	- Basic Settings	
RAID Level : RAID-5	UD Size: 1862 00 GB	Onenation:
	V 0120.1002.00 0D	operación
		Init.
	UD Name UD-DAS1	
DATE OF LA CALL	VD Hames VD DHOI	
KAID State: Uptimal		Progress :
- Physical Disks	-[] Advanced Settings	A.1
This Tour Prono	i i navanceu seccinys	17.
DISK ID Size #	Strip	
00:00:00 931 00 CB 00	Flement Size: 64VR	Time Loft .
00.00.00 JJ1.00 dB	LICHCHC JIZC. DIND	TIME TELC .
00:00:01 931.00 GB 01		02:28:39
00:00:02 931 00 CR 02		
00.00.02 JJ1.00 db 02		

- 8. When initialization is completed, select **OK** to confirm.
- 9. Press **ESC** to close out of the RAID Configuration Utility. Select **OK** to confirm exiting. You will be prompted to reboot your system.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netder

3.3 Installing ESXi on Host Server

This section will walk you through installing VMware ESXi to your host servers. Please note that the content in the images below will vary based on your system. The instructional steps are the same.

If you are using VMware ESXi 4.1 U2, please review the information in the <u>Getting</u> <u>Started with ESXi Server Installable</u> guide. You may use this guide as a reference for ESXi 4.1 Installable and ESXi 4.1 Embedded versions, with the exception that using **ESXi Installable requires performing the installation procedure detailed on page 7, Install ESXi 4.1 using the Interactive Mode.**

If you are using VMware ESXi 4.01, please review the information in the <u>Getting Started</u> <u>with ESXi Server Installable</u> guide. You may use this guide as a reference for ESXi 4.01 Installable and ESXi 4.01 Embedded versions, with the exception that using **ESXi Installable requires performing the installation procedure detailed on page 9** *Install ESXi 4.0.* Keep in mind that the procedures detailed in this guide recommends and illustrates the use of ESXi 4.1.

If you have not done so already, burn a copy of the ESXi 4.1 Installable ISO.

Instructions for installing ESXi:

- 1. Turn on or restart your system.
- 2. Insert your ESXi 4.1 media to install the ESXi Installable.
- 3. You are presented with the Installation screen below.



- 4. Press Enter to continue.
- 5. You are prompted with the User Agreement. Press **F11** to Accept and Continue.
- 6. On the Select a Disk screen, choose the 2TB drive.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.co

		Select	a Disk			
torage Dev	ice				Capac	ity
ocal:						
DELL PERC	H700 (naa.67	82bcb04bd2f	500167de6da	11824013)		TB
DELL PERC	H700 (naa.67	B2bcb04bd2f	500167de6ff	13bb30c4)	1.64	TB
emote:						

This screenshot is for the 3x2TB HDD configuration. For 3x1TB HDD configuration, there is only one drive available, select it.

- 7. Press Enter to continue.
- 8. Press F11 to begin the installation.
- 9. Installation can take 1-5 minutes depending on hardware.
- 10. When the Installation Complete window appears, press **Enter** to reboot the machine.

3.4 Configure Root Password

In this task, you will log into the console and setup the root password.

1. When the ESXi host server is booted, you are prompted with the console screen.



- 2. Press F2 to begin setting up the host.
- 3. Choose **Configure Password** and press **Enter** to change.



4. In the Configure Password window, enter a New Password. Confirm the password and press **Enter** to make the change.

Be sure to record this password in a safe place. You will need it to integrate the ESXi host with the vCenter Server on the management workstation.

3.5 Network Configuration

In this task, you will configure ESXi server network configuration.

1. From the console, choose **Configure Management Network** and press **Enter.**

System Customization	Configure Management Network
Configure Password Configure Lockdown Mode	Hostname:
Configure Management Network Restart Management Network Test Management Network	IP Address:
Disable Management Network Restore Standard Switch	To view or modify this host's мападемепt network settings in detail, press <enter>.</enter>

- 2. In the Configure Management Network window, choose **IP Configuration** and press **Enter.**
- 3. In the IP Configuration window, choose **Set static IP address and network configuration**, press **Spacebar** to select and enter an inside or outside IP configuration (described in the next two sections respectively).



The IP configuration parameters entered here will be used either for the outside or inside interface, depending on the networking model you have chosen. Please see the following discussion to determine which interface IP values to use.



Outside ESXi Connection. If you are NOT using Secure+ network model, your ESXi host connects to the outside campus LAN. Enter the IP address, subnet mask, and default gateway assigned by the campus LAN administrator.



Single Homed Networking





Dual-Homed Networking



Inside ESXi Connection ONLY. If you are using the Secure+ network model, your ESXi host connects only to the inside control switch. Use one of the following IP configurations for each ESXi server.

Inside Interface	IP Address	Subnet Mask	Default Gateway
ESXi Server 1 Inside	169.254.0.241	255.255.255.0	not set
ESXi Server 2 Inside	169.254.0.242	255.255.255.0	not set
ESXi Server 3 Inside	169.254.0.243	255.255.255.0	not set
ESXi Server 4 Inside	169.254.0.244	255.255.255.0	not set
ESXi Server 5 Inside	169.254.0.245	255.255.255.0	not set
ESXi Server 6 Inside	169.254.0.246	255.255.255.0	not set
ESXi Server 7 Inside	169.254.0.247	255.255.255.0	not set
ESXi Server 8 Inside	169.254.0.248	255.255.255.0	not set
ESXi Server 9 Inside	169.254.0.249	255.255.255.0	not set

Secure+ Networking





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

- 4. Press **Enter** to confirm changes and return to the Configure Management Network window.
- 5. In the Configure Management Network window, choose **DNS Configuration** and press **Enter**.
- 6. Select Use the following DNS server addresses and hostname.

DNS Configuration
This host can only obtain DNS settings automatically if it also obtains its IP configuration automatically.
() Obtain DNS server addresses and a hostname automatically (o) Use the following DNS server addresses and hostname:
Primary DNS Server I] Alternate DNS Server [SEE TABLE BELOW] Hostname []
<pre><up down=""> Select <space> Mark Selected <enter> OK <esc> Cancel</esc></enter></space></up></pre>

7. Enter either outside or inside DNS parameters from the table below based on your networking selections in section 3.5.

	If you configured outside interface and IP parameters in section 3.5	If you configured inside interface and IP parameters in section 3.5
Primary DNS Server	Primary DNS server IP address for your campus LAN.	Leave Blank
Alternate DNS Server	Alternate DNS server IP address for your campus LAN.	Leave Blank
Hostname	Use outside IP address as the host name (recommended), or the fully qualified domain name of the host if mapped in your DNS.	Use inside IP address as the host name.

- 8. Press **Enter** to confirm changes and return to the Configure Management Network window.
- 9. Press **Esc** to return to the main console.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

10. You will be prompted to **Apply changes and restart management network**. Press **Y** to confirm changes.



11. You will return to the main console. Press **Esc** to log out.

Reminder: The installer available from the VMware Academy is for ESXi 4.1. NDG recommends upgrading to ESXi 4.1 U2.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netd

4 VMware vCenter Server Setup

In this section, you will setup the vCenter server and related utilities.

A separate server running a **64-bit** Windows Server operating system is required for vCenter Server Standard. This server can be a physical server (bare metal) or virtual machine running on a VMware ESXi 4.1 U2 host. In either case, the physical server on which vCenter resides should be a dedicated "management server" to provide ample compute power. VMware vCenter server requires at least two CPU cores.

NDG does not support configurations where vCenter is running on a heavily loaded ESXi host and/or an ESXi host that is also used to host virtual machines for NETLAB+ pods. Such configurations have exhibited poor performance, API timeouts, and sporadic errors in NETLAB+ operations.

4.1 vCenter Configuration Options

There are four configuration options that are recommended by VMware and supported by NDG.

Option	Host	Operating System	vCenter Version	Database	Host/VM Limit
1	VMware ESXi 4.1 U2	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Server	
2	VMware ESXi 4.1 U2	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Express (built-in database)	5 / 50
3	Bare Metal	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Server	
4	Bare Metal	Windows Server 2008 R2 (64-bit)	vCenter 4.1 for Windows	Microsoft SQL Express (built-in database)	5 / 50



4.1.1 vCenter Configuration Option 1

vCenter Option 1 provides the benefits of virtualization with a full version of Microsoft SQL Server database to support a large number of virtual machines.

Microsoft SQL Server 2008 R2
<u> </u>
VMware vCenter
Windows Server 2008 R2 64-bit
vCenter VM
vCenter VM Mare ESXi 4.1 U2
vCenter VM M VMware ESXi 4.1 U2
vCenter VM VMware ESXi 4.1 U2

vCenter Option 1

The following components are required for Option 1:

- Physical Management Server running VMware ESXi 4.1 U2
- Virtual machine running Windows Server 2008 R2 64-bit
- VMware vCenter 4.1 for Windows
- Microsoft SQL Server 2008 R2

This option does not provide a physical console to access Windows Server or vCenter. If this option is combined with the Secure+ networking configuration, your management server must have a network connection to the campus to provide access to vCenter and/or the Window Server virtual machine.



4.1.2 vCenter Configuration Option 2

Option 2 provides the benefits of virtualization without a requirement for a separate database. This option uses the Microsoft SQL Express that is included with vCenter for Windows.

The Microsoft SQL Express database included with the vCenter installation supports no more than 5 hosts or 50 virtual machines. It is suitable for a small installation, such as managing a NETLAB+ configuration consisting only of Cisco pods. The express database also has a maximum data capacity of 4GB. As the size of the database approaches 4GB, vCenter performance may degrade. If the 4GB limit is reached, vCenter will no longer function.



vCenter Option 2

The following components are requirement for Option 2:

- Physical Management Server running VMware ESXi 4.1 U2
- Virtual machine running Windows Server 2008 R2 64-bit
- VMware vCenter 4.1 for Windows
- Microsoft SQL Express (packaged with and installed by vCenter)

This option does not provide a physical console to access Windows Server or vCenter. If this option is combined with the Secure+ networking configuration, your management server must have a network connection to the campus to provide access to vCenter and/or the Window Server virtual machine.



4.1.3 vCenter Configuration Option 3

Option 3 provides vCenter running directly on a dedicated physical server, with a full version of Microsoft SQL Server installation to support a large number of virtual machines.

vCenter Option 3



The following components are required for Option 3:

- Physical server with at least 2 CPU cores (4 cores recommended)
- Windows Server 2008 R2 64-bit
- VMware vCenter 4.1 for Windows
- Microsoft SQL Server 2008 R2

If you use an "older" physical server for you vCenter system, please be sure it meets the minimum requirements in the VMware documentation. vCenter absolutely requires a 64-bit processor and at least two cores. An underpowered server will likely cause performance problems and errors in your infrastructure.

Important: Broadcom NIC chipsets have shown intermittent traffic issues, including dropped packets between NETLAB+ and vCenter, if using the Microsoft Drivers from Windows Update. It is recommended that you use the Broadcom Drivers available from their support website for your appropriate chipset. I.E. The Dell R710 uses the Broadcom[®] NetXtreme II 5709 chipset for the embedded NICs. These drivers can be downloaded at http://www.broadcom.com/support/ethernet_nic/netxtremeii.php.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

4.1.4 vCenter Configuration Option 4

Option 4 provides vCenter running on a dedicated physical server, without a requirement for a separate database. This option uses the Microsoft SQL Express that is included with vCenter for Windows.

The Microsoft SQL Express database supports no more than 5 hosts or 50 virtual machines. It is suitable for a small installation, such as managing a NETLAB+ configuration consisting only of Cisco pods. The express database also has a maximum data capacity of 4GB. As the database approaches 4GB, performance may degrade. At 4GB, the database will not function.



vCenter Option 4

(purchased separately)

The following components are requirement for Option 4:

- Physical server with at least two CPU cores
- Windows Server 2008 R2 64-bit
- VMware vCenter 4.1 for Windows
- Microsoft SQL Express (packaged with vCenter)

If you use an "older" physical server for you vCenter system, please be sure it meets the minimum requirements in the VMware documentation. vCenter absolutely requires a 64-bit processor and at least two cores. An underpowered server will likely cause performance problems and errors in your infrastructure.

Important: Broadcom NIC chipsets have shown intermittent traffic issues, including dropped packets between NETLAB+ and vCenter, if using the Microsoft Drivers from Windows Update. It is recommended that you use the Broadcom Drivers available from their support website for your appropriate chipset. I.E. The Dell R710 uses the Broadcom[®] NetXtreme II 5709 chipset for the embedded NICs. These drivers can be downloaded at http://www.broadcom.com/support/ethernet_nic/netxtremeii.php.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

4.2 Networking Overview for vCenter Server

The following sub-sections enumerate both physical and virtual setups for vCenter in the context of the three NETLAB+ networking models (Single Homed, Dual Homed, and Secure+).

Section 4.2.1 discusses virtual vCenter Server (Options 1 and 2) networking setups for Single Homed Networking (section 4.2.1.1), Dual Homed Networking (section 4.2.1.2), and Secure+ Networking (section 4.2.1.3).

Section 4.2.2 discusses vCenter on bare metal server (Options 3 and 4) networking setups for Single Homed Networking (section 4.2.2.1), Dual Homed Networking (Section 4.2.1.2, and Secure+ Networking (section 4.2.1.3)

4.2.1 Virtualized vCenter Server Networking Options

The following three setups apply to vCenter Options 1 and 2 (virtualized vCenter).

4.2.1.1 Virtualized vCenter Server with Single Homed Networking

This setup applies to vCenter Options 1 and 2 when used in a Single Homed Networking configuration.

- The management server will have one physical network connection to the campus LAN. The management server's ESXi kernel NIC on vSwitch0 is assigned an IP address on the campus LAN. You will connect directly to this address via vCenter to manage the ESXi server before vCenter is installed.
- The vCenter server virtual machine will have one virtual network adapter. This adapter will be assigned an IP address on your campus LAN. This IP address will be used to connect to vCenter once it is installed.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.co



4.2.1.2 Virtualized vCenter Server with Dual Homed Networking

This setup applies to vCenter Options 1 and 2 when used in a Dual Homed Networking configuration.

- The management server will have two physical network connections.
 - o vmnic0 and vSwitch0 connect to the campus LAN.
 - vmnic1 and vSwitch1 connect to a NETLAB+ control switch.
 - The management server's ESXi kernel NIC on vSwitch0 is assigned an IP address on the campus LAN. You will connect directly to this address via vCenter to manage the ESXi server before vCenter is installed.
 - The management server's ESXi kernel NIC on vSwitch1 is assigned as 169.254.0.239 (recommended).
- The vCenter server virtual machine will have two virtual network adapters.
 - Network adapter 1 connects to vSwitch0 (campus). This adapter will be assigned an IP address on your campus LAN. This IP address will be used to connect to vCenter (once it is installed).
 - Network adapter 2 connects to vSwitch1 (campus). The recommended IP address is 169.254.0.253.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com



4.2.1.3 Virtualized vCenter with Secure+ Networking

This setup applies to vCenter Options 1 and 2 when used in a Secure+ Networking configuration. This setup is identical to Dual Homed Networking (previous section), except that the system administrator will connect to the vCenter server using RDP and run the vSphere client from the vCenter server.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.cor



Technically, you may connect to vCenter by running the vSphere client from a desktop on the campus LAN. However, you will not be able to console into a VM on the pod hosts. This is because the pod hosts are not connected to the campus LAN and the vCenter VM is not a router. Launching the vSphere client from the vCenter host circumvents this problem because outgoing connections from vSphere client will originate from the 169.254.0.253 interface and therefore do not require routing.



4.2.2 Bare Metal vCenter Server Networking Options

The following three setups apply to vCenter Options 3 and 4 (vCenter on bare metal server).

4.2.2.1 Bare Metal vCenter Server with Single Homed Networking

This setup applies to vCenter Options 3 and 4 when used in a Single Homed Networking configuration.

- The vCenter server will have one physical network connection to the campus LAN and is assigned a campus IP address.
- This IP address will be used to connect to vCenter once it is installed.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

4.2.2.2 Bare Metal vCenter Server with Dual Homed Networking

This setup applies to vCenter Options 3 and 4 when used in a Dual Homed Networking configuration.

- The management server will have two physical network connections (inside and outside).
 - The outside connection is assigned a campus IP address, which will be used to connect to vCenter once it is installed.
 - The inside connection is assigned the address 169.254.0.253. This will be used for management and NDG troubleshooting.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netc

4.2.2.3 Bare Metal vCenter Server with Secure+ Networking

This setup applies to vCenter Options 3 and 4 when used in a Secure+ Networking configuration. This setup is identical to Dual Homed Networking (previous section), except that the system administrator will connect to the vCenter server using RDP and run the vSphere client from the vCenter server.

In this configuration, it is technically possible to omit the connection from vCenter server to campus LAN. However, it will require many setup and management tasks to be performed from the server console. Therefore, we recommend leaving the campus connection in place.



Technically, you may connect to vCenter by running the vSphere client from a desktop on the campus LAN. However you will not be able to console into a VM on the pod hosts. This is because the pod hosts are not connected to the campus LAN and the vCenter VM is not a router. Launching the vSphere client from the vCenter host circumvents this problem because outgoing connections from vSphere client will originate from the 169.254.0.253 interface and therefore do not require routing.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netd

4.3 Configure Management Server

This task applies to (virtual) vCenter Options 1 and 2 only.

- 1. Install ESXi 4.1 U2 on your management server using guidance from section 3.
- 2. Open the vSphere client from a workstation on your campus network.
- 3. Connect directly to the IP address of the management server.
- 4. Enter the username of **root**.
- 5. Enter the password of used during ESXi installation.
- 6. Create a new virtual machine for vCenter using the following parameters.

vCPUs	2
vRAM	4GB or higher if using Microsoft SQL Server (Option 1) 3GB or higher if using Microsoft SQL Express (Option 2)
Network Adapters	 1 for Single Homed Network configuration 2 for Dual Homed Networking configuration 2 for Secure+ Networking configuration

- Bind network adapter 1 to the vSwitch that is connected to the campus LAN. This is usually vSwitch0 and is called "VM network" in the pull down menu by default.
- 8. Bind network adapter 2 (if Dual Homed or Secure+) to the vSwitch that connects with the NETLAB+ control switch. This is typically vSwitch1.

4.4 Install Windows Server 2008 R2 64-bit

At this time, you should install **Windows Server 2008 R2 64-bit.**

- For a virtual vCenter configuration (Option 1 or 2), you will install Windows Server on the virtual machine created in section 4.3.
- For a bare metal vCenter configuration (Option 3 or 4), you will install Windows Server on the physical server you have dedicated for vCenter.

Older NETLAB+ servers have CD-ROM drive. Please DO NOT accidently install Windows Server on the NETLAB+ server! This will overwrite the NETLAB+ operating system and software.

Apply all operating system updates are applied before continuing.



4.5 Configuring TCP/IP on vCenter Server Network Adapters

In the next two sections, you will setup TCP/IP on the vCenter server network interfaces.

4.5.1 Outside Interface

In this task, you will configure TCP/IP on the Windows Server network adapter that connects the vCenter server to your campus LAN (outside).



Please refer to the illustration on the next page for the following tasks.

- 1. On your vCenter server Windows installation, navigate to Network Connections.
 - a. Click on the **Start Menu.** In the search field enter **network connections**.
 - b. Click on View network connections.
- Identify the outside LAN adapter on the Network Connections window. On a
 physical server, you may temporarily unplug the outside LAN Ethernet cable
 while viewing the Network Connections window to identify the outside adapter;
 the corresponding Windows adapter will transition from connected to not
 connected. Reconnect the cable once you have identified the outside LAN
 adapter.
- 3. Right-click on the Local Area Connection that connects to your campus and select **Rename**. Give an appropriate name such as **Outside**, **Campus**, or **Internet** so that it is easily identifiable.
- 4. Right click on the Local Area Connection that connects to your campus LAN (outside), then select **Properties**.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

- 5. Click on "Internet Protocol Version 4 (TCP/IPv4)", then click the Properties button.
- 6. Click "Use the following IP address".
- 7. Enter the vCenter server IP address, Subnet Mask, and Default Gateway that will be used to access vCenter from your campus LAN connection.
- 8. Click "Use the following DNS server addresses".
- 9. Enter the primary and alternate DNS server addresses used to resolve names on your campus LAN.
- 10. Click **OK** on all dialogs to finish complete the task.
- 11. Open a command prompt and ping the outside address of the NETLAB server (as locally assigned) to verify that your vCenter outside interface is working properly.





4.5.2 Inside Interface

In this task you will configure TCP/IP on the interface that connects the vCenter server to your inside network (control switch if physical server).





Please refer to the illustration on the next page for the following tasks.

- 1. On your vCenter server, navigate to Network Connections.
 - a. Click on the **Start Menu.** In the search field enter **network connections**.
 - b. Click on View network connections.
- Identify the inside LAN adapter on the Network Connections window. On a
 physical server, you may temporarily unplug the inside LAN Ethernet cable while
 viewing the Network Connections window to identify the inside adapter; the
 corresponding Windows adapter will transition from connected to not
 connected. Reconnect the cable once you have identified the inside LAN adapter.
- 3. Right-click on the Local Area Connection that connects to your NETLAB+ control switch and select **Rename**. Give an appropriate name such as **NETLAB Inside** so that it is easily identifiable.
- 4. Right click on the Local Area Connection that connects to your NETLAB+ control switch (the inside LAN adapter), then select **Properties**.



- 5. Click on Internet Protocol Version 4 (TCP/IPv4) and then click the Properties button.
- 6. Click "Use the following IP address".

ping 169.254.0.254

- 7. Enter the vCenter server inside IP address and Subnet Mask.
 - a. The recommended IP address is 169.254.0.253.
 - b. The subnet mask is **255.255.255.0**.
 - c. The default gateway should be blank (not set).
- 8. Click "Use the following DNS server addresses". Leave these settings blank (not set).
- 9. Click **OK** on all dialogs to finish complete the task.
- 10. Open a command prompt and ping the NETLAB+ server inside interface at to verify that your inside vCenter interface can communicate on the inside network.

🕥 💽 🔹 Control Panel 🔹 Network and Internet 🔹 Networ	rk Connections 👻	
Organize 👻 Disable this network device Diagnose this connect	ion Rename this connection View status of this connection Change settings	of thi
Outside-Campus Network Intel(R) PRO/1000 MT Network Conn.	LAB Inside Jentified network 4(R) PRO/1000 MT Network Conn	
🖣 NETLAB Inside Properties	Internet Protocol Version 4 (TCP/IPv4) Properties	×
Networking Sharing	General	
Connect using:	You can get IP settings assigned automatically if your network supports this capability. Otherwise, you need to ask your network administrator for the appropriate IP settings.	
Configure This connection uses the following items:	C Obtain an IP address automatically	
Client for Microsoft Networks	IP address: 169 . 254 . 0 . 253	
G upos Packet Scheduler Scheduler	Subnet mask: 255 , 255 , 255 , 0	
 Internet Protocol Version 4 (TCP/IPv4) 	Default gateway:	
	Obtain DNS server address automatically	
	Use the following DNS server addresses:	
Install Uninstall Properties	Preferred DNS server:	
Description	Alternate DNS server:	
I ransmission Control Protocol/Internet Protocol. The default wide area network protocol that provides communication across diverse interconnected networks.	Validate settings upon exit Advanced	
		=1


Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

4.6 Installing VMware vCenter and Related Software

vCenter Server requires Windows Server 2008 R2 64-bit. Please install this on your virtual machine or physical machine depending on your setup. Make sure all operating system updates are applied before continuing. If you install vCenter Server on a virtual machine, be sure to install VMware Tools before continuing.

4.6.1 Installing vCenter with Microsoft SQL Server 2008 R2 (Option 1 and 3)

This section is recommended for vCenter deployments **that will exceed 50 virtual machines**. If you are not exceeding 50 virtual machines, but plan to in the near future, it is recommended that you use this section.

This setup is the same for both virtual vCenter (Option 1) and bare metal vCenter (Option 3) installs. Please have your license key for Microsoft SQL Server 2008 R2 available. Microsoft SQL Server 2008 R2 must be installed and a database created before installing vCenter Server. This section will assist you in that setup.



4.6.1.1 Configure Hostname and Create User Account

 Before installation, you must set the hostname of the machine to something identifiable. Click on the Start Menu. Right-click on Computer and then click on Properties.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

2. Under the *Computer name, domain, and workgroup settings* section, click on the **Change settings** link.

Computer name, domain, an	d workgroup settings	 	 	
Computer name:	WIN-F2LUN7QQI9H			Change settings
Full computer name:	WIN-F2LUN7QQI9H			
Computer description:				
Workgroup:	WORKGROUP			
 .ue.1				

3. A *System Properties* window will appear. Make sure the **Computer Name** tab is selected and click the **Change** button.

System Properties		×			
Computer Name Hardwa	are Advanced Remote				
Windows uses on the network	the following information to identify your computer <.				
Computer description:					
	For example: "IIS Production Server" or "Accounting Server".				
Full computer name:	WIN-F2LUN7QQI9H				
Workgroup:	Workgroup: WORKGROUP				
To rename this computer workgroup, click Change	or change its domain or Change				
	OK Cancel Apply				



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

4. Clear the **Computer Name** field and enter **vCenter** or a unique name of your choice. Click on **OK** to save settings.

Computer Name/Domain Changes
You can change the name and the membership of this computer. Changes might affect access to network resources. <u>More information</u>
Computer name: vCenter
Full computer name: vCenter
More
Member of
O Domain:
• Workgroup:
WORKGROUP
OK Cancel

- 5. You will be prompted with a window explaining you must restart your computer to apply these changes. Click **OK**.
- 6. Click close on the **Computer Name** window. You will be prompted to restart your computer. Click **Restart Now.**
- 7. You need to create a **sqladmin** account for starting SQL Server services. Click on the **Start Menu** and right-click on **Computer**. Click on **Manage**.

	Computer	Open
	Network	🚱 Manage
		Map network drive
	Control Panel	Disconnect network drive
	Devices and Pr	Show on Desktop Rename
	Administrative	Properties
	Help and Suppo	ort
	Run	
All Programs	Windows Secur	ity
Search programs and files	Log off	
🎦 Start 🕌 🗾 🚞 🥹		



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgr

8. Click on **Configuration->Local Users and Groups->Users** on the left hand side. Right-click in the white space below the listed users and click **New User...** to create a new user.



9. On the New User window, in the Username field, enter sqladmin. In the Password field, enter a unique password for this account. It is strongly recommended that you do not use the same password as your Administrator account. Confirm the password. Uncheck User must change password at next logon and put a checkmark next to Password never expires. Click Create to continue.

New User	? ×
User name: sqladmin	
Full name:	
Description:	
	_
Password:	_
Confirm password:	
User must change password at next logon	
User cannot change password	
Password never expires	
C Account is disabled	
Help Create Close	

10. Click Close on the New User window.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.r

11. Confirm the **sqladmin** user shows up in the list of users. Close the *Server Manager* window.

4.6.1.2 Install Microsoft SQL Server 2008 R2

- 1. Please insert the Microsoft SQL Server 2008 R2 CD/DVD into the drive. If this is a virtual machine, attach the appropriate image to the virtual machine.
- 2. You may be prompted to install .NET Framework and update Windows installer. Click **OK** to continue.



- 3. The *SQL Server Installation Center* window will appear. Click on **Installation** on the left hand side.
- 4. Click on New installation or add features to an existing installation.





5. SQL Server 2008 R2 setup will perform a series of checks on rules. When the operation has completed, confirm all passed and click **OK** to continue.

🚼 SQL Server 2008 R2 Setup		
Setup Support Rules	5	
Setup Support Rules identify before Setup can continue.	problems that might occur when you install SQL Server Setup support files. Failures must be corrected	
Setup Support Rules	Operation completed. Passed: 7. Failed 0. Warning 0. Skipped 0.	
	Show details >> View detailed report	Re-run

- 6. You will be prompted to enter your Product Key. Note that if using a MSDN-AA version or comparable, the key may already be entered. Click **Next** to continue.
- 7. Read the Microsoft Software License Terms. If you accept, click the checkbox net to I accept the license terms. Click Next to continue.
- 8. You may begin installing the **Setup Support Files**. Click **Install** to continue installation.
- 9. Once the support files have been installed, another check is performed on setup support rules. Confirm all passed. (You may get a warning on Windows Firewall. This is ok.) Click **Next** to continue.

Hide View	e details << detailed report	Re-run
Rul	Rule	Status
\bigcirc	Fusion Active Template Library (ATL)	Passed
0	Unsupported SQL Server products	Passed
0	Performance counter registry hive consistency	Passed
0	Previous releases of SQL Server 2008 Business Intelligence Develop	Passed
0	Previous CTP installation	Passed
0	Consistency validation for SQL Server registry keys	Passed
0	Computer domain controller	Passed
0	Microsoft .NET Application Security	Passed
0	Edition WOW64 platform	Passed
0	Windows PowerShell	Passed
	Windows Firewall	Warning

10. On the *Setup Role* window, select **SQL Server Feature Installation** and click **Next** to continue.



11. On the *Feature Selection* window, click on **Select All**. Then uncheck **Analysis Services** and **Reporting Services**. Click on **Next** to continue.

Features:	Description:
Instance Features ✓ Database Engine Services ✓ SQL Server Replication ✓ Full-Text Search Analysis Services Reporting Services Shared Features ✓ Business Intelligence Development Studio ✓ Client Tools Connectivity ✓ Integration Services ✓ Client Tools Sorkwards Compatibility ✓ Client Tools SDK ✓ SQL Server Books Online ✓ Management Tools - Complete ✓ SQL Client Connectivity SDK ✓ Microsoft Sync Framework Redistributable Features	Server features are instance- aware and have their own registry hives. They support multiple instances on a computer.
Select All Unselect All	

- 12. On the *Installation Rules* window, confirm all rules passed and click **Next** to continue.
- 13. On the *Instance Configuration* window, leave the default setting of settings and click **Next** to continue.
- 14. On the *Disk Space Requirements* window, review the information and click **Next** to continue.
- 15. On the *Server Configuration* window, click the **Use the same account for all SQL Server services** button.

Service	Account Name	Password	Startup Type
SQL Server Agent			Manual 🗾
SQL Server Database Engine			Automatic 📃
SQL Server Integration Services 10.0	NT AUTHORITY\Network		Automatic 🗾
SQL Full-text Filter Daemon Launcher	NT AUTHORITY\LOCAL S		Manual
SQL Server Browser	NT AUTHORITY\LOCAL S		Disabled 💌

16. A window will appear. Click **Browse...** to continue.



17. On the *Select User or Group* window, enter the **sqladmin** account you created earlier and click **Check Names** to confirm the name. If found, the name will change to the COMPUTER-NAME\sqladmin (i.e. VCENTER\sqladmin). Click **OK** to continue.

Select User or Group		? ×
Select this object type:		
User, Group, or Built-in security principal		Object Types
From this location:		
VCENTER		Locations
Enter the object name to select (<u>examples</u>):		
VCENTER\sqladmin		Check Names
		_
Advanced	OK	Cancel

18. Confirm the *Account Name* field is now filled and enter the password for that account. Click on **OK.**

Use the same a	ccount for all SQL Server 2008 R2 services		
Specify a user n	ame and password for all SQL Server service accounts.		
Account Name:	sqladmin	•	Browse
Password:	*****		
		ОК	Cancel

19. On the *Server Configuration* window, locate the **SQL Server Agent** service and change the **Startup Type** to **Automatic.** Click **Next** to continue.

rvice Accounts Collation	eparate account for each SOL	Server service.			
Service Account Name Password Startup Type					
SQL Server Agent	sqladmin	•••••••••	Automatic 💦 💆		
SQL Server Database Engine	sqladmin	••••••	Automatic 🖉 💌		
SQL Server Integration Services 10.0	sqladmin	••••••	Automatic 🛛 💌		
SQL Full-text Filter Daemon Launcher	NT AUTHORITY\LOCAL S		Manual		
SQL Server Browser	NT AUTHORITY\LOCAL S		Disabled 💌		

- 20. On the Database Engine Configuration window, click on the Add... button.
- 21. On the *Select User or Group* window, enter the **sqladmin** account and click **Check Names** to confirm the name. If found, the name will change to the COMPUTER-NAME\sqladmin (i.e. **VCENTER\sqladmin**). Click **OK** to continue.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

Select User or Group		? ×
Select this object type:		
User, Group, or Built-in security principal		Object Types
From this location:		
VCENTER		Locations
Enter the object name to select (<u>examples</u>):		
VCENTER\sgladmin		Check Names
1		_
Advanced	OK	Cancel

22. On the *Database Engine Configuration* window, click on the **Add Current User** button. You should see the account added in the list. Confirm the two accounts are there and click **Next** to continue.

	Specify SQL Server administrators	SQL Se have u the Da
	Add Current User Add Remove	
-	< Back Next >	

- 23. On the Error Reporting window, click Next to continue.
- 24. On the *Installation Configuration Rules* window, additional checks are performed. Confirm all have passed and click **Next** to continue.
- 25. On the *Ready to Install* window, review the information and click on **Install** to begin installation.
- 26. On the *Installation Progress* window, you can follow the progress. This may take some time depending on your hardware.
- 27. When the installation has completed, confirm it was successful and click **Close** to exit the installation.
- 28. You may close the SQL Server Installation Center window.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

4.6.1.3 Create vCenter Database and ODBC drivers

 Next you will use the Management Studio to create the vCenter database. Click the Start Menu>All Programs>Microsoft SQL Server 2008 R2>SQL Server Management Studio.



2. In the *Connect to Server* window, click on the drop-down for the *Server Name* field and click on **<Browse for more...>.**

Jennect to Server	x
SQL S	Server ⁻ 2008 R2
Server type:	Database Engine
Server name:	
Authentication:	<browse for="" more=""></browse>
User name:	VCENTER \Administrator
Password:	
	Remember password
Connec	t Cancel Help Options >>



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

3. On the *Browse for* Servers window, click the + sign next to **Integration Services** and click the computer name you configured in section 4.6.1.1. Click **OK** to continue.

Browse for Servers	×
Local Servers Network Servers Select the server to connect to:	
Database Engine Analysis Services Reporting Services SQL Server Compact Databases SQL Server Compact Databases Integration Services Integration Services	
OK Cancel Help	

4. On the *Connect to Server* window, change the **Server Type** to **Database Engine**. Click **Connect** to continue.

Connect to Server	Server 2008 R2
Server type:	Integration Services
Server name:	Analysis Services
Authentication:	Reporting Services SQL Server Compact
User name:	Integration Services
Password:	
	Remember password
Conne	ect Cancel Help Options>>

5. Under *Object Explorer* on the left hand side, click the + sign next to **Databases**. Right-click on **Databases** and select **New Database...**



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.cc

Microsoft SQL Server	Managem	ent Studio)		
File Edit V <mark>iew De</mark> bu	g Tools	Window	Community	Hel	
🕴 🔔 New Query 🛛 👔 🔤) 🔁 🛅	🕞 🛛 🚰		Ę	
Object Explorer		-	Ψ×		
Connect 🕶 📑 📑 🔳	7 🛃 🍒				
🗆 🔼 VCENTER (SOL Serv	er 10.50.16	00 - VCENT	ER\Adı	_	
□ □ □ Databases	New Data	oase			
🛨 🧰 Databa	Attach				
E Server Ob	Restore Database				
Replication	Restore Files and Filegroups				
	Start Powe	erShell			
	Reports			•	
	Refresh				

- 6. In the *New Database* window, enter **VCENTER** in the **Database Name** field. Click **OK** to continue.
- 7. Confirm the VCENTER database is listed under Databases on the left hand side.



- 8. Right-click on the VCENTER database and click Properties.
- 9. Under *Select a page* on the left, select **Options**.
- 10. Change the **Recovery Model** to **Simple** via the drop-down box. Click **OK** to continue.
- 11. Exit Microsoft SQL Server Management Studio.



12. Next you need to create the ODBC drivers for VMware vCenter. Click on the **Start Menu>Administrative Tools>Data Sources (ODBC).**



- 13. On the *ODBC Data Source Administrator* window, click on the **System DSN** tab. Click on the **Add...** button.
- 14. On the *Create New Data Source* window, click on **SQL Server Native Client 10.0** source and click **Finish.**



15. On the *Create a New Data Source to SQL Server* window, enter **VCENTER** in the *Name* field. Enter your computer-name you set in section 4.6.1.1 (i.e. VCENTER). Click **Next** to continue.

Create a New Data Soui	rce to SQL Server 🛛 🗙
Monorth on	This wizard will help you create an ODBC data source that you can use to connect to SQL Server. What name do you want to use to refer to the data source?
SQL Server 2008 R2	Name: VCENTER
	How do you want to describe the data source?
	Description:
	Which SQL Server do you want to connect to?
	Server: VCENTER
	Finish Next > Cancel Help

- 16. When asked, "How should SQL Server verify the authenticity of the login ID?" leave the default settings and click **Next** to continue.
- 17. Click the box next to **Change the default database to** and select **VCENTER** from the drop-down. Click **Next** to continue.

reate a New Data Sour	ce to SQL Server	×
SQL Server 2008 R2	 Change the default database to: master model msdb tempdb VENTER Attach database filename: Use ANSI quoted identifiers. Use ANSI nulls, paddings and warnings. 	
	< Back Next > Cancel	Help

18. On the next screen, leave the default settings and click **Finish** to continue.



- 19. On the *ODBC Microsoft SQL Server Setup* window, click the **Test Data Source...** button to check the database.
- 20. On the *SQL Server ODBC Data Source Test* window, confirm the tests completed successfully and click **OK** to continue.
- 21. On the ODBC Microsoft SQL Server Setup window, click **OK** to continue.
- 22. On the *ODBC Data Source Administrator* window, make sure the **VCENTER** data source is listed with the **SQL Server Native Client 10.0** driver. Click **OK** to complete the setup.

4.6.1.4 Install vCenter with SQL Server 2008 R2 database

If you are installing vCenter on a physical machine, you can burn the .iso image to a DVD. Insert the DVD into the CD/DVD drive.

If you are installing to a virtual machine, upload the image file to the ESXi datastore and add the CD to the virtual machine's CD/DVD drive.

1. Click the vCenter Server link.



- 2. If prompted for a security warning, click on **Run.**
- 3. Choose the setup language, *"English"*, and click **OK**.
- 4. On the Welcome page, click **Next**.
- 5. On the End-User Patent Agreement page, click **Next**.
- 6. On the License Agreement page, select **"I agree to the terms in the license agreement"** and click **Next**.

Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

- On the Customer Information page, enter your information. In the License key field, enter the license key assigned to you when you downloaded vCenter Server from the VMware e-academy website and then click Next.
- 8. On the Database Options page, select **Use an existing support database** and select **VCENTER (MS SQL)** from the **Data source Name (DSN)** drop-down box. Click **Next** to continue.
- 9. Confirm the correct DSN and ODBC Driver are selected and click **Next** to continue.
- 10. On the vCenter Server Service page, enter the account password and confirm the password. Click **Next** to continue.
- 11. On the Destination Folder page, leave the default and click **Next**.
- 12. On the vCenter Linked Mode Options page, leave "Create a standalone VMware vCenter Server instance" selected and click Next.
- 13. On the Configure Ports page, leave the defaults and click **Next**.
- 14. On the vCenter Server JVM Memory page, select "Small (less than 100 hosts)" and click Next.
- 15. On the Ready to Install the Program page, click **Install.** The installation will take a few minutes to complete.
- 16. When the installation is complete, click **Finish** to exit the wizard. Leave the VMware vCenter Installer window open.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

4.6.2 Installing vCenter with Microsoft SQL Express (Option 2 and 4)

This section is for vSphere ESXi 4.1 U2 deployments that will <u>not</u> exceed 50 virtual machines.

If you are installing to a virtual machine (Option 2), upload the image file to the ESXi datastore and add the CD to the virtual machine's CD/DVD drive.

If you are installing vCenter on a physical machine (Option 4), you can burn the .iso image to a DVD. Insert the DVD into the CD/DVD drive.

1. Click the **vCenter Server** link.



- 2. If prompted for a security warning, click on Run.
- 3. Choose the setup language, *"English"*, and click **OK**.
- 4. On the Welcome page, click **Next**.
- 5. On the End-User Patent Agreement page, click **Next**.
- 6. On the License Agreement page, select **"I agree to the terms in the license** agreement" and click Next.
- 7. On the Customer Information page, enter your information. In the License key field, enter the license key assigned to you when you downloaded vCenter Server from the VMware e-academy website and then click Next.
- 8. On the Database Options page, make sure "Install a Microsoft SQL Server 2005 Express instance (for small scale deployments)" is selected and click Next.
- 9. On the vCenter Server Service page, leave the "Use SYSTEM Account" check box selected and click Next.
- 10. On the Destination Folder page, leave the default and click **Next**.



- 11. On the vCenter Linked Mode Options page, leave "Create a standalone VMware vCenter Server instance" selected and click Next.
- 12. On the Configure Ports page, leave the defaults and click **Next**.
- 13. On the vCenter Server JVM Memory page, select "Small (less than 100 hosts)" and click Next.
- 14. On the Ready to Install the Program page, click **Install.** The installation will take a few minutes to complete.
- 15. When the installation is complete, click **Finish** to exit the wizard. Leave the VMware vCenter Installer window open.

4.7 Install the vSphere Client on the vCenter Server System

In this task, you install the VMware vSphere[™] Client on the vCenter server.

1. In the VMware vCenter Installer window, click **vSphere Client** to launch the installation wizard.



- 2. If prompted for a security warning, click on Run.
- 3. Choose the setup language, *"English"*, and click **OK**.
- 4. On the Welcome page, click Next.
- 5. On the End-User Patent Agreement page, click **Next**.
- 6. On the License Agreement page, select "I agree to the terms in the license agreement" and click Next.
- 7. On the Customer Information page, enter your information, and then click Next.
- 8. On the Destination Folder page, leave the default and click **Next**.
- 9. On the Ready to Install the Program page, click **Install**. The installation does not take long to finish.
- 10. Click **Finish** when the installation is complete.



4.8 Install vCenter Converter

VMware vCenter Converter can be used to import virtual machines from other VMware production versions. Installation of this component is optional.

1. Click on the vCenter Converter link in the VMware vCenter Installer.



- 2. If prompted for a security warning, click on Run.
- 3. Choose the setup language, "English" and click OK.
- 4. On the Welcome page, click **Next**.
- 5. On the End-User Patent Agreement page, click **Next**.
- 6. On the License Agreement page, select **"I accept the terms in the License Agreement"** and click **Next**.
- 7. On the Destination Folder page, leave defaults and click **Next**.
- 8. On the VMware vCenter Server Information page, set the **Server** to the IP address of the vCenter Server. Set the username to local administrator on the vCenter Server and the password for that account. Leave the **Port** setting to default of port 80. Click **Next**.
- 9. If an untrusted SSL certificate message appears, click **Yes** to continue.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.



- 10. On the VMware vCenter Converter Port Settings page, leave the defaults and click **Next**.
- 11. On the VMware vCenter Converter Identification page, make sure the IP address of your vCenter Server is selected and click on **Next.**
- 12. On the Ready to Install page, click Install.
- 13. When the installation completes, click Finish.
- 14. Click **Exit** to close the VMware vCenter Installer.

4.8.1 Install and Enable the vCenter Converter Plug-in

If you installed vCenter Converter, you may also install the vCenter Converter Plug-in. This will enable you to use vCenter Converter from the vSphere Client.

- 1. Double-click the vSphere Client icon.
- 2. At the vSphere Client login screen, set the username to local administrator on the vCenter Server and the password for that account. Click **Login**.
- 3. In the menu bar of the vSphere Client, select **Plug-ins > Manage Plug-ins.** The Plug-in Manager appears.



- 4. Under **Available Plug-ins**, click the **Download and Install** link, next to the entry for vCenter Converter.
- 5. When the download completes, do the following:
 - a. Choose the setup language, English and click OK.
 - b. On the Welcome page, click **Next.**
 - c. On the End-User Patent Agreement page, click **Next**.
 - d. On the License Agreement page, select **"I accept the terms in the License Agreement"** and click **Next**.
 - e. On the Ready to Install page, click Install.
- 6. Click **Finish** when the installation completes.
- 7. Verify that the vCenter Converter plug-in has a status of **Enabled** in the Plug-in Manager.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

Plug	-in Name	Vendor	Version	Status	Description
Inst	alled Plug-ins				
3	vCenter Storage Monitoring	VMware Inc.	4.1	Enabled	Storage Monitoring and
					Reporting
3	vCenter Hardware Status	VMware, Inc.	4.1	Enabled	Displays the hardware status of
					hosts (CIM monitoring)
8	vCenter Service Status	VMware, Inc.	4.1	Enabled	Displays the health status of
					vCenter services
3	Licensing Reporting Manager	VMware, Inc.	4.1	Enabled	Displays license history usage
8	vCenter Converter	VMware, Inc.	4.2.0	Enabled	Converts physical and virtual
					machines, and backup images to
					VMware virtual machines.

- 8. In the Plug-in Manager, click **Close**.
- 9. Close the vSphere Client window.

4.9 Creating a Virtual Datacenter in vCenter

In this section, you will create a virtual datacenter that will contain your ESXi host systems and virtual machines.

- 1. Open the vSphere Client application.
- 2. Login to the vCenter instance by entering the IP address of the vCenter server. You may enter **localhost** if you are accessing the vSphere client from the same machine where vCenter is installed.

vmware VMware vSphere [™] Client		
To directly manage a single To manage multiple hosts, e vCenter Server.	host, enter the IP address or host i inter the IP address or name of a	name.
IP address / Name:	localhost 🗸 🗸]
User name:	VCENTER-1\vadmin	
Password:		
	 Use Windows session credentia Login Close 	ils Help



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

3. A security warning may appear. Check the box to install the certificate and click **Ignore**.

Security Warning					
Certificate Warnings					
An untrusted SSL certificate is installed on "vcenter-1" and secure communication cannot be guaranteed. Depending on your security policy, this issue might not represent a security concern. You may need to install a trusted SSL certificate on your server to prevent this warning from					
appeaning.					
Click Ignore to continue using the current SSL certificate.					
View Certificate Ignore Cancel					
Install this certificate and do not display any security warnings for "vcenter-1".					

4. Verify that the Hosts and Clusters Inventory view is displayed in vCenter. You should see the label **Hosts and Clusters** in the navigation bar.

🕑 v	CENT	ER -	vSphere (lient			7		
File	Edit	View	Inventory	Admin	istration Pl	ug-ins	Help		
			🛕 Home	▶ 🗗	Inventory		Hosts and Clusters		
ø									
	🦻 VCE	ENTER	ર		VCENT	ER, vc	enter-1 ¥Mware v	Center Server, 4.1.	0, 258902
					Gettin	ig Start	ed Datacenters	Virtual Machines 🔤 H	losts Tasks
					Gettin	ig Start	ed Datacenters	Virtual Machines H	losts Tasks



a. Create a datacenter. **Right-click** your vCenter Server in the inventory, then choose **New Datacenter.**

🚱 VCENTER - 🕫	iphere Client				
File Edit View	File Edit View Inventory Administration Plug-ins Help				
💽 🗈 🔮	Home 🕨 🚮 Inventory	Hosts and	d Clusters		
p 📑					
	New Folder	Ctrl+F	Mware vCenter Server, 4		
	New Datacenter	Ctrl+D	centers Virtual Machines		
	Add Permission	Ctrl+P ▶	iter 🛛 Add a host		
	Open in New Window	Ctrl+Alt+N	nter Server		
	Remove Rename		et up vCenter Server. T nter.		
		_			

b. Enter a datacenter name. The suggested name is **NETLAB**.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

4.10 Create Windows User Account and vCenter Role for NETLAB+

For improved security, we recommend that you create a separate Windows user account and vCenter role that NETLAB+ will use for accessing the vCenter system.

- 1. Create a new Windows user account on the vCenter server system that will be used exclusively by the NETLAB+ system.
- 2. Click on the Start Menu and right-click on Computer. Click on Manage.

	Computer	Open
	Network	Map network drive
	Control Panel	Show on Deskton
	Devices and Pr	Rename
	Administrative	Properties
	Help and Suppo	rt
	Run	
All Programs	Windows Securi	ity
Search programs and files	Log off	
🍂 Start 🕌 🛃 👔		

3. Click on **Configuration->Local Users and Groups->Users** on the left hand side. Right-click in the white space below the listed users and click **New User...** to create a new user.





- a. Enter a unique user name of your choice in the user name field. A secure name is recommended here (not NETLAB).
- b. Enter **NETLAB** in the Full Name field. This will easily identify the account but will not be used as a login credential.
- c. Enter a secure password and confirm.
- d. Check "Password never expires".
- e. Click Create.

New User		? ×
User name:	Enter unique user name here	
Full name:	NETLAB	
Description:		
Password:	•••••	
Confirm password	d: •••••	
🔲 User must ch	ange password at next logon	
🔲 User cannot	change password	
Password ne	ver expires	
🗌 Account is di	sabled	
Help	Create Close	:

- f. After clicking the create button, you may find that the New User form may be blank and the form does not close. The account was created; simply click **Close** and the newly created account should appear in the user list.
- 4. Create a vCenter role for NETLAB+ by cloning the administrator role.
 - a. Open the vSphere client (if not already open from the previous task) and login using administrator account.
 - b. Click **Home** at the top of the vSphere Client.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

c. Select Roles.

🤣 VCENTER - 🕫	phere Client		
File Edit View Ir	nventory Administration) Plug-ins Help)
	Home		
Inventory			
Q	<u>F</u>	Ð	
Search	Hosts and Clusters	VMs and Templates	D
Administration			
6	<u>></u>		
Roles	Sessions	Licensing	Sy
	-		
Management			

- d. Right click on the Administrator role.
- e. Select **Clone** from the context menu.





f. Change the new role name to **NETLAB**.

🕝 VCENTER - vSphere Client
File Edit View Inventory Administration Plug-ins Help
🔄 🗈 🏠 Home 🕨 👯 Administration 🕨 🏭 Roles 🕨 🛃 VCENTER
호* Add Role 🙀 Clone Role
Roles Usage: Clone of Administrator
Name This role is not in use
No access
Read-only
Administrator
Virtual machine power user (sample)
Virtual machine user (sample)
Resource pool administrator (sample)
VMware Consolidated Backup user (sample)
Datastore consumer (sample)
Network consumer (sample)
NETLAB T
· · · ·

- g. Add the NETLAB user created in Windows in task 3.
- h. Click **Home** at the top of the vSphere client.
- i. Click Hosts and Clusters.



j. Right-click on the vCenter instance.



k. Click on Add Permission from the context menu.

🕑 VCENTER - vSp	here Client		
File Edit View In	ventory Administration Plu	ug-ins Help	
🖸 🖸 🏠	Home 🕨 🚮 Inventory	Hosts an	d Clusters
r 🖬			
- 🔁 VCENTER	HERNT	FR	Manaro of
NET 💋	New Folder	Ctrl+F	
ET.	New Datacenter	Ctrl+D	nters
	Add Permission	Ctrl+P	
	Alarm ^M	I	· ·
	Open in New Window	Ctrl+Alt+N	the Ho
	Remove		ding a
	Rename		select

I. Click the **Add** button at the Assign Permissions dialog.

🔗 Assign Permissions		2
To assign a permission to an individu select one or more of the names and Users and Groups These users and groups can interact	al or group of users, d assign a role.	add their names to the Users and Groups list below. Then Assigned Role Selected users and groups can interact with the current
object according to the selected role	e. Propagate	object according to the chosen role and privileges. Read-only Image: All Privileges Image: All Priv
		Description: Select a privilege to view its description
Add	Remove	Propagate to Child Objects



- m. Locate and select the newly created Windows user's account you created for NETLAB+ then click **Add**.
- n. Verify that the user name now appears in the Users field.
- o. Click OK.

🚱 Select Users and Groups		X	
Select users and groups to include in this r Check Names feature to validate your enti	ole. You can also manually enter nam ries against the directory.	es and use the	
Domain: (server)		-	
Users and Groups			
Show Users First		Search	
Name	Description / Full Name	<u>^</u>	
🔒 Guest			
Street Lado (1993)	NETLAB		
SUPPORT_388945a0	SUPPORT_388945a0 CN=Microsoft Corporation,L=Redmond,S=		
Vadmin			
Administrators	Administrators have complete and	unrestrict	
	Backup Operators can override sec	urity restr	
Add			
Users:			
Groups:			
, Note: Separate multiple name	es with semicolons.		
Check Names			
	ок	Cancel	

p. Set the Assigned Role to **NETLAB** (this role was created in the previous task).



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

q. Click **OK**.

🕗 Assign Permissions		
Assign Permissions To assign a permission to an individ select one or more of the names an Users and Groups These users and groups can interac object according to the selected rol Name Role NETLAB	ual or group of users, - d assign a role. t with the current e. Propagate Yes	Assigned Role Selected users and groups can interact with the current object according to the chosen role and privileges. NETLAB Administrator Virtual machine power user (sample) Virtual machine user (sample) Virtual machine user (sample) VMware Consolidated Backup user (sample) Datastore consumer (sample) Network consumer (sample) Network consumer (sample) Network consumer (sample) Net Selder
Add	Remove	Description: Select a privilege to view its description Propagate to Child Objects
Help		OK Cancel

- 5. Exit the vSphere client.
- 6. Verify that the newly created Windows account credentials can be used to login to vCenter.
 - a. Open the vSphere client.
 - b. Uncheck Use Windows session credentials.



c. Login using the user name and password that you specified when creating the new Windows account for NETLAB+.

🚱 VMware vSphere Clie	nt 🔀
vmware VMware vSphere Client	
To directly manage a single To manage multiple hosts, vCenter Server.	e host, enter the IP address or host name. enter the IP address or name of a
IP address / Name:	localhost 👻
User name:	Intel (Intel (In
Password:	****
[Use Windows session credentials

If you can login to vCenter without an error, you have successfully created a new Windows account and vCenter role for NETLAB.

If you cannot login using the new Windows account user name and password, please recheck all setup tasks from this section.

The user name and password for this account will be registered in NETLAB+ in the next task. Henceforth, the user name and password for the NETLAB account should only be used by NETLAB+ automation; it should not be used for interactive vCenter logins (except for troubleshooting purposes).



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrour

4.11 Registering a Virtual Datacenter in NETLAB+

In this section, you will add vCenter datacenter(s) created in the last section to NETLAB+.

- 1. Login to the NETLAB administrator account.
- 2. Select Virtual Machine Infrastructure.
- 3. Select the Virtual Datacenter and Management Agents option.
- 4. Click the Add Datacenter button.
- 5. Enter the required information for the datacenter you have set up using the vSphere Client in the previous task. Field descriptions are provided below.

The values for this form may vary based on your local settings.

Virtual Datacenters and Management Agents			
Admin Logout			
Add Datacenter			
Datacenter Name	NETLAB	(required)	
Agent Hostname		(required)	
Agent Username		(required)	
Agent Password		(required)	
	shov	v help tips 📄	
🕜 Add Datacente	r 🛛 🔀 Cancel		

a. **Datacenter Name:** The exact name of the **datacenter** as registered in vCenter (see the red box below). This will be **NETLAB** if you used the recommended name when adding a datacenter in vCenter.







c. Agent Hostname: This value is set to an IP addresses assigned to your vCenter server. The address you should use depends on the networking model you have chosen and will influence the path (outside or inside) NETLAB+ will use to connect to the vCenter server. The following table depicts which IP address you should use based on the model you selected.

Network Model	Management Path	IP Address
Single-Homed	OUTSIDE	Campus LAN IP
Dual-Homed	OUTSIDE	Campus LAN IP
Secure+	INSIDE	169.254.0.253

- d. **Agent Username:** Enter the username for the account created in section 4.10. This username will be used when NETLAB+ connects to vCenter.
- e. **Agent Password**: Enter the password for the account created in section 4.10. This password will be used when NETLAB+ connects to vCenter.
- 6. Click Add Datacenter to complete the registration.
- 7. Click the **Test** button to verify that NETLAB+ can connect to vCenter server using the settings you provided. If the test fails, recheck connectivity and your datacenter settings.

4.12 Setting the Database Retention Policy

The purpose of this section is to prevent the vCenter database from filling the hard disk with unnecessary information.

Not changing this setting has shown that database sizes grow exponentially and eventually cause vCenter to slow down and even become non-responsive.

- 1. Open the vSphere client and login using administrator account.
- 2. Click on the Administration menu at the top and select vCenter Server Settings.

🕝 YCENTER - vSphere Client				
File Edit View Inventory Ad	ministration Plug-ins Help		_	
🕞 🕞 🔥 Home	Custom Attributes		Г	
	vCenter Server Settings		H	
	Role	×.		
	Session	×	æ,	
	Edit Message of the Day		'irt	
	Export System Logs			
	Export vCenter Converter Logs		ŕ	

3. Click on **Database Retention Policy** on the left hand side. Click the box next to **Tasks retained for** and enter a preferred number of days (NDG recommends 30



days). Click the box next to **Events retained for** and enter a preferred number of days (NDG recommends 30 days). Click **OK** to save settings.

Database Retention Polic How long should tasks an	c y Id events be retained in vCenter database?	
Licensing Statistics Runtime Settings Active Directory Mail SNMP Ports Timeout Settings Logging Options Database Database Database SSL Settings Advanced Settings	Tasks retained for 30 days Vents retained for 30 days Use these options to limit the growth of the database. Do not use these options if you want to maintain a complete history of vCenter tasks and events.	
Help	ОК	Cance

4. Exit the vSphere Client.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.net

5 Adding ESXi Hosts to vCenter and NETLAB+

In this section, you will add your ESXi hosts to vCenter, and then register them in NETLAB+.

Repeat the steps in this section for each ESXi host.

5.1 Adding ESXi hosts to vCenter

In this task you will add the ESXi host servers into vCenter Server for management and license the servers using the license key you obtained in section 2.4.

- 1. Log in to your vCenter Server system.
- 2. Double-click the vSphere Client icon.
- 3. At the vSphere Client login screen, set the host name as **localhost**, and then enter the username and the password. Click **Login**.
- 4. Verify that the Hosts and Clusters Inventory view is displayed. You should see the label **Hosts and Clusters** in the navigation bar.



- 5. Add your ESXi host to the datacenter:
 - a. Right-click the on the datacenter (NETLAB), then choose **Add Host**. The Add Host wizard appears.
 - b. When prompted by the wizard, enter the following values. Click **Next** to continue.

Fields	Values
Host	First ESXi host machine IP address
Username	Root
Password	user configured password used during host setup (section 3.4)


c. When the Security Alert window appears, select **Yes**.

Security	Security Alert 🛛 🔀		
2	Unable to verify the authenticity of the specified host. The SHA1 thumbprint of the certificate is:		
	50:78:5D:65:52:5F:06:14:2C:F9:2A:C8:D4:E7:77:54:07:68:5E:45		
	Do you wish to proceed with connecting anyway?		
	Choose "Yes" if you trust the host. The above information will be remembered until the host is removed from the inventory.		
	Choose "No" to abort connecting to the host at this time.		
	Yes No		

- d. On the Host Information window, view the information and then click **Next**.
- e. On the Assign License window, select Enter Key.
- f. Enter the key you received from VMware in section 2.4.
- g. Click Next.
- h. On the Configure Lockdown Mode window, leave **Enable Lockdown Mode** unchecked. Click **Next**.
- i. On the Virtual Machine Location window, click on your **NETLAB** datacenter. Click **Next.**
- j. On the Ready to Complete window, review the information given and select **Finish** to add the host.
- k. In the **Recent Tasks** pane at the bottom of the vSphere Client window, monitor the progress of the task.
- I. After the task is completed, maximize the **NETLAB** datacenter and verify that your ESXi host appears in the inventory.
- 6. Repeat these steps for your other ESXi host servers.

After registering an ESXi host in vCenter, it should be managed through vCenter. You should no longer connect directly to an ESXi host from the vSphere client.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

5.2 ESXi Host Virtual Switches

A virtual switch (vSwitch) on the physical ESXi host bridges between physical networks, virtual machines, and the ESXi host kernel. Each vSwitch is an internal LAN, implemented entirely in software by the ESXi kernel.



Your ESXi host(s) may connect to the outside network, inside network, or both depending on the network model you are using. The following table indicates which virtual switches are used for outside and inside connections.

Network Model	OUTSIDE vSwitch	INSIDE vSwitch
Single-Homed	vSwitch0	
Dual-Homed	vSwitch0	vSwitch1
Secure+		vSwitch0



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

5.3 Verifying vSwitch0 Configuration

vSwitch0 is automatically created during the ESXi software installation (section 3.5). Using vCenter, confirm that networking on vSwitch0 is properly configured (refer to the red numbered items in the screen below):

- 1. vSwitch0 is bound to the correct physical NIC (vmnic).
- 2. The physical NIC is connected and with correct speed/duplex.
- 3. The VMkernel port has the IP address you assigned when configuring your ESXi host. The IP address should be one of the following:
 - a. A campus LAN address if your ESXi host connects to the outside.
 - b. An address in the range of 169.254.0.241 to 169.254.0.249 if your ESXi host only connects to the inside network.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrc

5.4 Inside Network Configuration

In this section, you will perform the final setup of ESXi host inside networking. This section only applies to networking configurations that connect the ESXi host(s) to the inside network (see table below). This section describes various ESXi host networking components. We recommend reviewing this section even if inside networking is not used in your ESXi host configuration.

Networking Configuration	Inside Networking
Single-Homed Networking	No
Dual-Homed Networking	Yes
Secure+ Networking	Yes

Repeat the setup tasks in this section for each ESXi host in your NETLAB+ / vCenter.





Three types of network traffic can flow across the ESXi inside network connection depending on the networking configuration.

- Management traffic between vCenter and ESXi (VLAN1)
- Remote display traffic (VLAN1)
- Remote PC traffic between virtual machines and real equipment (VLANs 100 -899)

The following diagram and table describes the various components of inside networking and will be referenced in later sections.



#	Component	Description	
1	ESXi Host	The physical server where, your virtual machines run.	
2	Inside Physical NIC	The physical network interface on the ESXi Host (1) that connect virtual machines to the inside physical network.	
3	vSwitch	A virtual switch on the physical ESXi host that bridges between physical networks (2,8,9), virtual machines (7), and the ESXi host kernel (4). Each vSwitch is an internal LAN, implemented entirely in software by the ESXi kernel.	
4	Kernel Port	A virtual network interface on the ESXi host (1) that provides connectivity between the ESXi host kernel and other components such vCenter (11).	



5	Virtual Network Adapter (vNIC)	A virtualized networking adapter inside of a virtual machine that connects the virtual machine to a virtual switch.	
6	Port Groups	A template for creating virtual network switch ports with a particular set of specifications. A port group allows a virtual network adapter (5) to be placed in a particular virtual LAN (VLAN). Port groups with specific VLAN IDs to connect virtual machines to real equipment.	
7	Virtual Machines	In NETLAB+, a <i>virtual machine</i> is a remote PC or remote server that runs on virtualized hardware. Although the hardware is virtualized, real operating systems and real application software can still be used.	
8	Uplink / Trunk	An uplink is a physical connection between ESXi Host (1,2) and a NETLAB+ control switch (9). If you are interfacing with real equipment pods (i.e. Cisco Networking Academy), your ESXi inside physical interface and the control switch port to which it is connected are configured in 802.1q trunk mode. Trunks allow multiple virtual LANs (VLANs) to exist on a single physical connection. VLAN assignments and the VLAN database on the control switch are managed by NETLAB+.	
9 Control Switch		A NETLAB+ control switch provides connectivity between the NETLAB+ server, ESXi host servers, vCenter server, asynchronous access servers, and switched outlet devices. Control switches are not accessed by lab users.	
		An NDG supported control switch is required. See the NDG website for a list of supported control switches.	
10 NETLAB+ Inside Connection		The NETLAB+ server inside interface connects to a designated reserved port on a control switch (9). The fixed addresses 169.254.0.254/24 and 169.254.1.1/24 are assigned to the inside interface (these cannot be changed).	
		802.1q trunk mode should NOT be enabled on the control switch port for this connection.	
11 vCenter Server Inside Connection		If you are using a physical server for vCenter, you server will connect to a designated reserved port on a control switch (9).	
		802.1q trunk mode should NOT be enabled on the control switch port for this connection.	
12	Real Equipment Pods	Real lab equipment (optional) is connected to one or more control switches (9).	



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

The following table summarizes the traffic types that will flow over the ESXi inside network.

Networking Configuration	Management Traffic (VLAN 1)	Remote Display (VLAN 1)	802.1q Trunk
Single-Homed Networking	n/a	n/a	n/a
Dual-Homed Networking	No	No	Real Gear*
Secure+ Networking	Yes	Yes	Real Gear*

* ESXi interface and corresponding control port is configured as 802.1q trunk when interfacing with real equipment.

5.4.1 Creating vSwitch1 and Binding to Physical NIC

If your ESXi host is dual-homed (connected to both outside and inside networks), you must create an inside virtual switch (vSwitch 1), bind a physical NIC to vSwitch1, and create a VMkernel port for management traffic. These tasks are performed through vCenter.

- 1. Login to vCenter using the vSphere client.
- 2. Navigate to Home > Inventory > Hosts and Clusters.
- 3. Click on the ESXi host to configure in the left sidebar.
- 4. Click on the **Configuration tab**.
- 5. Click on **Networking** in the Hardware group box.
- 6. Click on the Virtual Switch view button if not already selected.
- 7. Click on Add Networking.



8. Add a VMkernel port to allow the ESXi host kernel to communicate with the inside network. Select the **VMkernel** radio button, then click **Next**.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

🔗 Add Network Wizard	
Connection Type Networking hardware	can be partitioned to accommodate each service that requires connectivity.
Connection Type Network Access Connection Settings Summary	Connection Types Virtual Machine Add a labeled network to handle virtual machine network traffic. VIMkernel The VMkernel TCP/IP stack handles traffic for the following ESXi services: VMware VMotion, ISCSI, NFS, and host management.
Help	≤ Back Next ≥ Cancel

Add Network Wizard				
VMkernel - Network A The VMkernel reacher	c cess s networks through uplink adapters attac	ched to virtual s	witches.	
Connection Type Network Access	Select which virtual switch will have	dle the network ters listed belov	traffic for this connection. You may also cre 	ate a new virtual switch
 Connection Settings Summary 	Create a virtual switch	Speed	Networks	
Summary	Vmnic1	1000 Full	None	
	🗌 📟 vmnic2	1000 Full	None	
	O Use vSwitch0	Speed	Networks	
	vmnic0	1000 Full	10.0.0.1-10.0.0.254	
	Preview:			
	-VMkernel Port 	Q.	hysical Adapters The second se	
<u>H</u> elp			<u>≤</u> Back I	Next ≥ Cancel

- 9. Select the **"Create a virtual switch"** radio button. The new switch will be named vSwitch1.
- 10. Select the physical NIC that will connect vSwitch1 to the control switch. We recommend using vmnic1 for inside connections (vmnic0 should already be connected to the outside network).



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

🕝 Add Network Wizard			
VMkernel - Connection S Use network labels to id	settings Jentify VMkernel connections while r	managing your hosts and datacenters.	
Connection Type Network Access Connection Settings IP Settings Summary	Port Group Properties	VMkernel None (0) Use this port group for vMotion Use this port group for Fault Tolerance logging Use this port group for management traffic	
	Preview: 	Physical Adapters	
Help		≤Back Next ≥	Cancel

- 11. Enter the port group properties as shown above.
 - a. Network Label: "VMkernel" (default)
 - b. VLAN ID: None(0) (default)
 - c. Check option "Use this port group for management traffic"
- 12. Click Next.
- 13. The VMkernel IP Connection Settings dialog appears (see next page).

Add Network Wizard		
VMkernel - IP Connectio Specify VMkernel IP set	n Settings Itings	
Connection Type Network Access Connection Settings IP Settings Summary	Obtain IP settings automatically Obtain IP settings: IP Address: Subnet Mask: Z55 . 255 . 0 VMkernel Default Gateway: Default Gateway:	11 to 249 Edit
	VMkernel Physical Adapters VMkernel 169.254.0. ■ Image: I	Cancel



Incido Interfaco	IP Address	Subnet Mask
inside interface	IF Address	Sublict Wash
ESXi Server 1 Inside	169.254.0.241	255.255.255.0
ESXi Server 2 Inside	169.254.0.242	255.255.255.0
ESXi Server 3 Inside	169.254.0.243	255.255.255.0
ESXi Server 4 Inside	169.254.0.244	255.255.255.0
ESXi Server 5 Inside	169.254.0.245	255.255.255.0
ESXi Server 6 Inside	169.254.0.246	255.255.255.0
ESXi Server 7 Inside	169.254.0.247	255.255.255.0
ESXi Server 8 Inside	169.254.0.248	255.255.255.0
ESXi Server 9 Inside	169.254.0.249	255.255.255.0

14. Enter a unique inside IP address and subnet mask from the following table.

No changes to the VMkernel Default Gateway setting should be necessary. If your server has an outside connection, this should already be set to the default gateway on your campus LAN. If your server only has an inside connection, the gateway should be 0.0.0.0 (not set); there is no off-net router/routing on the inside network by design.

15. Click **Next** to continue.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

16. Confirm that vSwitch1 appears as follows (IP varies for each host).

- a. VMkernel port (vmk1) has correct IP address.
- b. vSwitch1 is bound to physical adapter (vmnic1)
- c. Physical adapter is up (speed and duplex are detected)

Virtual Switch: vSwitch1		Remove Properties
P	VMkernel Port VMkernel	Physical Adapters
	vmk1: 169.254.0.241	

A × mark displayed near the Physical Adapter indicates that the connection has not yet been physically cabled or the corresponding control switch port is shut down.

firtu	al Switch: vSwitch1	Remove Properties
þ	VMkernel Port VMkernel vmk1: 169.254.0.241	Physical Adapters
		Interface Down



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

5.4.2 Configuring Control Switch 802.1q Trunk Ports

To allow virtual machines to communicate with real equipment pods, configure 802.1q trunk mode on all control switch ports connecting to your ESXi inside vmnics. In addition, all uplinks between control switches must also be 802.1q trunks.

You must console into the control switches to perform this action. The control switch console password is **router**. The enable secret password is **cisco**. These passwords are used by NETLAB+ automation and technical support - please do not change them.





After you have configured the control ports for ESXi inside host connections and interswitch uplinks, verify that the ports are up and operating in trunk mode.

1. Verify each ESXi host control port and inter-switch uplink is connected and line protocol is up. Substitute the designated interfaces names for x/x.

```
netlab-cs1# show interface x/x
x/x is up, line protocol is up (connected)
```

2. Verify each ESXi host control port and inter-switch uplink is operating in 802.1q trunk mode.

netlab-cs1# show interface trunk								
Port	Mode	Encapsulation	Status	Native vlan				
x/x	on	802.1q	trunking	1				
x/x	on	802.1q	trunking	1				
x/x	on	802.1q	trunking	1				



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

5.4.3 Connecting Virtual Machines to Real Equipment Pods

This section discusses the configuration of networks to communicate between virtual machines and real lab devices in the topology. You can skip this section if your virtual machines do not need to communicate with real lab equipment.



Virtual machines [7] running on an ESXi host talk to real equipment pods [12] using port groups [6], the inside virtual switch [3] and control switches [9]. Inside networking is always used for this purpose. In the last two sections (5.4.1 and 5.4.2), you established an inside network connection and configured 802.1q VLAN trunking on the link connecting to your control switch [8].

A real equipment pod may have one or more networks. 802.1q virtual LANs (VLANs) are the glue that binds virtual machines [7] and real equipment [12] with the proper pod networks. A unique set of VLAN identifiers for each pod is automatically allocated by NETLAB+ and programmed into the control switches [9] by NETLAB+ when the pod is created. Port groups [6] are assigned to a specific VLAN ID, thereby allowing virtual machine network adapters [5] to be placed in a specific VLAN.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

As of version 2011.R2, NETLAB+ will automatically setup and teardown VLAN based port groups on the inside vSwitch on NDG standard pods.

Automatic network setup on NDG standard pods occurs when a pod is reserved. Automatic network teardown on NDG standard pods occurs when a reservation completes. These features are enabled by default but can be disabled on a per pod basis.

Refer to Appendix A for pods that require manual networking, such as custom pods.

5.4.3.1 Creating a Real Equipment Pod

Creating a real equipment pod in NETLAB+ should be done first. This will automatically generate the required number of VLANs for the selected pod type, and add those VLANs to the control switches. This should be done before ESXi host networking is configured.

- 1. Login to the NETLAB+ administrator account.
- 2. Select Equipment Pods.
- 3. Click the **Add a Pod** button at the bottom of the page.
- 4. Select the desired pod type. Only pod types that use both real lab equipment and remote PCs are relevant to this section. Our examples use the Multipurpose Academy Pod, which contains 3 routers, 3 switches, and 3 remote PCs.
- 5. Complete the **New Pod Wizard.** Please refer to the <u>NETLAB+ Administrator</u> <u>Guide</u>, NDG website pod specific web pages, and NDG pod guides for pod specific installation instructions.

After the pod is created, you will be placed in the Pod Management page. You will notice that all virtual machines are initially ABSENT. You will add virtual machines to the pod later.

5.4.3.2 Determining the Base VLAN and VLAN Pool

This is now an automated task for NDG standard pods as NETLAB+ version 2011.R2. Refer to Appendix A for manual networking setup guidance for custom pods or pods that do not support automatic networking.

5.4.3.3 Creating Port Groups for Pod VLANs on the Inside Network

This is now an automated task for NDG standard pods as NETLAB+ version 2011.R2. Refer to Appendix A for manual networking setup guidance for custom pods or pods that do not support automatic networking.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

5.4.3.4 Increasing the Inside vSwitch Port Count

By default, a vSwitch is provisioned with 56 virtual ports. This means that 56 virtual network adapters can be connected to the virtual switch, regardless of which port group the adapter is connected. For most NETLAB AE setups, this is sufficient for the inside vSwitch. For large NETLAB PE setups or systems with custom real equipment pods, you may need to increase this value to accommodate more virtual machine connections to the inside vSwitch.

This discussion applies to virtual machines that are part of real equipment pods. Pods that contain only virtual machines are usually placed on separate vSwitches that do not connect to the inside network / real equipment.

To calculate the number of inside vSwitch ports required on a particular ESXi host, add up the number of virtual machines in real equipment pods that are assigned to the host. This is the number of virtual ports required on the inside vSwitch (assuming one connection per VM). If this number exceeds 50, you should select the next highest port count setting (120). In special cases, an even higher setting may be required. Note: 50 is not an error; 6 extra ports were subtracted (from 56) to allow for VMkernel ports and other possible connections. Higher port count settings consume additional host resources, so you should set this value to the lowest possible setting that provides enough ports for every virtual machine connecting to the inside vSwitch.

There is no warning when the number of virtual ports is exceeded and the problem is not obvious. Some of the virtual machines will fail to communicate for no apparent reason. The only clue may be a disconnected network status from the guest operating system.

The following procedure is used to increase the number of virtual ports on the inside virtual switch. The same procedure can be used on any virtual switch should the need arise.

- 1. Login to vCenter using the vSphere client.
- 2. Navigate to Home > Inventory > Hosts and Clusters.
- 3. Click on the ESXi host where the pod's virtual machines will run.
- 4. Click on the **Configuration tab**.
- 5. Click on **Networking** in the Hardware group box.
- 6. Click on the **Virtual Switch** view button if not already selected.
- 7. Click **Properties** on the INSIDE vSwitch. The inside vSwitch is the one that is connected to the control switch (typically vSwitch0 if single homed, vSwitch1 if dual homed).



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

🕜 VCENTER - vSphere Clie	ent 1		
<u>F</u> ile <u>E</u> dit Vie <u>w</u> I <u>n</u> vento	ry <u>A</u> dministration <u>P</u> lug-ins <u>H</u> elp		
🖬 🗈 🏠 Home	Mosts and Cluster	·s 2	Search Inventory
	10.0.0.38 VMware E5Xi, 4.0.0, 208167 Getting Started Summary Virtual Mac	hines Resource Allocation Perform	ance Configuration Tasks & Events Alarms Permission:
÷ 10.0.0.38	Hardware	View: Virtual Switch Distribut	ed Virtual Switch
	Processors	Networking 6	Refresh Add Networking Properties
	Memory Storage Networking 5 Storage Adapters Network Adapters Advanced Settings	Virtual Switch: vSwitch0 Virtual Machine Port Group TDA 10.0.0.0 NET Vilkernel Port Vilkernel Network	Remove Properties Physical Adapters wmnico 0000 Full
	Software	vmk0 : 10.0.0.38	
	Licensed Peatures Time Configuration DNS and Routing Power Management Virtual Machine Startup/Shutdown	Virtual Switch: vSwitch1	Remove Properties

- 8. Click on the **vSwitch** configuration item.
- 9. Click the **Edit** button.

1	- 1	0.11.0		
Configuration	Summary	VSwitch Properties		
🕼 vSwitch	56 Ports	Number of Ports:	56	
👷 Management Net	VMotion and IP	Default Policies		
		Security		
		Promiscuous Mode:	Reject	
		MAC Address Changes:	Accept	
		Forged Transmits:	Accept	
		Traffic Shaping		
		Average Bandwidth:		
		Peak Bandwidth:		
		Burst Size:		
		Failover and Load Balancing		
		Load Balancing:	Port ID	
		Network Failure Detection:	Link Status only	
		Notify Switches:	Yes	
		Failback:	Yes	
		Active Adapters:	vmnic0	
		Standby Adapters:	None	
Add	Edit Remov	Unused Adapters:	None	



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.cc

General Security Traffic Shaping NI	C Teaming
Number of Ports:	56 💌
🔶 Changes will not take effect until	24
	120 248 504 1016 2040 4088<
USE LOWEST VALUE THAT WIL	L ACCOMODATE INSIDE VIRTUAL MACHINES
	OK Cancel Help

10. Increase the number of ports, but not higher than needed.

11. Click **OK**.

The ESXi host must be restarted for the change to take effect. Make sure there are no active NETLAB+ reservations that affect this host, or virtual machines running on this host as user work may be affected.

- 12. **Right click** on the ESXi host in the left sidebar to activate the context menu.
- 13. Select Reboot.





5.5 Creating a Safe Staging Network

Most virtual machines in your pods will typically be equipped with one or more virtual network adapters, which are used to communicate with other peer devices in a pod, such as other virtual machines or real networking equipment (Cisco Netacad).

Virtual switches and port groups form networks that interconnect virtual machines and real equipment. These networks are created on each ESXi host in one of two ways:

- Automatic Networking. Many NETLAB+ virtual pods support automatic networking. When a pod is activated, NETLAB+ will create the necessary virtual switches and port groups, and bind the virtual network adapters automatically.
- **Manual Networking.** For pods that do not support automatic networking, the VMware administrator will create virtual switches and port groups, and bind the virtual network adapters using the vSphere client.

When creating new virtual machines, the vSphere client will require and prompt for an existing network to place the virtual machine. If a network has not been created yet using automatic or manual networking, this creates a dilemma: on what network should I place my virtual machine in the mean time? The default value is usually the network connecting to your campus LAN. This is a potential security risk and should be avoided, unless you want the VM to have temporary access to the campus network or Internet for the purpose of software installation and patches.

A *safety net* is a safe staging network where you can temporarily connect your VMs. It consists of a virtual switch and port group that is not connected to any other networks (virtual or real). Should the virtual machine be powered, its traffic will be confined to the safety net. This ensures that the virtual machine will not pose a security risk to your campus LAN or interfere with other pods, until it is relocated to its final network via automatic or manual networking.

To create the safety net on each ESXi host, repeat the following steps:

- 1. Login to vCenter using the vSphere client.
- 2. Navigate to Home > Inventory > Hosts and Clusters.
- 3. Click on the ESXi host to configure in the left sidebar.
- 4. Click on the **Configuration tab**.
- 5. Click on **Networking** in the Hardware group box.
- 6. Click on the Virtual Switch view button if not already selected.
- 7. Click on Add Networking.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.co

2 VCENTER - vSphere Client 1		
<u>F</u> ile <u>E</u> dit Vie <u>w</u> I <u>n</u> ventory <u>A</u> dministration <u>P</u> lug-ins <u>H</u> elp		
💽 💽 🏠 Home 🕨 🚮 Inventory 👂 👘 Hosts and Clusters	2	🖅 - Search Inventory 🔍
Image: With the second secon	ines Resource Allocation Performance Configu	ration Tasks & Events Alarms P(
+ 10.0.0.38 3 Hardware	View: Virtual Switch Distributed Virtual Switch	h 4
Processors	Networking 6 Ref	resh Add Networking Properties
Memory		7
Networking 5	Virtual Switch: vSwitch0	Remove Properties
Storage Adapters	-Virtual Machine Port Group TDA 10.0.00 NET	ical Adapters wmnic0 1000 Full □
Advanced Settings	Management Network	
Software	vmk0 : 10.0.0.38	
Licensed Features		

8. Select the Virtual Machine connection type and click Next.

🕜 Add Network Wizard	
Connection Type Networking hardware	can be partitioned to accommodate each service that requires connectivity.
Connection Type Network Access Connection Settings Summary	Connection Types Virtual Machine Add a labeled network to handle virtual machine network traffic.
	C VMkernel The VMkernel TCP/IP stack handles traffic for the following ESXi services: VMware VMotion, ISCSI, NFS, and host management.
Help	Cancel

- 9. Select the Create New Virtual Switch radio button.
- 10. **UNCHECK** any vmnics under the radio button; this virtual switch should not be bound to any real network adapters on the ESXi host.

Connection Type	Select which virtual switch will handle the	network traffic for this connec	ction. You may also create a new virtu	al switch
Connection Settings Summary	Create a virtual switch	Speed 100 Full	Networks 192. 168. 1.4-192. 168. 1.4	
	O Use vSwitch0	Speed 1000 Full	Networks 10.0.0.1-10.0.0.254	
	O Use vSwitch1	Speed 100 Full	Networks None	
	C Use vSwitch2	Speed 1000 Full	Networks None	
	Preview: Virtual Machine Port Group Virtual Machine Network	Physical Adapters		



11. Enter "SAFETY NET" in the port group's Network Label property.

12. Click Next.

🙆 Add Network Wizard						
Virtual Machines - Connec Use network labels to ider	tion Settings ntify migration compatible connect	ions common to two or more hosts.				
Connection Type	Port Group Properties					
Connection Settings	Network Label:	SAFETY NET				
Summary	VLAN ID (Optional):		▼			
	Preview:			1		
	-Virtual Machine Port Group	Physical Adapters				
<u> </u>	9					
			<u>≤ Back</u> Next ≥	Cancel		

13. Confirm that a new vSwitch and port group named **SAFETY NET** appears in the ESXi host's network configuration.

Virtu	ual Switch: vSwitch6	Remove	Properties.
P	Virtual Machine Port Group — SAFETY NET	Physical Adapters No adapters	

SAFETY NET may now be used as a safe temporary network location for new virtual machines. New virtual machine creation will be discussed in section 7.

Tip. The safety network is an ideal place to bind the network interface(s) of master virtual machines. Automatic networking will the bind network interfaces of cloned VMs to their runtime networks when their respective pods are started.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

5.6 Adding ESXi hosts in NETLAB+



In this section, we will add the virtual machine host servers in your datacenters to NETLAB+.

- 1. Login to the NETLAB+ administrator account.
- 2. Select Virtual Machine Infrastructure.
- 3. Select Virtual Machine Host Servers.
- 4. Click Add Host.
- 5. If you have more than one datacenter defined, you will be prompted to select a datacenter. NETLAB+ scans the datacenter to discover hosts. Hosts that have not been registered in NETLAB+ are displayed. Select the host you wish to add by clicking on the host name.

١	/irtual Mach	ine Host S	arvers				_	_
A	Admin Logout							
New Virtual Machine Host								
Select the host from datacenter that you wish to add.								
	Host Name	Vendor	System	CPU Mod	el	# CPU	Memory	Connection State
	10.0.0.139	VMware, Inc.	VMware Virtual Platform	Intel(R) Xeon(R) CPU E5	620 @ 2.40GHz	2	2 GB	CONNECTED
	🔀 Cancel							
	/irtual Machi	ine Host Ser	vers					
ŀ	Admin Logou	ıt				_		
	Edit Host 10.0.0).38						
		Host Name	10.0.0.38			E ADD	RESSE	3
		Datacenter Name	• NETLAB					
	Οι	utside IP Address	10.0.0.38					
	1	nside IP Address	169.254.0.241	•				
	Insi	de vSwitch Name	vSwitch1					
	Con	nmunication Path	• outside network					
			inside network					
	Proactive Res	ource Awareness	🛛 🔲 enable feature on	this host				

6. Enter the settings for this host based on the networking model you have chosen. The table below shows typical settings.

show help tips

Update

🔀 Cancel



Networking Model	Outside Address	Inside Address	Inside vSwitch	Communication Path
Single-Homed	Campus LAN IP address	Not set	Not applicable	Outside
Dual-Homed	Campus LAN IP address	169.254.0.X	vSwitch1	Outside
Secure+	Not set	169.254.0.X	vSwitch0	Inside

Host Name. The IP address or fully qualified domain name of the ESXi host. This should be the same value entered in vCenter for the host name.

Outside IP Address. The IP address of the ESXi host outside interface. Leave blank for the Secure+ network model.

Inside IP Address. The IP address of the ESXi host inside interface. Use the "not set" option for single-homed networking.

Communication Path. This setting determines the network path that is used for remote display connections (proxied by the NETLAB+ server).

If you choose inside, the Inside IP Address setting must be valid (i.e. you have completed the tasks in section 5.4.1).

Inside vSwitch Name. This is the name of the virtual switch that connects to your control switch as shown in section 5.4. This is typically "vSwitch1" for dual homed networking and "vSwitch0" for Secure+. Leave this setting blank for single-homed networking (which does not connect to a control switch).

The inside virtual machine name is case sensitive and must be entered exactly as shown in vCenter. This setting must be correct to use automatic networking on real equipment pods (i.e. MAP, CRP, BRPv2).

Proactive Resource Awareness. See the next section for details.



5.7 **Proactive Resource Awareness**

Proactive Resource Awareness (PRA) allows you to time-share virtualization servers using the NETLAB+ scheduler. PRA is designed with 3 goals in mind.

- 1. Ensure a quality lab experience for all trainees by reserving CPU and memory resources on VM servers at the scheduled lab times. This proactively prevents the VM servers from becoming overloaded and unresponsive.
- 2. Increase the number of trainees that you can run through your training programs by spreading out the trainee lab sessions over time.
- 3. Reduce costs of your training programs by using fewer virtualization servers.

For each VM server in your inventory, you may set 3 limits.

- Total number of active virtual machines
- Active number of virtual CPUs
- Active maximum memory usage

With these limits defined, the scheduler will proactively manage the servers CPU and memory resources. If scheduling a particular pod would exceed one of the set limits in a 30 minute time slot, the pod cannot be scheduled at that time and the limitation will be clearly indicated on the scheduler.

PRA is enabled in Virtual Machine Host Servers on a per host basis. You can enable PRA when adding a host or editing the host settings of an existing host.

Administrator Account > Virtual Machine Inventory > Virtual Machine Host Servers

The values you set for PRA will depend on your host server specifications and the types of workloads that you run on your servers.

Proactive Resource Awareness is a scheduling algorithm based on workload forecasting; it does not monitor real time workload. PRA assumes that all virtual machines running on a host server are under the control of NETLAB+; it does not account for VMs that are powered on independently of NETLAB+.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.co

Virtual Machine Host Server	S
Admin Logout	
Edit Host	
Host Name	10.0.0.38
Datacenter Name	10x40
Outside IP Address	10.0.0 38
Inside IP Address	inat sati
Communication Path	eutode network
	C made refuert
Proactive Resource Awareness	✓ enable feature on this host
Maximum Running VMs	enable this limiter
	40 virtual machines (max)
Maximum Virtual CPUs	v enable this limiter
	64 virtual CPUs (max)
Maximum Memory Usage	✓ enable this limiter
	73600 Megabytes (max)
	show help tips
🚱 Update 🛛 🔀 Cancel	

- **Proactive Resource Awareness.** Check this box to enable PRA on this host. Uncheck the box to completely disable PRA on this host.
- Maximum Running VMs. To set the maximum number of virtual machines that can be scheduled at one time on this host, check enable this limiter and set the number of virtual machines.
- Maximum Virtual CPUs. VMware ESXi supports VMs with more than one virtual CPU (vCPU) and Symmetric Multiprocessing. Virtual machines with more than one vCPU typically use more processing power on the host than VMs with only one vCPU. The number of vCPUs assigned to a VM can be seen when viewing the Virtual Machine Inventory table in NETLAB+. By setting Maximum Virtual CPUs, you can achieve more granular control over CPU resources. Check enable this limiter and set the number of virtual CPUs that can be running at one time.
- Maximum Memory Usage. This setting limits the amount of memory (in Megabytes) that can be scheduled for VMs at one time. It is based on the memory assigned to the virtual machines when the VM was created. It does not include overhead memory required for the VMs or the host. For best performance, this value should not be set higher than the physical memory on the host. If you are setting up a host server for the VMware ICM v5.0 course, number should be set to 122880 Megabytes.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

6 vCenter Update Manager

Depending on the installation method you selected in section 4.6, please choose the appropriate sub section below.

You will need to perform updates if you installed an older version of ESXi on your host machines (i.e. ESXi 4.1 or ESXi 4.1 U1).

NETLAB+ requires you to use ESXi 4.1 U2. This section will assist in making sure you are at the appropriate version.

VMware vCenter Update Manager requires Internet access to update ESXi hosts. If you do not have Internet access on your vCenter Server machine, there are ways to download the patches from VMware.

Refer to the <u>VMware vCenter Update Manager Installation and Administration Guide</u> from VMware for more information.

VMware vCenter Update Manager helps the administrator keep VMware components up to date. In this section, you will be installing vCenter Update Manager and perform updates to assure you are at the correct patch level for NETLAB+ operation.

If you are installing vCenter Update Manager on a physical machine, you can burn the vCenter Server .iso image to a DVD. Insert the DVD into the CD/DVD drive.

If you are installing to a virtual machine, upload the vCenter Server image file to the ESXi datastore and add the CD to the virtual machine's CD/DVD drive.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

6.1 Installing vCenter Update Manager with Microsoft SQL Server 2008 R2 (Options 1 and 3)

This section is recommended for vSphere ESXi 4.1 U2 deployments that **will exceed 50 virtual machines**. If you are not exceeding 50 virtual machines, but plan to in the near future, it is recommend you use this section.

 You will use the Management Studio to create the vCenter database. Click the Start Menu>All Programs>Microsoft SQL Server 2008 R2>SQL Server Management Studio.





2. In the *Connect to Server* window, click on the drop-down for the *Server Name* field and click on **<Browse for more...>.**

Connect to Server		
SQL S	erver [*] 2008 R2	
Server type:	Database Engine	
Server name:		
Authentication:	<browse for="" more=""></browse>	
User name:	VCENTER \Administrator	
Password:		
	Remember password	
Connect	Cancel Help Options >>	

3. On the *Browse for* Servers window, click the + sign next to **Integration Services** and click the computer name you configured in section 4.6.1.1. Click **OK** to continue.

Browse for Servers
Local Servers Network Servers
Select the server to connect to:
Database Engine Analysis Services Reporting Services SQL Server Compact Databases Integration Services Integration Services VCENTER
OK Cancel Help



4. On the *Connect to Server* window, change the **Server Type** to **Database Engine**. Click **Connect** to continue.

Microsoft* SQL Server* 2008 R2		
Server type:	Integration Services	
Authentication:	Reporting Services SQL Server Compact Integration Services	
Password:	Remember password	
Conne	ct Cancel Help Options >>	

5. Under *Object Explorer* on the left hand side, click the + sign next to **Databases**. Right-click on **Databases** and select **New Database...**

Kicrosoft SQL Server	Managem	ent Studio)	
File Edit V <mark>iew De</mark> b	ug Tools	Window	Community	Help
🕴 🔔 New Query 🛛 🛅 🛛 🖷	b 🗈 😳	🕒 🔂		-
Object Explorer 				
Konstanting Konstanting Konstanting Konstanting Konstanting Konstanting	ver 10.50.16 New Data	i00 - VCENT	ER\Adi	-
 ⇒ Jystein ⇒ Databa ⇒ OCENT ⇒ Security ⇒ Server Obj ⇒ Replication ⇒ Manageme ⇒ SQL Server 	Attach Restore [Restore f)atabase iles and File	groups	
	Start Pow Reports	verShell		•
	Refresh			

6. In the *New Database* window, enter **UMANAGER** in the **Database Name** field. Click **OK** to continue.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

7. Confirm the **UMANAGER** database is listed under **Databases** on the left hand side.



- 8. Right-click on the UMANAGER database and click Properties.
- 9. Under Select a page on the left, select Options.
- 10. Change the **Recovery Model** to **Simple** via the drop-down box. Click **OK** to continue.
- 11. Exit Microsoft SQL Server Management Studio.
- 12. Next, you need to create the ODBC drivers for VMware vCenter Update Manager. vCenter Update Manager requires a 32-bit system DSN. This is different from the 64-bit system DSN that vCenter Server required.
- 13. Click on the **Start Menu** and click in the **Search** field. Enter **Run**, make sure the **Run** program is highlighted and press **Enter**.
- 14. In the **Open** field, enter the following:

[WindowsDir]\SysWOW64\odbcad32.exe

(i.e. C:\Windows\SysWOW64\odbcad32.exe)

15. Click **OK** to launch the program.

	(
Type the name of a program, folder, document, or Internet resource, and Windows will open it for you.	
C:\Windows\SysWOW64\odbcad32.exe	
🚱 This task will be created with administrative privileges.	
OK Cancel <u>B</u> rowse	
	Type the name of a program, folder, document, or Internet resource, and Windows will open it for you. C:\Windows\SysWOW64\odbcad32.exe This task will be created with administrative privileges. OK Cancel Browse



16. On the *ODBC Data Source Administrator* window, click on the **System DSN** tab. Click on the **Add...** button.

👼 ODBC Da	ita Source Administrator	×
User DSN	System DSN File DSN Drivers Tracing Connection Pooling About	
User Data	a Sources:	
Name	Driver Add	
	Remove	
	Configure	
	An ODBC User data source stores information about how to connect to the indicated data provider. A User data source is only visible to you, and can only be used on the current machine.	
	OK Cancel Apply Help	

- 17. On the *Create New Data Source* window, click on **SQL Server Native Client 10.0** source and click **Finish.**
- On the *Create a New Data Source to SQL Server* window, enter UMANAGER in the *Name* field. Enter your computer-name you set in section 4.6.1.1 (i.e. VCENTER). Click Next to continue.

	This wizard will help you create an ODBC data source that you can use to connect to SQL Server.
SQL Server 2008 R2	What name do you want to use to refer to the data source?
	How do you want to describe the data source? Description:
	Which SQL Server do you want to connect to?



- 19. When asked, "How should SQL Server verify the authenticity of the login ID?" leave the default settings and click **Next** to continue.
- 20. Click the box next to **Change the default database to** and select **UMANAGER** from the drop-down. Click **Next** to continue.

Create a New Data Sour	ce to SQL Server	×
ŠQL Server 2008 R2	 Change the default database to: master model msdb tempdb UMANAGER VCENTER Attach database mename: ✓ Use ANSI quoted identifiers. ✓ Use ANSI nulls, paddings and warnings. 	
	< Back Next > Cancel	Help

- 21. On the next screen, leave the default settings and click **Finish** to continue.
- 22. On the *ODBC Microsoft SQL Server Setup* window, click the **Test Data Source...** button to check the database.
- 23. On the *SQL Server ODBC Data Source Test* window, confirm the tests completed successfully and click **OK** to continue.
- 24. On the ODBC Microsoft SQL Server Setup window, click **OK** to continue.
- 25. On the *ODBC Data Source Administrator* window, make sure the **UMANAGER** data source is listed with the **SQL Server Native Client 10.0** driver. Click **OK** to complete the setup.
- 26. Open the CD/DVD drive with the vCenter Server installation files.
- 27. On the VMware vCenter Installer window, click on vCenter Update Manager.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c



- 28. Select **English** and click **OK** to continue.
- 29. On the Welcome window, click Next to continue.
- 30. On the End-User Patent Agreement window, click Next to continue.
- 31. On the *License Agreement*, review the information, click I accept the terms in the license agreement and click Next to continue.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

- 32. On the *vCenter Server Information* window:
 - a. Enter the vCenter's IP address in the IP Address/Name field.
 - b. Leave the HTTP Port field set to port 80.
 - c. Enter the username in the **Username** field.
 - d. Enter the password in the **Password** field.
 - e. Click Next to continue.

🙀 VMware vCenter Update Manag	jer 🗙	
vCenter Server Information Enter vCenter Server location and credentials		
Please provide the necessary informa Update Manager will need this informa WMware vCenter Server Information	ation about vCenter Server below. VMware vCenter ation to connect to the vCenter Server at startup.	
IP Address / Name:	HTTP Port:	
Username: administrator	Password:	
InstallShield	< Back Next > Cancel	

- 33. On the *Database Options* window, select **Use an existing supported database** and select **UMANAGER (MS SQL)** from the **Data Source Name (DSN)** drop-down box. Click **Next** to continue.
- 34. On the *Database Information* window, review the information and click **Next** to continue.
- 35. On the *VMware vCenter Update Manager Port Settings* window, select your IP address that can communicate with the ESXi hosts. Click **Next** to continue.
- 36. On the *Destination Folder* window, click **Next** to continue.
- 37. On the *Ready to Install the Program* window, click **Install** to begin installation.
- 38. When the installation is completed, click **Finish**.
- 39. Close the VMware vCenter Installer window.
- 40. Proceed to section 6.3.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgr

6.2 Installing vCenter Update Manager with Microsoft SQL Server Express (Options 2 and 4)

This section is recommended for vSphere ESXi 4.1 U2 deployments that **will not exceed 50 virtual machines**.

- 1. Open the CD/DVD drive with the vCenter Server installation files.
- 2. On the VMware vCenter Installer window, click on vCenter Update Manager.



- 3. Select **English** and click **OK** to continue.
- 4. On the *Welcome* window, click **Next** to continue.
- 5. On the *End-User Patent Agreement* window, click **Next** to continue.
- 6. On the *License Agreement*, review the information, click **I accept the terms in the license agreement** and click **Next** to continue.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgr

- 7. On the *vCenter Server Information* window:
 - a. Enter the vCenter's IP address in the IP Address/Name field.
 - b. Leave the HTTP Port field set to port 80.
 - c. Enter the username in the **Username** field.
 - d. Enter the password in the **Password** field.
 - e. Click Next to continue.

🙀 VMware vCenter Update Manag	ger 🛛 🔀	
vCenter Server Information		
Enter vCenter Server location and credentials		
Please provide the necessary informa Update Manager will need this inform	ation about vCenter Server below. VMware vCenter ation to connect to the vCenter Server at startup.	
VMware vCenter Server Informatio	n	
IP Address / Name:	HTTP Port:	
enter your ip address	80	
Username:	Password:	
administrator	•••••	
Install5hield	· · · · · · · · · · · · · · · · · · ·	
	< Back Next > Cancel	

- 8. On the *Database Options* window, select **Install a Microsoft SQL Server 2005 Express instance**. Click **Next** to continue.
- 9. On the *VMware vCenter Update Manager Port Settings* window, select your IP address that can communicate with the ESXi hosts. Click **Next** to continue.
- 10. On the *Destination Folder* window, click **Next** to continue.
- 11. On the *Ready to Install the Program* window, click **Install** to begin installation.
- 12. When the installation is completed, click **Finish**.
- 13. Close the VMware vCenter Installer window.
- 14. Proceed to section 6.3.

6.3 Install the vCenter Update Manager plug-in

Please complete this section after completing your choice of either section 6.1 or 6.2 as appropriate for your installation.

If you installed vCenter Update Manager, you may also install the vCenter Converter Plug-in. This will enable you to use vCenter Update Manager from the vSphere Client.

1. Double-click the vSphere Client icon.


- 2. At the vSphere Client login screen, set the username to local administrator on the vCenter Server and the password for that account. Click **Login**.
- 3. In the menu bar of the vSphere Client, select **Plug-ins > Manage Plug-ins.** The Plug-in Manager appears.

🔗 VCENTER-1 - vSphere Client								
Eile	<u>File Edit View</u> Inventory <u>A</u> dministration <u>Plug-ins</u> <u>H</u> elp							
	Ð			Home	⊳ 🛃	Inventor	Ma	anage Plug-ins

- 4. Under Available Plug-ins, click the Download and Install link, next to the entry for VMware vCenter Update Manager Extension.
- 5. When the download completes, do the following:
 - a. Choose the setup language, **English** and click **OK**.
 - b. On the Welcome page, click Next.
 - c. On the License Agreement page, select **"I accept the terms in the License Agreement"** and click **Next**.
 - d. On the Ready to Install page, click Install.
- 6. Click **Finish** when the installation completes.
- A Security Warning window may appear for the connection to vCenter. Click on the box next to Install this certificate and do not display any security warnings for your-ip. Click Ignore to continue.
- 8. Verify that the VMware vCenter Update Manager Extension plug-in has a status of **Enabled** in the Plug-in Manager.

Plug-in Name	Vendor	Version	Status	Description
Installed Plug-ins				
🍣 🛛 vCenter Storage Monitoring	VMware Inc.	4.1	Enabled	Storage Monitoring and
				Reporting
inter Converter 🕹 🕹	VMware, Inc.	4.2.0	Enabled	Converts physical and virtual
				machines, and backup images to
				VMware virtual machines.
🍣 🛛 vCenter Hardware Status	VMware, Inc.	4.1	Enabled	Displays the hardware status of
				hosts (CIM monitoring)
Inter Service Status	VMware, Inc.	4.1	Enabled	Displays the health status of
-				vCenter services
🎒 🛛 Licensing Reporting Manager	VMware, Inc.	4.1	Enabled	Displays license history usage
Summare vCenter Update Manager Extension	VMware, Inc.	4.1.0	Enabled	VMware vCenter Update Manager
-				extension

- 9. In the Plug-in Manager, click **Close**.
- 10. Exit vSphere Client.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevc

6.4 Performing updates using the vCenter Update Manager plug-in

The following example is performing a VMware ESXi 4.1 U2 update on older ESXi 4.1 hosts. There are many updates generally available. Please review documentation at VMware for more information.

- 1. Double-click the vSphere Client icon.
- 2. At the vSphere Client login screen, set the username to local administrator on the vCenter Server and the password for that account. Click **Login**.
- 3. Click on the **Home** view.
- 4. Click on Update Manager under Solutions and Applications.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

5. Click on **Download Patches** under **Basic Tasks**.



6. Click on the **Baselines and Groups** tab at the top.



7. Click Create... on the Baselines view.

Update Manager Administration for VEENTER							
View Baselines for: Hosts VMs/VAs		Repusitory Those opgrade Releases	•				
Baselines	Baselines Create Edit Delete						
Baseline Name	Content	Туре	Component	Last Modified			
Titical Host Patches	129	Dynamic	Host Patches	12/20/2011 11:45:17			
in Non-Critical Host Patches	179	Dynamic	Host Patches	12/20/2011 11:45:17			

- 8. You are going to create a baseline for an ESXi 4.1 system. Hosts that are added later to this baseline will be checked for compliance and updated to the latest version.
 - a. On the *Baseline Name and Type* window, enter **ESXi 4.1** in the **Name** field. Leave the rest to default settings and click **Next** to continue.
 - b. On the *Patch Options* window, make sure **Dynamic** is selected and click **Next** to continue.
 - c. On the *Dynamic Baseline Criteria* window, select **embedded Esxi 4.1.0** under the **Product** section. Leave the rest of the settings to defaults and click **Next** to continue.
 - d. On the *Patches to Exclude* window, leave defaults and click **Next** to continue.
 - e. On the *Other Patches to Add* window, leave defaults and click **Next** to continue.
 - f. On the *Ready to Complete* window, click **Finish** to create the baseline.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.ne

9. You should now see an ESXi 4.1 baseline in the list.

Update Manager Administration for VCENTER				
Getting Started Baselines and Groups Configuration Events Notifica				
View Baselines for: Hosts VMs/VAs				
Baselines				
Baseline Name	Content			
Titical Host Patches 129				
In Non-Critical Host Patches 179				
🌆 ESXi 4.1 17				

10. Click on the **Getting Started** tab at the top.



11. Click on Go to Compliance View under Basic Tasks.

в	asic Tasks
	🖊 Download patches
Ľ	😨 Go to Compliance View
	🚰 Create a new baseline
	崔 Create a new baseline group
	How-to videos
	12. In this step you will be attaching baselines to your ESXi hosts.a. Click on one of your ESXi hosts under the NETLAB data

- ts under the **NETLAB** datacenter.
- b. Click on the Update Manager tab at the top, if not already selected.

Permissions Maps Update Manager

c. Right-click under Attached Baselines and select Attach...

A	Туре

- d. On the Attach Baseline or Group window, click the box next to ESXi 4.1. Click **Attach** to continue.
- 13. Repeat step 12 for all your ESXi hosts.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netd

14. Click on your vCenter in the left side pane (i.e. VCENTER).



15. Make sure the **Update Manager** tab is still selected at the top.

16. Click the **Remediate...** button in the lower right hand corner.

	Stage	Remediate
Name, Target or Status contains: 👻		Clear ×
License Period: 40	6 days remaining	Administrator

- a. On the *Remediation Selection* window, make sure everything is selected and click **Next** to continue.
- b. On the *Patches and Extensions* window, review the updates to be applied and the number of affected hosts. You can click on the number of hosts to see which host is affected. Click **Next** to continue.
- c. On the *Host Remediation Options* window, enter **Upgrading ESXi 4.1** in the **Task Name** field. If you are performing these steps for the first time, leave the default settings and click **Next** to continue. You can optionally set the **Remediation Time** to a later point if necessary. This is great to schedule for a maintenance window, if your setup requires such.
- d. On the *Ready to Complete* window, review the information and click on **Finish** to perform remediation.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

17. You can watch the progress in the **Recent Tasks** pane at the bottom of the vSphere Client. As you can see, it will put the host in maintenance mode if required and may need to reboot it after install on some patches. This process may take several minutes to hours depending on how many updates and patches are needed.

Recent Tasks					
Name	Target	Status			
Install Enter maintenance mo Enter maintenance mo		77%			
🚰 Tasks 🞯 Alarms 📃					

- 18. You can make sure each host is compliant by selecting the host under the **NETLAB** datacenter and click on the **Update Manager** tab at the top. If compliant, you will see a green dot on the right hand side. If not, it will so remaining patches/upgrades needed for compliancy.
- 19. You can remediate a single host by selecting the host under the **NETLAB** datacenter and click on the **Update Manager** tab at the top. Repeat steps 16-18.

Refer to the <u>VMware vCenter Update Manager Installation and Administration Guide</u> from VMware for more information.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

7 Building Virtual Machines

Detailed guidance on the provisioning of virtual machines is provided in the <u>vSphere</u> <u>Virtual Machine Administration Guide</u>. Becoming familiar with the material in this guide will assist you identifying the best strategy for provisioning your VMs to suit your needs.

There are several techniques for building up an inventory of virtual machines and deploying them in NETLAB+ pods:

- Using NDG Virtual Machine Templates
- Using 3rd Party Virtual Appliances
- Creating Virtual Machines from Scratch
- Cloning Individual VMs Using NETLAB+
- Cloning Pods Using NETLAB+



The diagram above illustrates several ways virtual machines can be created.

1. An initial set of virtual machines are populated in VMware vCenter. These VMs can be derived from NDG virtual machine templates, 3rd party prebuilt virtual appliances, or created from scratch using the VMware vSphere client.



- 2. The initial set of VMs is imported into the NETLAB+ virtual machine inventory.
- 3. Individual VMs can be cloned directly from NETLAB+ to create more virtual machines. Cloned virtual machines can be modified to perform different roles.
- 4. After virtual machines are assigned to a NETLAB+ pod, the pod and its virtual machines can be cloned in one operation.

In this section, you will learn three methods that can be used to create an initial set of virtual machines, starting from an empty inventory:

- Using NDG Template Virtual Machines
- Using 3rd Party Virtual Appliances
- Creating Virtual Machines from Scratch

This section also discusses two important tasks that apply to all virtual machines:

- Configuring virtual machines to provide optimal user performance.
- Taking virtual machine snapshots that can be used to place pods in a clean state.

Detailed guidance on the provisioning of virtual machines is provided in the <u>vSphere</u> <u>Virtual Machine Administration Guide</u>. Becoming familiar with the material in this guide will assist you in identifying the best strategy of provisioning your VMs to suit your needs.

All tasks in this section are performed using only the VMware vSphere client.

In section 8, you will learn how to import your initial set of virtual machines into NETLAB+, then create additional virtual machines quickly using cloning techniques.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netc

7.1 Using NDG Template Virtual Machines and 3rd Party Virtual Appliances

The easiest way to build a virtual machine is to start with an NDG Template Virtual Machine or a 3rd Party Virtual Appliance.

For selected lab content, NDG provides template virtual machines that can be downloaded from the NDG website (<u>http://www.netdevgroup.com/</u>). NDG's templates are designed for specific pods and lab content. They can contain one or more of the following elements, depending on software licenses and distribution restrictions of the software required to satisfy lab content requirements:

- Virtual machine configuration file with suitable default settings
- Operating system
- Application software

This strategy is used to setup NETLAB+ systems to teach the VMware Install, Configure, Manage (ICM) course. More information about setup of the ICM course is available in the <u>Lab Resource Center</u>.

Please refer to the <u>supported lab content options</u> on the NDG website to see if templates are available for particular NDG pods and labs.

Some templates are complete virtual appliances that can be deployed and cloned without modifications. Other templates are partial solutions that require local changes, such as adding an operating system or application software.

In cases where NDG does not provide a template virtual machine, you may find a suitable 3rd party virtual appliance at VMware's website: http://www.vmware.com/appliances

Virtual Appliances are complete solutions that contain a VM configuration file, operating system and applications that are distributed as a single downloadable unit. The VMware vSphere Client (vSphere Client) allows you to deploy virtual machines that are packaged in Open Virtual Machine Format (OVF). Deploying an OVF template is similar to deploying a virtual machine from a template, except that you may deploy an OVF template from any local file system accessible from the vSphere Client machine, or from a remote web server.

An OVF template is stored as set of files (.ovf, .vmdk, and .mf). An OVA template is used to package an OVF template into a single .ova file. Instructions on deploying OVF templates are available in Chapter 5, "Deploying OVF Templates," of the <u>vSphere Virtual</u> <u>Machine Administration Guide</u>.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevg

Virtual machines based on the OVA or OVF file are created using the vSphere client:

File > Deploy OVF Template



The OVF wizard will prompt for the location of an .ova or .ovf file. This can be a file on your local disk, or a web URL.

This process will create a virtual machine in vCenter only. This VM must be later imported into the NETLAB+ virtual machine inventory (discussed in section 8).

Some NDG templates are partial solutions. In this case, please refer to NDG pod-specific guides for additional information required to finalize the installation of the operating system and/or software applications.



7.2 Creating Virtual Machines from Scratch

You may use the VMware vSphere client to create a virtual machine from scratch. Choose this option if prebuilt virtual appliances or NDG virtual machine templates do not meet the requirements you are looking for, such as a particular operating system or hardware configuration. Keep in mind that you can create a single virtual machine and install an operating system on it, then use that virtual machine as a template to clone other virtual machines (see section 8).

- 1. Open the vSphere client and connect to vCenter server.
- 2. Select the ESXi host where the new virtual machine will run.
- 3. Click on the **Getting Started** tab if it is not already selected.
- 4. Select the Create a new virtual machine option.
- 5. Select the **Custom** option for your virtual machine configuration.
- 6. The subsections below will provide information on each step you will need to follow using the **New Virtual Machine Wizard.**



Configuration Select the configuration fo	r the virtual machine
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	Configuration C Typical Create a new virtual machine with the most common devices and configuration options. C Custom Create a virtual machine with additional devices or specific configuration options.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

7.2.1 Providing a Name for Your Virtual Machine

You will be prompted to enter a name for your new virtual machine. Choose a name for the virtual machine very carefully.

For virtual machines used in pods, we recommend including the pod type, numeric ID of the pod and the name of the remote PC (as named in the pod) as part of the name. For example, "Map Pod 12 PC_A" or "ICM Pod 1150 vcenter".

For master virtual machines that will be used as "golden master" virtual machines (used for pod cloning), we recommend including the pod type, role, and name of the remote PC as part of the name. For example, "ICM Host 1 Master vcenter".

🕝 Create New Virtual Machir	
Name and Location Specify a name and loca	tion for this virtual machine
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	Name: Virtual machine (VM) names may contain up to 80 characters and they must be unique within each vCenter Server VM folder. Inventory Location: Inventory Location: NETLAB NETLAB NETLAB NETLAB Netra
Help	≤Back Next ≥ Cancel

7.2.2 Selecting a Datastore

Virtual machine files are stored in a *datastore*. Select a datastore for the virtual machine that will be adequate to store the guest operating system and all of its software applications for pod labs.

Ø Create New Virtual Machine	•					
Datastore Select a datastore in which to store the virtual machine files						
Configuration Name and Location	Select a datastore in which	to store the virt	ual machine files	s:		
Datastore	Name	Capacity	Provisioned	Free	Туре	Thin Provisioning
Virtual Machine Version	[datastore-38-local-1]	227.25 GB	287.48 GB	77.62 GB	VMFS	Supported
Guest Operating System	[datastore-37-san-1]	110.75 GB	998.25 GB	57.07 GB	VMFS	Supported
CPUs Memory Network SCSI Controller	[datastore-37-san-3]	930.75 GB	1.45 TB	634.85 GB	VMFS	Supported
Select a Disk Ready to Complete	•		III			4
	Compatibility:					
	Validation not applicable this	s time.				
Help			ļ	<u>≤</u> Back	Next	Cancel



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

7.2.3 Select the Virtual Machine Hardware Version

If your host supports more than one virtual machine version, you will be prompted to select the virtual machine version to use. Select **Virtual Machine Version 7**.

🕜 Create New Virtual Machine	
Virtual Machine Version	
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	Virtual Machine Version This host or cluster supports more than one VMware virtual machine version. Specify the virtual machine version to use. Virtual Machine Version: 4 This version will run on VMware ESX Server version 3.0 and later, and VMware Server 1.0 and later. This version is recommended when sharing storage or virtual machines with ESX Server versions up to 3.5. Virtual Machine Version: 7 This version will run on VMware ESX Server version 4.0 and later, and VMware Server 2.0. Choose this version if you need the latest virtual machine features and do not need to migrate to ESX 3.
Help	≤ Back Next ≥ Cancel

7.2.4 Selecting the Guest Operating System

The Guest Operating system and version of your choice that you will install on the virtual machine must be selected.

In this example, Microsoft Windows XP Professional 32-bit is selected as the Guest Operating System.

🕝 Create New Virtual Machine	
Guest Operating System Specify the guest operatin	g system to use with this virtual machine
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	Guest Operating System:
Help	≤Back Next ≥ Cancel



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup

7.2.5 Selecting the Number of Processors

Selecting the default value of **1** for number of processors in the virtual machine is typically sufficient, depending on the applications you will run on the virtual machine. Please refer to the pod specific documentation for processor guidance.

Each virtual processor consumes core time on the ESXi hosts physical processors. A value of 2 or higher should be used sparingly.

🕜 Create New Virtual Machine		
CPUs Select the number of virtu	al processors in the virtual machine.	Virtual Machine Version: 7
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	Number of virtual processors:]
Help		Sack Next ≥ Cancel



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

7.2.6 Configuring the Memory Size

Choose the amount of memory that will be allocated to the virtual machine. In most cases, you may use the default settings for memory. If memory space is a concern, you may need to select a value closer to the recommended minimum.

The amount of memory you select for the virtual machine is also the value used in NETLAB+ Proactive Resource Awareness calculations when this virtual machine is scheduled in NETLAB+. See section 5.7 for details.

Setting the memory size too low may lead to page swapping to disk and poor virtual machine performance. Setting the memory size too high may limit the number of virtual machines that can be scheduled at the same time, depending on the amount of physical memory in your ESXi hosts and NETLAB+ Proactive Resource Awareness settings.





7.2.7 Choosing Network Connections

If your equipment pod will consist of only one individual PC, a Network Adapter is not necessary and number of NICs may be set to **None**.

In most cases, it will be necessary to connect a Network Interface Card (NIC) to the virtual machine.

Create New Virtual Machine	
Network Which network connection	will be used by the virtual Machine ?
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	Create Network Connections How many NICs do you want to connect? I Connect at Power On NIC 1: SAFETY NET Flexible Fixed KnowledgeBasefor more information on choosing among the network adapters supported for various guestoperating systems and hosts.
Help	≤ Back Next ≥ Cancel

How many NICs do you want to connect? This determines how many virtual network adapters will be created for the virtual machines. In most cases, this will be **1**. For some pods, a value of 2 or higher may be required for certain remote PCs. Please refer to the NDG pod specific guides for guidance.

Network. This is the name of an existing port group on the ESXi host that the virtual machine network adapter will connect to.

- If the VM is part of a real equipment pod, select an inside port group (previously created in section 5.4.3.3).
- If the target pod type supports NETLAB+ automatic networking, select SAFETY NET (previously created in section 5.5); NETLAB+ will move the VM from SAFETY NET to an automatically generated network when the pod is started.
- Select SAFETY NET if the final network has not been created and you need a safe temporary network until the VM can be relocated to its final network.

Adapter. Network adapter choices depend on the version number and guest operating system running on the virtual machine. Only those network adapters that are



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

appropriate for the virtual machine you are creating are listed as available configuration options. The following adapter choices are described in VMware Knowledge Base article 1001805. Please refer to the article on the VMware website for the latest information.

Vlance	An emulated version of the AMD 79C970 PCnet32 LANCE NIC, an older 10 Mbps NIC with drivers available in most 32bit guest operating systems except Windows Vista and later. A virtual machine configured with this network adapter can use its network immediately.	
VMXNET	The VMXNET virtual network adapter has no physical counterpart. VMXNET is optimized for performance in a virtual machine. Because operating system vendors do not provide built-in drivers for this card, you must install VMware Tools to have a driver for the VMXNET network adapter available.	
Flexible	The Flexible network adapter identifies itself as a Vlance adapter when a virtual machine boots, but initializes itself and functions as either a Vlance or a VMXNET adapter, depending on which driver initializes it. With VMware Tools installed, the VMXNET driver changes the Vlance adapter to the higher performance VMXNET adapter.	
E1000	An emulated version of the Intel 82545EM Gigabit Ethernet NIC. A driver for this NIC is not included with all guest operating systems. Typically Linux versions 2.4.19 and later, Windows XP Professional x64 Edition and later, and Windows Server 2003 (32-bit) and later include the E1000 driver.	
VMXNET3	A next generation of a paravirtualized NIC designed for performance, and is not related to VMXNET or VMXNET 2. It offers all the features available in VMXNET 2, and adds several new features like multiqueue support (also known as Receive Side Scaling in Windows), IPv6 offloads, and MSI/MSI-X interrupt delivery.	
	VMXNET 3 is supported only for virtual machines version 7 and later, with a limited set of guest operating systems:	
	 32 and 64bit versions of Windows XP,7, 2003, 2003 R2, 2008,and 2008 R2 32 and 64bit versions of Red Hat Enterprise Linux 5.0 and later 32 and 64bit versions of SUSE Linux Enterprise Server 10 and later 32 and 64bit versions of Asianux 3 and later 32 and 64bit versions of Debian 4 32 and 64bit versions of Ubuntu 7.04 and later 32 and 64bit versions of Sun Solaris 10 U4 and later 	



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

VMXNET2	Adapter based on the VMXNET adapter but provides some high-performance features commonly used on modern networks, such as jumbo frames and hardware offloads. This virtual network adapter is available only for some guest operating systems on ESX/ESXi 3.5 and later.
	VMXNET 2 is supported only for a limited set of guest operating systems:
	• 32 and 64bit versions of Microsoft Windows 2003 (Enterprise and Datacenter Editions).
	Note : You can use enhanced VMXNET adapters with other versions of the Microsoft Windows 2003 operating system, but a workaround is required to enable the option in VMware Infrastructure (VI) Client or vSphere Client. See Enabling enhanced vmxnet adapters for Microsoft Windows Server 2003 (VMware KB 1007195) if Enhanced VMXNET is not offered as an option.
	 32bit version of Microsoft Windows XP Professional 32 and 64bit versions of Red Hat Enterprise Linux 5.0 32 and 64bit versions of SUSE Linux Enterprise Server 10
	 64bit versions of Red Hat Enterprise Linux 4.0 64bit versions of Ubuntu Linux

Connect at Power On. This box should be checked so that the network adapter is automatically enabled when the virtual machine powers up.

7.2.8 Selecting the Disk Controller

In most cases, you may use the default setting for disk controller. Be aware also of any O/S specific driver requirements due to your selection of guest operating system. For example, older versions of Microsoft Windows may require that you provide SCSI drivers during install.

🕝 Create New Virtual Machine		
SCSI Controller Which SCSI controller type	would you like to use?	Virtual Machine Version: 7
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Ready to Complete	SCSI controller BusLogic Parallel LSI Logic Parallel LSI Logic SAS VMware Paravirtual (not recommended for this guest OS)	MMENDED
	<u>≤</u> Back	Next ≥ Cancel



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgr

7.2.9 Creating a Virtual Hard Disk

Select a Disk	Virtual Machine Version
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Create a Disk Advanced Options Ready to Complete	A virtual disk is composed of one or more files on the host file system. Together these files appear as a single hard disk to the guest operating system. Select the type of disk to use. Disk Create a new virtual disk C Use an existing virtual disk Reuse a previously configured virtual disk. Raw Device Mappings Give your virtual machine direct access to SAN. This option allows you to use existing SAN commands to manage the storage and continue to access it using a datastore. D on ot create disk
Help	≤ Back Next ≥ Cancel

Use the default settings to **Create a new virtual disk** for your virtual machine.

Specify the disk capacity for this virtual machine. Select a disk size that will be adequate to store the guest operating system and all of its software applications for pod labs. The example below shows the default selection of 8GB; your requirements may vary.

Check the **"Allocate and commit space on demand"** check box to enable *thin provisioning*. This is recommended to conserve disk space. Thin provisioning allows real disk space to be allocated on a just-enough and just-in-time basis.

Create New Virtual Machine Create a Disk Specify the virtual disk size	Virtual Machine Versior and provisioning policy	1: 7
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Create a Disk Advanced Options Ready to Complete	Capacity Disk Size: 8 → GB ▼ Disk Provisioning ✓ Allocate and commit space on demand (Thin Provisioning) RECOMMENDED ✓ Support dustering features such as Fault Tolerance Location ⓒ Store with the virtual machine ⓒ Specify a datastore:	
Help	≤ Back Next ≥ Cancel	



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.cc

7.2.10 Specifying Advanced Options

In most cases, you may use the default settings for the **Advanced Options**.

The use of SCSI drivers in a Windows XP or Windows Server 2003 virtual machine requires a special SCSI driver. You may <u>download the driver from the VMware website</u>.

🕝 Create New Virtual Machine		
Advanced Options These advanced options de Configuration Name and Legation	o not usually need to be changed. Specify the advanced options for this virtual disk. These options do not normally	Virtual Machine Version: 7
Varile and Locaton Datastore Virtual Machine Version <u>Guest Operating System</u> <u>CPUs</u> Memory <u>Network</u> S <u>CSI Controller</u> <u>Select a Disk</u> <u>Create a Disk</u> Advanced Options Ready to Complete	 Virtual Device Node SCSI (0:0) IDE (0:0) IDE (0:0) Independent Independent disks are not affected by snapshots. Persistent Changes are immediately and permanently written to the disk. Nonpersistent Changes to this disk are discarded when you power off or revert to the snapshot. 	
Help	<u>≤</u> Back Nex	t ≥ Cancel



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

7.2.11 Verifying the Settings

Create New Virtual Machin Ready to Complete Click Finish to start a task	e s that will create the new s	irtual machine		Virtual	Machine Version: 7
Configuration Name and Location Datastore Virtual Machine Version Guest Operating System CPUs Memory Network SCSI Controller Select a Disk Create a Disk Advanced Options Ready to Complete	Settings for the new v Name: Folder: Datastore: Guest OS: CPUs: Memory: NIC 1 Network: NIC 1 Type: SCSI Controller: Create disk: Disk capacity: Datastore: Virtual Device Node: Disk mode:	virtual machine: Map Pod 2 PC_A NETLAB 10.0.0.38 datastore-37-san-1 Microsoft Windows XP Professional (1 256 MB 1 SAFETY NET Flexible BusLogic Parallel New virtual disk 8 GB datastore-37-san-1 IDE (0:0) Persistent	32-bit)		
	Creation of the v system. Install a	chine settings before completion rtual machine (VM) does not include a guest OS on the VM after creating the	automatic installa e VM.	ation of the gue	st operating
Help			<u><</u> Back	Finish	Cancel

Review the configuration settings displayed on the page and select Finish.

Your virtual machine will now be listed in the virtual machine inventory.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

7.3 Installing a Guest Operating System

After you have configured the virtual machine settings, you must install an operating system on the virtual machine. Refer to VMware's <u>vSphere Virtual Machine</u> <u>Administration Guide</u>, Installing a Guest Operating System for details on the procedure to install a guest operating system.

7.4 Editing the Virtual CD/DVD Device

You may have configured your virtual machine to access a physical CD/DVD drive or access an ISO image in order to install the guest operating system. In the process, you may have enabled the **Connect at Power On** setting. For optimal pod performance, please verify the **Connect at Power On** option is **Unchecked**.

This setting must be edited after installing the guest operating system.

- 1. From the vSphere Client, select the virtual machine from the inventory list.
- 2. Select the **Getting Started** tab if not already selected.
- 3. Click Edit virtual machine settings.





- 4. Select the CD/DVD drive in the hardware list.
- 5. **Uncheck** the **Connect at power on** box. This is necessary to prevent the virtual machine from attempting to connect to the ESXi host's CD/DVD device, which could result in undesired properties or boot errors.

🚱 Map Pod 2 PC_A - Virtual Mach	ine Properties	
Hardware Options Resources		Virtual Machine Version: 7
Show All Devices	Add Remove	Device Status
Hardware Memory CPUs Video card VMCI device Floppy drive 1 Hard disk 1 CD/DVD Drive 1 (edited) Network adapter 1	Summary 256 MB 1 Video card Restricted Client Device Virtual Disk Image File SAFETY NET	Connect at power on UNCHECK Device Type C Clent Device Note: To connect this device, you must power on the virtual machine and then click the Connect CD/DVD button in the toolbar. C Host Device C Datastore ISO File Browse Mode C Passtbrough IDE (recommended) C a set to the toolbar.
Help		Virtual Device Node

Note: You may also point the CD/DVD device connection to a unique ISO image on the local ESXi host. If you choose this option, make sure each VM you create does not point to the same ISO file. Otherwise, you may experience some undesired properties or boot errors.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.ne

7.5 Essential Virtual Machine Performance Optimizations

This section and subsections outline essential performance optimizations for virtual machines that are required for basic operation and good performance in NETLAB+. Our example throughout this section shows the optimization of a Windows XP virtual machine. The same techniques should be applied on all operating systems.

7.5.1 Installing VMware Tools

Installation of VMware Tools is **required** for proper NETLAB+ operation and essential for optimal performance.

The mouse will not work in the NETLAB+ Remote PC Viewer if VMware Tools is not installed.

Your virtual machine must be powered on to install VMware Tools.

- 1. Select the virtual machine in the inventory list.
- 2. Select the Getting Started tab if not already selected.
- 3. Click **Power on the virtual machine**.





The option to install VMware Tools will now be available.

- 1. Select the virtual machine in the inventory list.
- 2. *Right-click* on the page, and select **Guest** from the context menu.
- 3. Select Install/Upgrade VMware Tools.

B Map Pod 2	PC A		
N3 ICN	Power	•	
N3 ICN	Guest	•	Answer Question
	Snapshot	•	Enter Full Screen (Ctrl+Alt+Enter)
	Open Console		Send Ctrl+Alt+del
💁 N3 VM 🍺	Edit Settings		Install/Upgrade VMware Tools
N3 VM	Migrate		
💁 N3 VM 🎥	Clone		

4. Assuming you have completed the installation of the guest operating system as described, you may proceed with the install of VMware Tools.

🕢 Install VMware Tools		
Installing the VMware Tools package will gre performance in your virtual machine.	eatly enhance graphi	cs and mouse
WARNING: You cannot install the VMware T operating system is running. If your guest choose Cancel and install the VMware Tools	Fools package until the operating system is no package later.	ie guest iot running,
	<u>о</u> к	Cancel



7.5.2 Disabling the Desktop Background

The desktop background **MUST** be set to **None** (solid color) to provide minimal bandwidth utilization and to ensure the responsiveness of the remote experience.

- 1. Boot the virtual machine.
- 2. *Right-click* on the display and select **Properties**.
- 3. Click on the **Desktop** tab.
- 4. Select **None** for the Background.

Display Propertie	s ? 🔀
Themes Desktop	Screen Saver Appearance Settings
	.0
Defet start and the first	
Sac <u>kg</u> round:	Browse
Ascent	
📄 🧾 Autumn	<u>Position:</u>
Azul	Stretch
Bliss	<u>C</u> olor:
Customize <u>D</u> es	ktop

A desktop background image will result in very slow screen updates and consume significantly more bandwidth than a solid one-color background.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

7.5.3 Setting the Virtual Machine Display Properties

For optimal performance and minimal bandwidth consumption, we recommend using the lowest possible resolution setting.

• The recommended screen resolution is 1024x768. All courses offered by NDG are compatible with this resolution.

The following task assumes a virtual machine running a Windows XP operating system. Adjust accordingly for other operating systems. To set the screen resolution and color quality:

- 1. Boot the virtual machine.
- 2. *Right-click* on the display and select **Properties**.
- 3. Click on the **Desktop** tab.
- 4. Click on the **Settings** tab.
- 5. Set screen resolution to your desired resolution (1024x768 is used in this example).
- 6. Set color quality to **32-bit**.





7.5.4 Adjusting Visual Effects

Visual effects must be adjusted to provide minimal bandwidth utilization and to ensure the responsiveness of the remote experience.

The following task assumes a virtual machine running a Windows XP operating system. Adjust accordingly for other operating systems.

Adjust the visual effects:

- 1. *Right-click* on My Computer and select Properties.
- 2. Click on the **Advanced** tab.
- 3. Click the **Settings** button for **Performance.**
- 4. Click the Visual Effects tab.
- 5. Select the radio button to Adjust for best performance.
- 6. Click **Ok** to accept changes.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

7.6 Adding Software Applications

You may now add new software to your virtual machine as required by the lab exercises you plan to use on your pods. It may be helpful to temporarily bind the virtual network adapter to the outside campus network to load applications from the Internet or local file server.

The Secure+ network model does not provide outside connectivity. In this model, you will need to load from ISO files on the ESXi servers or from a file server VM on an inside network.

7.7 Virtual Machine Snapshots

A snapshot preserves the state and data of a virtual machine at a specific point in time.

- State includes the virtual machine's power state (powered-on, powered-off, suspended, etc).
- Data includes all the files that make-up the virtual machine, including configuration file, disks, memory, and other devices, such as virtual network interface cards.

The VMware vSphere client provides several operations for creating and managing snapshots and snapshot trees. These operations let you create snapshots, revert to any snapshot in the tree, and remove snapshots.

Large snapshot trees are not recommended for production virtual machines (see Snapshot Best Practices, section 7.7.2 below).

7.7.1 How NETLAB+ Uses Snapshots

In NETLAB+, snapshots are used for two purposes.

Virtual Machine Reset. A virtual machine can be optionally reset to a specific state at the beginning or end of a lab reservation, or when the user initiates the scrub action on the virtual machine or pod.

Linked Virtual Machines. A linked virtual machine shares a base virtual disk with a "master" virtual machine in an ongoing manner. This conserves disk space and allows multiple virtual machines to use the same software installation. A snapshot on the master virtual machine becomes the disk starting point for linked virtual machines that are derived from the master.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.n

7.7.2 Snapshot Best Practices

Here are some best practices for using snapshots in the NETLAB+ environment and some common pitfalls to avoid.

- 1. Be sure to take a new snapshot each time you edit virtual machine settings or install new applications on a virtual machine. Unsaved changes to a virtual machine are lost after reverting to snapshot. A revert to snapshot can occur manually when invoked from the vSphere client, or automatically by NETLAB+ (depending on your virtual machine inventory settings).
- 2. Take snapshots in a powered-off state. Snapshots taken in a powered-on or suspended state can reduce virtual machine boot time. However, this consumes significantly more disk space per VM than in a powered-off state. The disk space required can easily exceed one gigabyte since the virtual machine's memory state and/or swap files must also be preserved for powered-on and suspended states. Resuming from a powered-on or suspended state can also result in failed network connection states in some cases. To avoid these issues, gracefully power down the virtual machine before taking a snapshot.
- 3. Take only one snapshot per virtual machine for best performance. In academic environments, there is often an affinity for large snapshot trees since snapshots provide an intuitive way to return to a topic in a course syllabus. This is not recommended for production virtual machines where user performance is paramount. Large snapshot trees can degrade virtual machine performance. Each level in the snapshot tree creates a "delta disk" which must be accessed before data can be read from the "base disk" (where most files typically reside). We recommend only one snapshot per virtual machine, particularly when the virtual machine is used in a production pod, or for master virtual machines that are the basis for linked virtual machines.
- 4. Specify a unique name for every snapshot (if you must keep more than one snapshot on the same virtual machine). The vSphere client will allow you to create multiple snapshots with the same name. To avoid ambiguity during cloning operations, NETLAB+ requires that all snapshots on a single VM be named uniquely. NETLAB+ cloning operations may fail with the error E_VM_SNAPSHOT_NOT_UNIQUE when attempting to clone a virtual machine that has two or more snapshots with the same name. Duplicate snapshot names will not be an issue if you follow the best practice of one snapshot per virtual machine for best performance.
- 5. Snapshots used for linked clones must be thoroughly tested before cloning. Replacing a snapshot on the master (parent) virtual machine does not propagate changes to existing linked clones (as one might hope). Only new linked clones will pick up the changes from the new snapshot. Linked clones will be discussed in detail in the Cloning section.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.nett

7.7.3 Taking a New Snapshot

- 1. Open the vSphere client.
- 2. Select Hosts and Clusters in the address bar.



3. Right-click on the virtual machine and select **Snapshot > Take Snapshot**.



- 4. Enter a name for the snapshot.
- 5. Enter a description. It is a good idea to include the date in the description for later reference.
- 6. Click **OK**.

🖉 Take Virtual Machine 💷 🖂
Name
Test VM Alpha Snapshot
Description
For Pod Cloning 2011-02-23
Snapshot the virtual machine's memory
Quiesce guest file system (Needs VMware Tools installed)
OK Cancel <u>H</u> elp



7.7.4 Managing Snapshots

vSphere can maintain multiple snapshots of your virtual machine. Use the **Snapshot Manager** to manage snapshots.

In this example, we see that three snapshots have been taken of a virtual machine (after installing the guest operating system, after configuring the remote display commands, and after installing an application).

Snapshots for NETLAB_Server_3_VM_2	
NETLAB_Server_3_VM_2 After_guest_os_install Onfiguration commands completed After Xwire Application Installed You are here	Name After Xwire Application Installed Description Snapshot after installing Xwire application
Go to Delete Delete All	Edit
	Close Help

Be aware of features available using the Snapshot Manager.

- The **You Are Here** icon represents the current operational state of the virtual machine. Each time you take a new snapshot, the Current Snapshot state is updated. NETLAB+ will revert to the current snapshot.
- **Delete** commits the snapshot data to the parent and removes the selected snapshot.
- **Delete All** commits all the immediate snapshots before the **You Are Here** current active state to the base disk and removes all existing snapshots for that virtual machine.
- **Go To** allows you to select the position of the current operational state of the virtual machine. You may maintain multiple snapshots and control which snapshot NETLAB+ will use by using **Go To** in order to modify the position of **You Are Here**, which indicates the current operational state of the VM.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

8 NETLAB+ Virtual Machine Inventory

NETLAB+ version 2011 introduces the Virtual Machine Inventory (VMI). The inventory is a mapping between NETLAB+ remote PCs and virtual machines in one or more vSphere datacenters. The inventory also tracks information about virtual machines that is not stored in vCenter, such as the role each VM plays in NETLAB+ and parent/child relationships between virtual machines.

V A	/irtual Machine Inventory Admin Logout	_	-	-	-		-		
	Import, clone, and manage t NETLAB+.	he inventory of virt	ual machii	nes to be	used with	I			
	Virtual Machine Name	Operating System	Role	Datacenter	Runtime Host	CPUs	Memory (MB)	Pod ID	Pod 1
	P2010120613 ICM 4.1 Template esxi-1	VMware ESX(i)	Template	NETLAB		2	2256		
	P2010120613 ICM 4.1 Template esxi-2	VMware ESX(i)	Template	NETLAB		2	2256		
	😴 2010120613 ICM 4.1 Template san no-Win-ISO	Linux	Template	NETLAB		1	1024		
	Solution 2010120613 ICM 4.1 Template vcenter-1 no-Win-inst	Windows Server 2003	Template	NETLAB		2	3072		
	S 2010120613 ICM 4.1 Template volient no-Win-inst	Windows XP	Template	NETLAB		1	512		
	BackTrack Master	Linux	Master	NETLAB	10.0.0.38	1	768		
	Backtrack Pod 2 Standalone PC	Linux	Normal	NETLAB	10.0.0.38	1	768	1071	Backtrack Pod 2
	BackTrack Pod1	Linux	Normal	NETLAB	10.0.0.38	1	768	1022	BackTrack Pod1
	BT4-R1-template	Linux	Template	NETLAB		1	768		
	Empty XP VM 1	Windows XP	Normal	NETLAB	10.0.0.38	1	256		

8.1 Virtual Machine Roles

Each virtual machine is assigned a *role* in the NETLAB+ inventory. This is a NETLAB+ specific value that is not stored in vCenter. The role indicates the intended function of the virtual machine in NETLAB+ and influences the operations that can be performed on the virtual machine, as well as default settings during those operations. The following roles are currently defined:

- Template. A template is a virtual machine that is used to deploy other virtual machines. Templates cannot be assigned to pods, powered on or edited, providing a more secure way of preserving a virtual machine configuration that you intend to deploy many times. A template VM in NETLAB+ is synonymous with a template VM in VMware vCenter. Virtual machines marked as templates in vCenter are always marked as templates in NETLAB+, and vice-versa.
- Master. By designating a virtual machine a master, you indicate your intention to use the virtual machine for the purpose of cloning other virtual machines and/or cloning the virtual machine as part of a cloned pod. When you clone VMs from a master, and subsequently choose to clone a VM from that clone, the default selection will be to create the clone from the original that was indicated to be the Master. A master VM differs from a template VM in that can be assigned to run on a host, used in a pod (typically a master pod), and powered

on for the purpose of adding and configuring new software. Once the virtual machine has all the required software components and thoroughly tested, a *golden master snapshot* is taken and becomes the basis for full or linked clones (discussed later).

- Normal. A normal VM is a virtual machine that will be assigned to a production pod. A normal VM may be configured to revert to a snapshot at the beginning and end of a lab reservation, or when the user explicitly invokes the scrub action on the VM or the entire pod. A normal VM is an ideal choice for pods that should always start in the same state, such as VMs in Cisco Netacad pods.
- Persistent. A persistent VM is assigned to a production pod, with the additional characteristic that its state will be retained from one reservation to the next. A persistent VM never reverts to a snapshot. This type of VM is appropriate when the user assigned to the pod is working on a curriculum that carries the result of each lab assignment progressively. An example scenario would be a course where the student performs "Lab One" and then on a subsequent reservation performs "Lab Two", with the pod in the state it was in at the end of "Lab One".

8.2 How Virtual Machines Become Part of the NETLAB+ Inventory

Virtual machines must first reside in the NETLAB+ Virtual Machine Inventory before they can be assigned to pods and function as NETLAB+ remote PCs. There are three ways virtual machines can become part of the NETLAB+ inventory:

- Import. NETLAB+ communicates with vCenter and scans the selected datacenter to identify existing VMs that are available to be added to the NETLAB+ virtual machine inventory. The import function is used to import virtual machines that were created outside of NETLAB+ (i.e. from the vSphere Client or VMware Converter).
- **Clone VM.** NETLAB+ communicates with vCenter and makes a copy of a virtual machine that already exists in the NETLAB+ inventory. When a virtual machine is cloned from the NETLAB+ inventory, the new virtual machine is added to both the NETLAB+ inventory and vCenter in one operation. NETLAB+ also supports linked clones (virtual machines that share disk content with another parent virtual machine).
- **Clone Pod.** If a pod contains only virtual machines, NETLAB+ can clone the entire pod in one operation. Each virtual machine in the source pod is cloned and added to both the NETLAB+ inventory and vCenter in one operation.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

8.3 Importing VMs into the Virtual Machine Inventory

The following procedure is used to import an existing virtual machine or template from vCenter into the NETLAB+ Virtual Machine Inventory.

1. From the Advanced Virtual Machine Infrastructure Administrator page, select **Virtual Machine Inventory.** Virtual Machines in the inventory are listed. If this is your first time importing VMs to the inventory, there is no list displayed.



- 2. Click the **Import Virtual Machines** button at the bottom of the page to add VMs to the inventory. If you have added more than one datacenter to your NETLAB+ system, you will be prompted to select a datacenter.
- 3. NETLAB+ will scan the datacenter to discover virtual machines that are not currently in the inventory. You may then click the checkbox next to the virtual machine(s) you wish to import and then click **Import Selected Virtual Machines**.

Select the virtual machines from datacenter tda41 that you wish to important system Select Virtual Machine Name Operating System CPUs Memory Image: ICM 4.1 Master_v1 esxi-2 Other Operating System (64 bit) (experimental) 2 2 2 N3DEV POD107 VC 4.1 TEST Windows XP Professional 1 2 2 2 N3Test ICM Pod 2 Other Operating System (64 bit) (experimental) 2 2 2 NESTED ESXi 4.1 10.0.0.139 Other Operating System (64 bit) (experimental) 2 2 2 Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2 2 2	Imp	ort Virtual Machines			
Select Virtual Machine Name Operating System CPUs Memory ICM 4.1 Master_v1 esxi-2 Other Operating System (64 bit) (experimental) 2 2 N3DEV POD107 VC 4.1 TEST Windows XP Professional 1 1 N3Test ICM Pod 2 Other Operating System (64 bit) (experimental) 2 2 NESTED ESXi 4.1 10.0.0.139 Other Operating System (64 bit) (experimental) 2 2 Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2 2	Sele	ct the virtual machines f	rom datacenter tda41 that you wi	sh to) import.
ICM 4.1 Master_v1 esxi-2 Other Operating System (64 bit) (experimental) 2 N3DEV POD107 VC 4.1 TEST Windows XP Professional 1 N3Test ICM Pod 2 Other Operating System (64 bit) (experimental) 2 NESTED ESXi 4.1 10.0.0.139 Other Operating System (64 bit) (experimental) 2 Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2	Selec	t Virtual Machine Name	Operating System	CPUs	Memory (MB)
N3DEV POD107 VC 4.1 TEST Windows XP Professional 1 N3Test ICM Pod 2 Other Operating System (64 bit) (experimental) 2 NESTED ESXi 4.1 10.0.0.139 Other Operating System (64 bit) (experimental) 2 Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2		ICM 4.1 Master_v1 esxi-2	Other Operating System (64 bit) (experimental)	2	2256
N3Test ICM Pod 2 Other Operating System (64 bit) (experimental) 2 2 NESTED ESXi 4.1 10.0.0.139 Other Operating System (64 bit) (experimental) 2 2 Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2 2		N3DEV POD107 VC 4.1 TEST	Windows XP Professional	1	384
Image: NESTED ESXi 4.1 10.0.0.139 Other Operating System (64 bit) (experimental) 2 2 Image: Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2 2		N3Test ICM Pod 2	Other Operating System (64 bit) (experimental)	2	2256
Test VM 2011 esxi-1 Other Operating System (64 bit) (experimental) 2		NESTED ESXi 4.1 10.0.0.139	Other Operating System (64 bit) (experimental)	2	2256
		Test VM 2011 esxi-1	Other Operating System (64 bit) (experimental)	2	2256
🖙 Import Selected Virtual Machines 🛛 🕂 Select All 🛛 📼 Unselect All 🛛 🖾 Cancel	s an	nport Selected Virtual Machi	nes 🛛 🕂 Select All 📁 Unselect All	×	Cancel



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter w

4. Select the appropriate configuration settings for each virtual machine.

Configure Virtu	al Machines			
Please identify th for each VM. To shared datastore	ne operating syste assign a VM to a ł e (SAN).	m, role, 10st gro	runtime host up, the VM r	t or host group nust reside on a
Virtual Machine Name	Operating System	m	Role	Runtime Host/Group
Test VM 2011 esxi-1	Windows Server 2008	v	Normal 🔽	Host 10.0.0.39 💌
Simport Selected	l Virtual Machines	🔀 Cance	Master e Normal Persistent	

- **Operating System:** This should match the operating system installed on the virtual machine. The default value is usually correct. However, if the virtual machine is running a nested/virtualized instance of VMware ESXi, you should change this value from "Other" to "VMware ESXi".
- Role: Select the role that this virtual machine will play in your inventory (see section 8.1 for definition of roles). If the virtual machine is marked as a Template in vCenter, it will also be set to Template in NETLAB+ (no selection is provided in this case).
- **Runtime Host/Group**: Select the physical VMware ESXi host that will run the virtual machine. The host currently assigned in vCenter is the default choice for virtual machines that are not marked as templates (master, normal, or persistent VMs). Template VMs cannot be powered on and therefore cannot be assigned to a host.

The runtime host should not be changed unless the virtual machine's disk files reside on a SAN and all physical ESXi hosts have access to the virtual machine disk files.

- 5. After selecting configuration settings and selecting **Import Virtual Machines**, the VMs will be added to the inventory.
- 6. Select **OK** to return to the virtual machine inventory.

Virtual Machine Name	Import Status
Test VM 2011 esxi-1	🥥 ок
© ок	

7. The imported VM is now displayed in the virtual machine inventory.


Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

Virtual Machine Name	Operating System	Role	Datacenter
📕 Test VM 2011 esxi-1	Windows Server 2008	Normal	tda41
		-	
🖳 Import Virtual Machines 🛛 🔀 E	kit		

8.4 Virtual Machine Cloning

Cloning virtual machines can save you a substantial amount of setup time if you are deploying many similar virtual machines. A *clone* is a copy of a virtual machine. Cloning a virtual machine creates a copy of a virtual machine, including its settings, any configured virtual devices, installed software, and other contents of the virtual machine's disks. You can create, configure and install software on a single virtual machine, and then clone it multiple times.

After a virtual machine is cloned, the clone can be modified as needed. For example, you may wish to change the IP address or client name on several cloned virtual machines. If there is a particular virtual machine configuration that you will want to clone many times, a good strategy is to create a *master VM* or *template VM*. Both roles designate that the virtual machine is to be used to create other virtual machines. A master VM can be part of a pod and can be powered on. A template cannot be powered on or edited, providing a more secure way of preserving a virtual machine configuration that you intend to deploy many times. A template VM in NETLAB+ is synonymous with a template VM in vSphere.

A *full clone* is an independent copy of a virtual machine that shares nothing with the parent virtual machine after the cloning operation. Ongoing operation of a full clone is entirely separate from the parent virtual machine.

A *linked clone* is a virtual machine that shares virtual disks with the parent virtual machine in an ongoing manner. This conserves disk space and allows multiple virtual machines to use the same software installation. Linked clones can be created very quickly because most of the disk is shared with the parent VM.

It is possible to create full clones using the VMware vSphere Client. Keep in mind, however, that cloning operations in NETLAB+ update both VMware vCenter and NETLAB+ inventory in one operation.



8.4.1 Golden Masters and Golden Snapshots

Master virtual machines and templates should be thoroughly tested before making production clones (full or linked). A master virtual machine (or template) that has been tested and deemed to be production quality is called a *golden master*. Linked clones also require a pristine snapshot that becomes the base disk for linked clones. This snapshot is called a *golden snapshot*.

Keep in mind the following important rules about golden snapshots.

- 1. Updates to the golden snapshot only affect NEW clones (full or linked).
- 2. Updates to the golden snapshot do not affect EXISTING clones (full or linked). Since snapshots operate on disk sectors and not individual files, this would lead to disk corruption and is prevented by vSphere.

It is inevitable that at some point your master VMs will need to be updated, either in response to a defect or to install new files. When this need arises, you must decide whether to update existing clones, or create them over again. Since linked clones can be made very quickly (including pod cloning), it may be easier to simply re-clone from the updated master VMs. Should you decide to update existing clones, be sure to create a new snapshot on each clone if the clone is a normal VM that reverts to snapshot.

8.4.2 Using NETLAB+ to Clone a Single Virtual Machine

NETLAB+ provides a convenient way to clone virtual machines. In order to clone virtual machines within NETLAB+, you must first create a source virtual machine or template (see section 7), to be used as the source for the clone. The source virtual machine must also be registered in the NETLAB+ inventory.

- To clone a virtual machine, select the source virtual machine to be cloned in the NETLAB+ Virtual Machine Inventory. In this example we have a selected a virtual machine that has been designated a master. Notice also that since this virtual machine has been assigned to a pod, the pod name and link to the pod management page are available.
- 2. Begin the cloning process by selecting the **Clone** button.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

Showing VM ICM Master e	esxi-1	
Virtual Machine Name	ICM Master esxi-1	Rename
Operating System	∨Mware ESX(i)	
Role	Sector Character Character	nge Role
Datacenter	tda41	
Runtime Host	10.0.0.39	
Host Group	n/a	
CPUs	2	
Memory (MB)	2256	
Pod ID	1000	Q
Pod Name	ICM Master Pod	A
PC Name	esxi-1	a
Parent	(no parent)	
Children	30	
Datacenter Unique Identifier	50003a87-c060-3e56-0b48-14070ac1f623	
Datacenter Storage Location	[datastore-37-san-3] ICM Master esxi-1/ICM Master esxi-1.vmx	
Comments		
sedit Chang	e virtual machine parameters.	
Clone Clone	this virtual machine.	
Remove Remove option	ve this virtual machine from the NETLAB+ inventory (ar ally from the datacenter).	nd
🔀 Exit Return	to inventory.	



3. The **Clone Virtual Machine** screen will be displayed. Select the appropriate settings for the fields as described below.

Cloning Virtual Machine (Generic Pod Master host1			
Parent Name Parent Role	Generic Pod Master host1 🛃 Master			
Parent Snapshot	Current Snapshot (GOLD	EN_MASTER)		
Clone Name	POD 1016 client1			
Clone Type	 Linked Clone Full Clone 			
Clone Role	 Premplate Master Normal Persistent 			
Runtime Host or Group	Host 10.0.0.38	•		
Datastore	datastore-37-san-3	•		
Storage Allocation	 On Demand Preallocated 			
		show help tips 🕅		
🕝 OK 🛛 🔀 Cancel				

- **Parent Name:** The name of the existing parent virtual machine. In this example we are making a clone of a virtual machine designated as a master. This master is the parent of the virtual machine.
- **Parent Role:** The role the parent virtual machine plays in the inventory.
- **Parent Snapshot:** A snapshot name on the parent virtual machine from which to base the clone. If this parameter is set, the clone is based on the snapshot point. This means that the newly created virtual machine will have the same configuration as the virtual machine at the time the snapshot was taken. If this property is not set, the clone is based on the virtual machine's current configuration. Linked cloning requires this parameter to be set to a snapshot.
- **Clone Name:** The name of the new virtual machine.



- **Clone Type:** There are two types of clones:
 - Linked Clone: A copy of a virtual machine that shares virtual disks with the parent virtual machine in an ongoing manner. This conserves disk space and allows multiple virtual machines to use the same software installation. Linked clones can be created very quickly because most of the disk is shared with the parent VM.
 - **Full Clone:** An independent copy of a virtual machine that shares nothing with the parent virtual machine after the cloning operation. Ongoing operation of a full clone is entirely separate from the parent virtual machine.
- **Clone Role:** Assign the clone to the role it will play in the inventory.
 - **Template:** A pristine virtual machine image used as the basis for cloning many virtual machines. Template VMs cannot be powered on, modified, or assigned to pods.
 - Master: A virtual machine used as the basis for cloning other virtual machines. Master VMs can be assigned to pods, modified and powered on.
 - Normal: A virtual machine that can be assigned to a pod. A normal VM will typically revert to a specified snapshot at the start of a lab reservation.
 - **Persistent:** A virtual machine that can be assigned to a pod and retains its state between labs. A persistent VM is typically used in conjunction with Pod Assigner to create long-term personal pods.
- **Runtime Host:** The host server where the virtual machine will run.
- **Datastore:** The VMware datastore that will contain the new VM's virtual disk files.
- **Storage Allocation:** Indicate the type of storage allocation to be used for the virtual machine.
 - **On Demand** allocates storage for the new virtual machine as needed; this favors storage reduction at the expense of decreased write performance.
 - **Preallocated** allocates all storage for the new virtual machine in advance; this favors better write performance at the expense of increased storage.



4. Indicate the appropriate selections for each field and then click the **Clone** button to initiate the cloning process. The results of the process will be displayed.

01	Virtual machine 'Generic Pod Master host1' was successfully cloned to 'POD 1016 client1'.						
	Return to Inventory	Return to Parent VM	■ Go To Cloned VM				

8.5 Assigning Virtual Machines to Pods

The virtual machines residing in the Virtual Machine Inventory must be assigned to an equipment pod in order to be available to users for scheduled access. Remote PCs are only available in pods where the network topology indicates the existence of lab PCs.



In this section, we will discuss the methods that may be used to assign virtual machines to equipment pods.

- Select the Equipment Pods administrator option. As an example, we will use a newly created equipment pod that is designed to include one PC. For details on creating equipment pods refer to the "Adding New Pods" section of the <u>NETLAB+</u> <u>Administrator Guide</u> Notice that the type of the PC/VM is indicated as "ABSENT". This is the default setting for PCs in new pods.
- 2. To display the PC settings, select the button to the left of the PC name.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

Pod Mana	gement				NETLAB+ 2011.R1V.beta.23
					administrator
POD 1016 - S	TATUS				
POD ID	POD NAME		STATUS	ACTIVITY	POD TYPE
1016	POD 1016	0	OFFLINE	IDLE	BB 2 CLIENT
POD 1016 - P	Cs AND SERVERS (cli	ck the GO b	uttons to reconfi	gure)	
GO	NAME	PC ID	STATUS	TYPE / VM	OPERATING SYSTEM
Q	client1	4188	ONLINE	ABSENT	
Q	client2	4189	ONLINE	ABSENT	
Q	server1	4190	ONLINE	ABSENT	

3. Select the option to Modify PC Settings.

	PC Configuration Admin Logout			
	🗊 POD 1016 - client1			
	Pod ID	1016		
	Pod Name	POD 1016		
	PC Name	client1		
	РС Туре	ABSENT		
	nodify PC Settings	Return to Pod Management		

4. Set the PC Type to **Use Virtual Machine Inventory.** This will allow you to use a virtual machine defined in the Virtual Machine Inventory (VMI). The other selections are for backward compatibility with other technologies that do not integrate with VMI and vCenter.

🚅 POD 1016 - client1		
Pod ID Pod Name PC Name	1016 POD 1016 client1	
PC Type	ABSENT	
PC Unavailable Message	ABSENT Use Virtual Machine Inventory	(optional)
	VMware ESXi 4.0 (no vCenter)	show help tips 🥅
Vpdate PC Settings 🛛 🛛 Car	VMware ESXi 3.5 U3 (no vCenter) VMware Server 2.0 VMware Server 1.0/GSX STANDALONE	



5. If your system includes more than one datacenter, you will be prompted to select the datacenter where the virtual machine you will be selecting resides.

🚅 POD 1016 - client1	
Pod ID	1016
Pod Name	POD 1016
PC Name	client1
PC Type	Use Virtual Machine Inventory
Base Datacenter	Select the datacenter containing the virtual machine you want to use.
	<select datacenter=""></select>
	<select datacenter=""></select>
Cancel	tda41

6. Select the virtual machine that you would like to assign to the remote PC. All virtual machines in your Virtual Machine Inventory that have not been assigned to a pod will be available for selection.

🚅 POD 1016 - client1		
Pod ID	1016	
Pod Name	POD 1016	
PC Name	client1	
РС Туре	Use Virtual Machine Inventory	
Base Datacenter	NETLAB	
Base Virtual Machine	Select the base virtual machine that will be used for this PC.	
	<select a="" machine="" virtual=""></select>	
Cancel	POD 1013 host1 POD 1013 host2 POD 1013 host3 POD 1013 host4	^
	POD 1016 client1 POD 1017 2C1S Student 2 client1 POD 1017 2C1S Student 2 client2 POD 1017 2C1S Student 2 server1	E



7. After selecting a virtual machine from the inventory list, you will set the PC configuration. Select the settings appropriate for the manner in which you will use the virtual machine. The PC settings are described below.

🚅 POD 1016 - client1			
Pod ID	1016		
Pod Name	POD 1016		
PC Name	client1		
РС Туре	Use Virtual Machine Inventory	-	
Base Datacenter	NETLAB	-	
Base Virtual Machine	POD 1016 client1	-	
Base Role	Persistent VM		
Base Snapshot	n/a (persistent virtual machine retains state)		
Shutdown Preference	Graceful Shutdown from Operating System	•	
Guest Operating System	Windows XP	•	
Options	😰 enable remote access to keyboard, video, and mouse	٦	
	enable remote display auto-configuration		
	enable network auto-configuration		
	enable advanced setting auto-configuration		
	enable minimum requirements verification		
Admin Status	ONLINE -		
✓ Update PC Settings ⊠ Cancel			

- **Pod ID:** The NETLAB+ identifier of the pod containing this PC.
- **Pod Name:** The user-defined name of the pod containing this PC.
- **PC Name:** The name of the PC as defined in the pod design.
- **PC Type:** The setting should remain at **Use Virtual Machine Inventory**. This allows you to select a virtual machine defined in the NETLAB+ Virtual Machine Inventory. The VMI offers the most advanced VM configuration and automation capabilities available in NETLAB+.
- **Base Datacenter:** The virtual datacenter that contains the virtual machine to be used for this PC (unless overridden by a lab).
- Base Virtual Machine: The virtual machine that will be used for this PC.



- **Base Role:** The pre-assigned role that the base virtual machine plays in the inventory.
- **Base Snapshot:** The snapshot that will be used to revert the base virtual machine to a clean state during pod initialization, user initiated scrub action, and at the end of a lab reservation. This setting does not apply to persistent VMs because that always retain state.
- Shutdown Preference: Select the preferred shutdown sequence if the virtual machine is still powered on at the end of a lab reservation. If a base snapshot is configured, it is reverted first. If the virtual machine is still powered on after reverting to the specified snapshot, the preferred shutdown sequence is executed. Otherwise, the final power state will be the same as the snapshot state.
 - Graceful Shutdown from Operating System: Perform an orderly shutdown from the operating system if possible (i.e. VMware Tools is supported and installed on the VM). This is best setting in most situations. If an orderly shutdown is not possible, the virtual machine is powered off.
 - **Power Off**: Powers off the virtual machine. Does not perform an orderly shutdown.

This option is not recommended for persistent virtual machines as this may lead to disk corruption.

• **Suspend Virtual Machine**: Suspend the virtual machine in its current state. When powered on, it will resume in the same state without booting.

This option will cause virtual machines memory and page files to be retained on disk. Significant disk space per VM may be required to support this option. Network connection state is not preserved.

• **Keep Running**: The virtual machine is left in the powered on state.

This is rarely desirable since the VM will continue to consume host CPU and memory resources.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter

- Guest Operating System: The operating system running on this virtual machine.
- **Options:** Enable or disable automated features.
 - Enable remote access to keyboard, video, and mouse: if enabled, NETLAB+ will allow the virtual machine to be accessed using a built-in remote PC viewers. Keep this option enabled unless you do not want users to access the VM; for example, a special server in the pod that should not be configured by the user.
 - Enable remote display auto-configuration: if enabled, NETLAB+ will automatically configure remote display parameters on the VM after a power on or revert to snapshot operation. This option allows VMs to be cloned without manual customization of the remote display settings. Therefore, we recommend you enable this setting and be happy about it.
 - Enable network auto-configuration: if enabled, NETLAB+ will automatically bind the virtual machine's network adapters to the correct port group. This option allows VMs to be cloned without manual customization of networking bindings. This option is ignored if the pod type does not support automatic networking. If this setting is disabled and/or the pod type does not support automatic networking, then the virtual machine network adapters must be manually bound to the correct port group(s) using the vSphere client.
 - Enable advanced setting auto-configuration: if enabled, NETLAB+ will automatically program advanced virtual machine settings that may be required for a particular pod type. For example, nested VM support for the VMware IT Academy program. It is usually safe to enable this option. It will be ignored if the pod type does not require advanced settings.
 - Enable minimum requirements verification: if enabled, NETLAB+ will verify that the current settings of the virtual machine meet or exceed any minimum requirements established for the remote PC to which it is assigned. It is usually safe to enable this option. The setting will be ignored if the pod type does not specify a set of minimum requirements for the remote PC.
- Admin Status: Set administrative status to ONLINE to enable this PC. You can temporarily disable this PC by setting the administrative status to OFFLINE.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.

- 8. After updating all PC configuration settings as needed, select Modify PC Settings.
- 9. After the settings have updated, you may continue to the Pod Page or the PC page.





Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

9 Cloning Virtual Machine Pods

The ability to clone entire virtual machine pods is a NETLAB+ feature that greatly reduces the amount of time needed for setup of your system, when the situation calls for several identical equipment pods consisting of virtual machines.

Pod cloning may be used to clone pods that consist of virtual machines only. At this time, it is not possible to clone pods that include hardware lab devices (such as routers and switches).

NETLAB+ pod cloning supports *linked clones*. A linked clone is a virtual machine that shares virtual disks with the parent virtual machine in an ongoing manner. This conserves disk space and allows multiple virtual machines to use the same software installation. The pod cloning utility can create linked clones very quickly because most of the virtual disk is shared with the parent VM.

There are two basic pod cloning strategies, depending on whether your virtual machines will be stored on an ESXi host local disk or on a Storage Area Network (SAN). Both strategies leverage linked clones so that production virtual machine pods can be created quickly with minimum disk space.

Local Disk Strategy. If your virtual machines are stored on an ESXi server local disk, you will create one master pod (per pod type) on each ESXi host. Production VMs on each server will link to their respective master pod VMs. Each host requires a master pod because Host A cannot access the disks of Host B, and vice-versa.





Storage Area Network Strategy. Only one master pod (per pod type) is required if your virtual machines are placed on a SAN. Virtual machines in the production pods will link to the virtual machines in the master pod, regardless of which ESXi host the virtual machines have been assigned to run on. This is possible because each ESXi host has access to the storage area network (where all virtual disks reside).



NDG performs all testing on servers with Internal Direct Attached Storage (i.e. RAID arrays and RAID controllers directly attached to each ESXi host server). This is the configuration that most academic institutions are likely to find affordable and adopt. A Storage Area Network (SAN) is a dedicated network that provides access to consolidated, block level data storage that can be used for disk storage in a VMware vSphere environment.

Currently NDG does not provide benchmarks, guidance or troubleshooting for SAN configurations. Our documentation may show an optional SAN in the environment, however this is not a recommendation or requirement to deploy a SAN.

NDG benchmarks and capacity planning guidance do not account for the additional latencies introduced by SAN.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

9.1 Golden Masters and Golden Snapshots

Virtual machines in a master pod should be thoroughly tested before pod cloning takes place (full or linked). A master virtual machine that has been tested and deemed to be production quality is called a *golden master*. Linked clones also require a pristine snapshot that becomes the base disk for linked clones. This snapshot is called a *golden snapshot*.

Keep in mind the following important rules about golden snapshots.

- 1. Updates to the golden snapshot only affect NEW clones (full or linked).
- 2. Updates to the golden snapshot do not affect EXISTING clones (full or linked). Since snapshots operate on disk sectors and not individual files, this would lead to disk corruption and is prevented by vSphere.

At some point, the VMs in your master pod will need to be updated, either in response to a defect or to install new files. When this need arises, you must decide whether to update existing cloned pods, or create them over again. Since linked clones can be created very quickly, it may be easier to simply re-clone the pod rather than touch every existing clone pod.

9.2 Creating a Master Pod

The first step in the pod cloning process is to create a master pod. At this time, the pod type must contain only virtual machines if it is to be cloned using the pod cloning utility. The following steps are used to create a master pod.

- 1. Create a new pod in NETLAB+ of the desired type. All of the pod VMs will initially be set to ABSENT.
- 2. Using the techniques outlined in section 7, build some virtual machines that will map to each remote PC position for the pod type you are targeting.
 - a. Each virtual machine in the master pod should have its role set to **master**. The role can be set when importing a virtual machine or cloning a virtual machine. You can also change the role of an existing virtual machine to master.
 - b. Whether you are storing VMs on an ESXi local disk or SAN, you should assign the master to run on one of the ESXi hosts so software can be installed on the VMs and the pod tested as a single functional unit.
- 3. Assign your master virtual machines to the new master pod (section 8.5).
- 4. You may wish to use Pod Assigner at this point to prevent others from scheduling the pod. You may assign the pod to an instructor account for this purpose.



- 5. Many NDG virtual pods support automatic networking. On the other hand, if you are using a custom built pod or other pod that does not have automatic networking support, you should make sure the virtual machines network adapters are bound to the correct networks at this time.
- 6. Bring the pod online so that it can be tested.
- 7. Test the master pod.
 - a. Login to an instructor account (use the same account used with pod assigner in step 3 if applicable).
 - b. Scheduled lab reservation in NETLAB+ and test the pod thoroughly including testing all the virtual machines to assure proper configuration.
- Linked clones in a master pod require a pristine snapshot that becomes the base disk for your production linked clones. This snapshot is called a *golden snapshot*. Once you are **very confident** that the virtual machines in the master pod are free of defects, it is time to create a golden snapshot.
 - a. In most cases, you will want to power down every virtual machine before taking the golden snapshot, for reasons explained in section 7.7.4. You can do this from the NETLAB+ action tab or from the vSphere client.
 - b. Take a snapshot on each master VM using a name such as GOLDEN_SNAPSHOT. Currently, you must take snapshots using the vSphere client.

Once each virtual machine in the master pod has a golden snapshot, linked clones are possible.

9.3 Cloning a Virtual Machine Pod Using Linked Clones

In this section you will learn how clone a master virtual machine pod using linked clones. Let's suppose that the master pod is called pod MP1 and we want to create similar cloned pods called CP1, CP2, CP3, etc. The first time you clone the master MP1 to create cloned pod CP1, you will need to specify some parameters for each virtual machine in CP1. NETLAB+ will keep track of the parameters used to clone CP1. You can then instruct NETLAB+ to "create a CP2 pod like CP1" by cloning from CP1 instead of MP1.

In our first example, we will make clone the master pod for the first time (MP1 to CP1). The 3 virtual machines are master VMs with a golden snapshot. Recall that a golden snapshot on master VM is required to create linked clones.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrc

Pod Man Admin	agemen	t	-	-		l	NETL	AB+ 2011.R1V.beta.23 administrator
POD 1020 - POD ID 1020	STATUS 2C1	POI IS Host	D NAME A Maste	r 1020	STATUS	ACTIVI	TY	POD TYPE
POD 1020 - GO	POD 1020 - PCs AND SERVER GO NAME Q Image: Client1 Q Image: Client2		6 (click th PC ID 4191 4192	e GO buttons STATUS ONLINE ONLINE	o reconfigure) TYPE / VM Host A Master client1 Host A Master client2		OPERATING SYSTEM Windows XP Windows XP	
Server1 4193 ONLINE Host A Master server1 Windows XP Pod 1020 Management Options								
Online Bring this pod ONLINE and make it available for reservations. Test Tell me if this pod is working properly.					S.			
Clone Create a new pod based of Rename Rename this pod. Remove this pod from NET				n the settings of this	pod.			
Delete Nemove this pourion included.								

If the clone button appears on the pod management screen, the pod can be cloned. If the clone button does not appear, the pod is not eligible for cloning (i.e. it contains real equipment, or contains VMs that do not use the NETLAB+ inventory).

- 1. We begin the pod cloning process by clicking the **Clone** button.
- 2. Select a numeric ID for the new pod and click **Next**.

Select an ID for the ne	w pod					
Source Pod ID	1020					
Source Pod Name	2C1S Host A Master 1020					
New Pod Type	NDG 2 Client / 1 Server Pod					
New Pod Type						
Next Cancel						

3. Specify a unique descriptive name for the new pod and click **Next**.

Ĩ	Select an ID for the new pod					
	Source Pod ID	1020				
i	Source Pod Name 2C1S Host A Master 1020					
	New Pod Type NDG 2 Client / 1 Server Pod					
ł	New Pod ID 1021					
ł	New Pod Name 2C1S Host A Pod 1021					
į.						
Ì	Next 🔄 Previous 🔯 Cancel					



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.

4. The pod cloning parameter form appears.

Sc	ource Pod ID	1020								
Source Pod Name 2C1S Host A Master 1020										
New Pod Type NDG 2 Client / 1 Server Pod										
	New Pod ID	1021								
Nev	w Pod Name	2C1S Host	A Pod 1021							
PC Name	Source Virtua	I Machine	Source Snapshot	⇒	Clone Name	Clone Type	Clone Role	Runtime Host	Clone Datastore	Storage Allocation
client1	Host A Master	r client1 💌	GOLDEN_SNAPSHOT	9 🛛 🖷	Pod 1021 client1	Linked 💌	Persistent 💌	Host 10.0.0.38 💌	datastore-37-san-3 💌	On Demand 💌
client2	Host A Master	r client2 💌	GOLDEN_SNAPSHOT	🜉 ස 🖏	Pod 1021 client2	Linked 👻	Persistent -	Host 10.0.0.38 💌	datastore-37-san-3 -	On Demand 💌
server1	Host A Master	r server1 👻	GOLDEN_SNAPSHOT	9 @ 5	Pod 1021 server1	Linked 👻	Persistent -	Host 10.0.0.38 💌	datastore-37-san-3 ▼	On Demand 💌
Clon	Clone Pod 🔄 Previous 🔀 Cancel						•			

- 5. Verify that each source virtual machine is set to the correct master virtual machine.
- 6. Verify that each source snapshot is set to the golden snapshot.
- 7. Set each clone type to Linked.
- 8. Set the role to **Normal** or **Persistent**. Use normal if the VMs will revert to a snapshot to reset state. Use persistent if the VMs will retain state between lab reservations.
- 9. Set each runtime host to the ESXi server where each VM will run. This should be set to the same host containing the master VMs unless your VMs are located on a SAN. If using a SAN, you can change the value as desired.
- 10. Set the datastore that will be used to store the new VMs. This datastore must be accessible to the runtime host.
- 11. Verify that the Storage Allocation setting is set to **On Demand**.
- 12. Click **Next** to start the cloning process.

Clone Pod							
	jout						
New Pod ID	New Pod Name		New Pod Type	Pod Clone Progress	Total Errors	Total Warnings	
1021	2C1S Host A Pod 1	021	NDG 2 Client / 1 Server Pod	DONE	0	0	
PC Name Virtual Machine Name Progress Notifications							
client1	Pod 1021 client1	DON	Cloned virtual machine	ne to pod.			
client2	Pod 1021 client2	DON	E Cloned virtual machine Attached virtual machine	 Cloned virtual machine. Attached virtual machine to pod. 			
server1 Pod 1021 server1 DONE			 Cloned virtual machine. Attached virtual machine to pod. 				
С ОК							



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdev

9.4 Tasks to Perform After Pod Cloning

Here are some additional tasks that may be required after pod cloning.

- If you want to restrict access to the new pod, use the Pod Assignment utility. This is often the case with persistent pods where you want to assign virtual machines to a student or team for a long period of time.
- If the pod contains normal VMs that should revert to snapshot, take a snapshot of each new VM using the vSphere client. Make sure each remote PC is set to revert to this snapshot.
- If the virtual pod does not support automatic networking, be sure to bind each VM to the proper port group using the vSphere client. If your VMs will revert to a snapshot, be sure your snapshot is taken after this binding is made.

When you are finished with all tasks, do not forget to bring the pod online, otherwise it will not appear in the scheduler.

9.5 Saving Time on Subsequent Pod Cloning

In our last example, we cloned the master pod (MP1) for the first time. We had to change several parameters on the pod cloning form to produce the cloned pod (CP1). You could produce a second and third pod (CP2 & CP3) by cloning MP1 repeatedly. However, you will have to make the same form changes again and again.

A more efficient way to produce CP2 and CP3 is to clone pod CP1. When NETLAB+ sees that CP1 is based on linked clones, it will assume you want to create a similar pod using linked clones based on the same master VMs. The default values will be the same as those you set for CP1, eliminating the need for changes. In most cases, you can click the Clone Pod button without form changes and NETLAB+ will produce the correct result.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.net

9.6 Creating a Full Clone of a Virtual Machine Equipment Pod

You can also use the pod cloning utility to create new pods that are full clones of the original. A full clone is an independent copy of a virtual machine that shares nothing with the parent virtual machine after the cloning operation. Ongoing operation of a full clone is entirely separate from the parent virtual machine.

The pod cloning procedure for full clones is similar to linked clones (see section 9.3). Only the VM cloning parameters will change.

Sc	Durce Pod ID 1020							
Source Pod Name 2C1S Host A Master 1020								
New Pod Type NDG 2 Client / 1 Server Pod								
New Pod ID 1030								
New Pod Name 2C1S Host B Master 1030								
PC Name	Source Virtual Machine	Source Snapshot	⇒	Clone Name	Clone Type	Clone Role	Runtime Host or Group	Clone Datastore
client1	Host A Master client1	GOLDEN_SNAPSHOT	9 🖹 9	Host B Master 1030 client1	Full 💌	Master 💌	Host 10.0.0.38	datastore-37-san-3
client2	Host A Master client2	GOLDEN_SNAPSHOT	909	Host B Master 1030 client2	Full 👻	Master 💌	Host 10.0.0.38 💌	datastore-37-san-3
server1	Host A Master server1	GOLDEN_SNAPSHOT	9 🗅 9	Host B Master 1030 server1	Full 💌	Master 💌	Host 10.0.0.38 💌	datastore-37-san-3
🕜 Clon	e Pod 🛛 🧼 Previous 🛛 💈	Cancel						

- 1. Set each Clone Type to **Full**.
- Specify a source snapshot to be used as the starting point of each cloned VM. You may also select **Current State** to clone the current disk state of the parent VM.
- 3. Specify the role of each new VM.
- 4. Select the runtime host for each new VM. Since you are making a full copy of each VM, you are free to choose a different runtime host.
- 5. After selecting the runtime hosts, select the datastores where the new VMs will reside.
- 6. Click the Clone Pod button.

9.7 Creating Pods that Run on a Multiple VMware ESXi Hosts

Virtual machines are assigned to run on a specific ESXi host. Virtual machines in the same pod should typically be assigned to the same host to simplify networking; this is required for NDG pods that support automatic networking.

To leverage the compute power of multiple ESXi hosts, you can place pods on each host so that the load will balance on average. When working with multiple ESXi hosts, your pod cloning strategy will depend on whether your virtual machines are stored on an ESXi host local disk or on a Storage Area Network (SAN).



Local Disk Pod Cloning Strategy. Refer to the diagram below. If your virtual machines are stored on an ESXi server local disk, you will create one master pod (per pod type) on each ESXi host (MPA and MPB). Production pods on each server will link to their respective master pod VMs. Each host requires a master pod because Host A cannot access the disks of Host B, and vice-versa.



The **local disk pod cloning strategy** is played by using the following moves.

- 1. Create master pod MPA on host A.
- 2. Create the first host A production pod CPA1 using linked clones (per section 9.3).
- 3. Create additional host A production pods CPA2, CPA3, and so on. Use CPA1 as the source pod to save time (per section 9.4).
- 4. Test your production pods thoroughly on host A before moving on to Host B.
- 5. Create master pod MPB on host B by performing a full pod clone of master pod MPA on host A. (per section 9.6).
- 6. Create the first host B production pod CPB1 using linked clones (per section 9.3).
- 7. Create additional host B production pods CPB2, CPB3, and so on. Use CPB1 as the source pod to save time (per section 9.4).



Storage Area Network Pod Cloning Strategy. Refer to the diagram below. Only one master pod (per pod type) is required if your virtual machines are placed on a SAN. Virtual machines in the production pods will link to the virtual machines in the master pod, regardless of which ESXi host the virtual machines have been assigned to run on. This is possible because each ESXi host has access to the storage area network (where all virtual disks reside). The only difference between the production pods is the ESXi host their virtual machines will run on.



The **SAN pod cloning strategy** is played using the following moves.

- 1. Create a master pod MPX. Assign it to Host A.
- 2. Create the first host A production pod CPA1 using linked clones (per section 9.3).
- 3. Create additional host A production pods CPA2, CPA3, and so on. Use CPA1 as the source pod to save time (per section 9.4).
- 4. Create the first host B production pod CPB1 using linked clones (per section 9.3). This time, change each VM's Runtime Host to Host B.
- 5. Create additional host B production pods CPB2, CPB3, and so on. Use CPB1 as the source pod to save time (per section 9.4).



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgro

10 Virtual Machine Operations

This section describes common virtual machines operations and how they should be performed in the NETLAB+ environment.

The following table summarizes which operations can currently be performed in NETLAB+ and which operations must be performed using the vSphere client. This table may change in future versions of NETLAB+.

Operation	Perform Using	See Section
Create Virtual Machines From Scratch	vSphere Client	7.2
Take Virtual Machine Snapshots	vSphere Client	7.7
Create Port Groups for Real Equipment Pods	vSphere Client	5.4.3
Create Port Groups for NDG Virtual Machine Pods	NETLAB+ (automatic)	11
Clone Individual VMs	NETLAB+	8.4
Clone Entire Virtual Machine Pods	NETLAB+	9
Delete All Virtual Machines in a Pod	NETLAB+	10.1
Delete Individual Virtual Machines	NETLAB+	10.2
Change Virtual Machine Name	NETLAB+	10.3
Change Virtual Machine Role	NETLAB+	10.4
Migrate Virtual Machine to a Different ESXi Host	vSphere Client	10.5



10.1 Delete All Virtual Machines in a Pod

If a pod contains virtual machines that reside in the inventory, you will have the option to delete those virtual machines in one operation.

Admin					NETLAB+ 2011.R 1V.beta.23 administrator	
POD 1016 - STATUS	POD NAME		STATUS	ACTIVITY	POD TYPE	
1016 F	POD 1016	0	OFFLINE	IDLE	I SERVER	
POD 1016 - PCs AND	SERVERS (dick	the GO bu	uttons to reconfi	aure)		
GO	NAME	PC ID	STATUS	TYPE / VM	OPERATING SYSTEM	
🔍 🛄 clie	ent1	4188	ONLINE	ABSENT		
🔍 📮 clie	ent2	4189	ONLINE	ABSENT		
🔍 🗐 se	rver1	4190	ONLINE	ABSENT		
Pod 1016 Ma	Bring this po	Option : d ONLIN	5 E and make i	t available for re	servations.	
사 Test	Tell me if this	s pod is v	working prope	erly.		
Create a new pod based on the settings of this pod.					d.	
诊 Rename	name Rename this pod.					
😑 Delete	Remove this	s pod from NETLAB.				

Delete Pod Admin Logout

A You are about to delete pod 1021.

If you would also like to delete the virtual machines in this pod, please choose a deletion option. This option will apply to all virtual machines in the pod.

© Do not delete any VMs (they will remain in the NETLAB+ inventory)

- © Remove VMs from NETLAB+ inventory only (VMs remains in datacenter)
- © Remove VMs from NETLAB+ inventory and datacenter (VM files not deleted from disk)
- Remove VMs from NETLAB+ inventory, datacenter, AND delete unshared VM files from disk

Warning: deleting virtual machines from disk cannot be undone!

Cancel



	Delete Virtual Machines From					
Option	NETLAB+ Inventory	vCenter Datacenter	Disk			
Do not delete any VMs	No	No	No			
Remove VMs from NETLAB+ Inventory	Yes	No	No			
Remove VMs from NETLAB+ Inventory and datacenter	Yes	Yes	No			
Remove VMs from NETLAB+ inventory, datacenter, AND delete unshared VM filed from disk	Yes	Yes	Yes			

You may choose one of four options. The option will apply to all VMs in the pod.

Before deleting virtual machines, NETLAB+ will make sure the VMs are powered down. Any automatic networks associated with the pod will also be deleted.

The option to delete virtual machines from disk cannot be undone.

10.2 Deleting Individual Virtual Machines

Use NETLAB+ to delete a virtual machine that is registered in the NETLAB+ inventory. This will remove the virtual machine from NETLAB+ and optionally from the vCenter Datacenter and/or disk.

NETLAB+ will not allow an individual virtual machine to be deleted if any the following conditions are true.

- The virtual machine is assigned to a pod. You must first set the remote PC for which it is assigned to ABSENT to remove the association.
- A master virtual machine that is referenced by linked clones. This is to ensure that there is an ongoing record of the parent/child relationship.

Do not use the vSphere client to delete a virtual machine that is registered in the NETLAB+ inventory. This will cause the virtual machine to become orphaned.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.ne

Use the following procedure to delete an individual virtual machine.

- 1. From the administrator account, select Virtual Machine Infrastructure.
- 2. Select Virtual Machine Inventory.
- 3. Select the VM you wish to delete from the inventory.

Virtual Machine	Inventor	/			
Admin Logout					
Showing VM Em	ipty XP VI	Л 1			
Virtual Mac	hine Name	Empty XP VM 1	Rename		
Operati	ing System	Windows XP			
	Role	🗏 Normal	Change Role		
1	Datacenter	NETLAB			
Ru	ntime Host	10.0.0.38			
1	Host Group	n/a			
	CPUs	1			
Me	emory (MB)	256			
	Pod ID	(VM is not assigned to a pod)			
	Parent	(no parent)			
	Children	0			
Datacenter Uniqu	e Identifier	5027288b-fa22-ab8a-a6d0-c5209e0	cbb37		
Datacenter Storag	e Location	[datastore-37-san-3] Empty XP VM	1/Empty XP VM 1.vmx		
	Comments				
sedit 🎤	Change	virtual machine parameters.			
Clone	Clone th	is virtual machine.			
Remove	Remove this virtual machine from the NETLAB+ inventory (and optionally from the datacenter				
🔀 Exit	Return to inventory.				

- 4. Check the pod id and verify the VM is not assigned to a pod. NETLAB+ will not proceed if the VM is assigned to a pod.
- 5. Verify the number of children is 0. NETLAB+ will not proceed if this virtual machine is the parent of linked virtual machine.
- 6. Click the **Remove** button.
- 7. Select the extent of the deletion. If you choose to delete unshared VM files from disk (third option), the VM disk files are erased and cannot be restored.

Remove virtual machine 'Empty XP VM 1' ?	
Remove VM from NETLAB+ inventory only (VM remains in datacenter)	
Remove VM from NETLAB+ inventory and datacenter (VM files not deleted from disk)	
Remove VM from NETLAB+ inventory, datacenter, AND delete unshared VM files from disk	
OK 🛛 Cancel	

8. Click **OK** to proceed.



10.3 Changing the Name of a Virtual Machine

Use NETLAB+ to change the name of a virtual machine that is registered in the NETLAB+ inventory. This will ensure that the virtual machine will be named the same in both NETLAB+ and vCenter.

- 1. From the administrator account, select Virtual Machine Infrastructure.
- 2. Select Virtual Machine Inventory.
- 3. Select the VM you wish to rename from the inventory.
- 4. Click the **Rename** button.

Showing VM Empty XP VM	Л 1	
Virtual Machine Name	Empty XP VM 1	Rename
Operating System	Windows XP	
Role	P Normal	Change Role
Datacenter	NETLAB	
Runtime Host	10.0.0.38	
Host Group	n/a	
CPUs	1	
Memory (MB)	256	
Pod ID	(VM is not assigned to a pod)	
Parent	(no parent)	
Children	0	
Datacenter Unique Identifier	5027288b-fa22-ab8a-a6d0-c5209e0cbb	537
Datacenter Storage Location	[datastore-37-san-3] Empty XP VM 1/E	mpty XP VM 1.vmx
Comments		

5. Provide a new name for the virtual machine. This name must be unique in both the NETLAB+ inventory and in vCenter.



6. Click **OK** to complete the rename operation.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgi

10.4 Changing the Role of a Virtual Machine

The role of a virtual machine is specific to NETLAB+ and changed using NETLAB+.

- 1. From the administrator account, select Virtual Machine Infrastructure.
- 2. Select Virtual Machine Inventory.
- 3. Select the VM whose role you wish to change from the inventory.
- 4. Click the **Change Role** button.

Showing VM Empty XP VM	Л 1	
Virtual Machine Name	Empty XP VM 1	Rename
Operating System	Windows XP	
Role	譻 Normal	Change Role
Datacenter	NETLAB	
Runtime Host	10.0.0.38	
Host Group	n/a	
CPUs	1	
Memory (MB)	256	
Pod ID	(VM is not assigned to a pod)	
Parent	(no parent)	
Children	0	
Datacenter Unique Identifier	5027288b-fa22-ab8a-a6d0-c5209e0c	bb37
Datacenter Storage Location	[datastore-37-san-3] Empty XP VM 1/	Empty XP VM 1.vmx
Comments		

5. Select the new role of the virtual machine.

Change Virtual Machine Role					
Virtual Machine Name Current Role	Empty XP VM 1 Image: Normal				
New Role	 Persistent Baster Fomplate 				
🕝 OK 🛛 🔀 Cance	I				

If the new role is set to Template, the virtual machine will also be marked as a Template in vCenter, and any ESXi host association will be lost. If the virtual machine is assigned to a pod, NETLAB+ will not allow the role to be changed to Template.



6. If the current role is Template, you must also specify a target runtime host for the virtual machine. This is because non-template VMs are associated with a runtime host. The virtual machine will be set to run on the target host and will no longer be marked as a template in vCenter.

Change Virtual Machine Role							
Virtual Machine Name Current Role	2010120613 ICM 4.1 Template vcli	ient no-Win-inst					
New Role	🖲 💻 Normal						
	 F Persistent Master 						
Target Host	10.0.0.36						
	10.0.0.36						
	10.0.038	show help tips 📄					
🕜 OK 🛛 🛛 Cancel							

7. Click **OK** to complete the role change.

10.5 Migrating a Virtual Machine to a Different ESXi Host

Each virtual machine (other than template VMs) is assigned to a runtime ESXi host in both NETLAB+ and vCenter. The runtime host is the ESXi server where the virtual machine will execute when powered on.

Migration is the process of moving a virtual machine from one ESXi host to another. This may involve moving the virtual machine disk files from one host to another if the files are located on an ESXi host local disk.

Do not use the vSphere client to migrate a virtual machine if the virtual machine is registered in the NETLAB+ inventory. This will create a discrepancy between NETLAB+ and vCenter. You should first unregister the host from the NETLAB+ inventory before migration using vCenter.

At the current time, NETLAB+ does not provide a migration facility. If you absolutely must change the runtime host of a virtual machine, you can follow the following procedure.

- 1. Unregister the host from the NETLAB+ inventory.
 - a. If the VM is assigned to a pod, you will need to set the corresponding remote PC to ABSENT.
 - b. Select the VM from the Virtual Machine Inventory.
 - c. Click remove.
 - d. Select "Remove VM from NETLAB+ inventory only."



2. Perform the migrate operation using the vSphere client. A wizard will guide you through the process.

🖉 VCENTER - vSphere Client	
File Edit View Inventory Adm	inistration Plug-ins Help
🖸 🖸 🏠 Home 🕨 🚮 🛛	Inventory 🕨 🗊 Hosts and Cl
□ ✓ VCENTER □ NETLAB □ 10.0.0.36 □ 10.0.0.38 □ BackTrack Master □ Backtrack Pod 2 S □ BackTrack Pod 1 □ BT4-R1 □ Empty XP VM 1	standalone PC
Empty XP VM 5 Generic Pod Generic Pod	Power Guest
Generic Pod Generic Pod Generic Pod Generic Pod	Snapshot Open Console
Host A Maste	Edit Settings Migrate

- 3. Re-import the virtual machine back into NETLAB+ if desired.
- 4. Assign the virtual machine to a pod if desired.

Remember, do not perform the migrate operation on a virtual machine that is currently registered in the NETLAB+ inventory.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netd

11 Using NDG Automated Pods

NDG provides many standard topologies that support special automation for virtual machines. Pods using these topologies are created using the techniques described earlier in this guide.

The automation described in this section requires VMware vCenter and the NETLAB+ Virtual Machine Inventory features as described in this guide.

11.1 NDG Virtual Machine Topologies

NDG standard topologies that contain only virtual machines can be replicated and deployed very quickly using a combination of pod cloning, automatic networking and automatic remote display setup features.

The list of NDG automated virtual machine topologies includes:

- 25 automated client/server pods for General IT usage. For details see http://www.netdevgroup.com/content/generalit
- An automated topology for CNSS 4011 cyber security. For details see http://www.netdevgroup.com/content/cybersecurity/topologies
- An automated topology for the VMware IT Academy Install, Configure, Manage course. This pod is available to member organizations of the VMware IT Academy. For details see: <u>http://www.netdevgroup.com/content/vmita</u> and the pod specific guide in the NDG Lab Resource Center for VMware IT Academy.

Once production pods are created using the pod cloning utility, they are ready to go. When an automated pod is scheduled and lab reservation begins, the following automation takes place:

- NETLAB+ will automatically create virtual switches and port groups required for the pod.
- NETLAB+ will bind the virtual network adapters of each VM to the correct port group.
- NETLAB+ will configure remote display parameters automatically.

When the lab reservation is over, NETLAB+ will tear down the virtual switches and port groups created for the pod to free resources on the host.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.net

11.2 NDG Real Equipment Topologies

As of NETLAB+ version 2011.R2, automatic networking and automatic remote display setup is supported on the following topologies:

- Multi-purpose Academy Pod (MAP)
- Cuatro Router Pod (CRP)
- Cuatro Switch Pod (CSP)
- LAN Switching Pod (LSP)
- Network Security Pod (NSP)
- Basic Router Pod Version 2 (BRPv2)
- Basic Switch Pod Version 2 (BSPv2)

When a pod is scheduled and lab reservation begins, the following automation takes place:

- NETLAB+ will automatically calculate the control switch VLANs that are required for the pod.
- NETLAB+ will automatically create VLAN based port groups on the virtual machine host's inside virtual switch.
- NETLAB+ will bind the virtual network adapters of each VM to the correct port group.
- NETLAB+ will configure remote display parameters automatically.

When the lab reservation is over, NETLAB+ will tear down the port groups created for the pod to free resources on the host.

For automatic networking on real equipment pods, you must setup the inside virtual switch on each ESXi host (section 5.4). You must also provide the name of the virtual switch in the NETLAB+ virtual machine host setup (section 5.6). Automatic networking will not occur if these tasks are not performed.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgrou

11.3 Setting the Local System ID When Using Multiple NETLAB+ Systems

If you have more than one NETLAB+ system, particularly if they access a common VMware vCenter and/or ESXi host systems, it is necessary to set the System Local ID of your NETLAB+ systems so that each NETLAB+ system is uniquely identified. This is necessary in order to support pod automation as described in the previous section. If you only have one NETLAB+ system, using default value '001' is sufficient.

System Wide Virtualization Settings are accessed from the Virtual Machine Infrastructure administrator option.

System-wide Virtualization Settings						
Admin Logout						
View and update system-wide virtualization settings.						
System Local DO1 Show help tips						
OK Cancel						



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netde

Appendix A Manual Network Setup and Troubleshooting

Since NETLAB+ version 2011.R2, automatic networking is supported on most pod types supplied by NDG. This appendix provides manual network setup and troubleshooting guidance. Manual network setup is only required when:

- You are working with a custom pod.
- You are using an older NDG pod design that does not support automatic networking.
- You have disabled automatic networking features on a pod and/or remote PC so that you can have manual control over network creation and binding.

Appendix A.1 Creating Inside VLANs that Connect to Real Equipment

This section discusses the configuration of networks to communicate between virtual machines and real lab devices in the topology. You can skip this section you are using an NDG pod design that supports automatic networking.



Refer to the numbers in the diagram below in the discussion that follows.

Virtual machines [7] running on an ESXi host talk to real equipment pods [12] using port groups [6], the inside virtual switch [3] and control switches [9]. Inside networking is always used for this purpose. In the last two sections (5.4.1 and 5.4.2), you established an inside network connection and configured 802.1q VLAN trunking on the link connecting to your control switch [8].



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.cd

A real equipment pod may have one or more networks. 802.1q virtual LANs (VLANs) are the glue that binds virtual machines [7] and real equipment [12] with the proper pod networks. A unique set of VLAN identifiers for each pod is automatically allocated by NETLAB+ and programmed into the control switches [9] by NETLAB+ when the pod is created. Port groups [6] are assigned to a specific VLAN ID, thereby allowing virtual machine network adapters [5] to be placed in a specific VLAN. Determining the correct VLAN numbers to use is explained in the next section.

It is advantageous to create port groups [6] before creating virtual machines [7]. By doing so, you will be able to bind the virtual machine to the proper port group/VLAN during virtual machine creation. Alternatively, you may place your virtual machine on a safe staging network (see section 5.5) and create port groups at a later time.

The following subsections describe this process, using the Multi-purpose Academy Pod (MAP) as an example. The MAP pod type was chosen because it supports over 200 Cisco Networking Academy labs and is the most widely deployed pod type used today.

The following sections assume you have successfully established and tested inside networking as described in the previous sections.

Appendix A.1.1 Creating a Real Equipment Pod

Creating a real equipment pod in NETLAB+ should be done first. This will automatically generate the required number of VLANs for the selected pod type, and add those VLANs to the control switches. This should be done before ESXi host networking is configured.

- 6. Login to the NETLAB+ administrator account.
- 7. Select Equipment Pods.
- 8. Click the **Add a Pod** button at the bottom of the page.
- 9. Select the desired pod type. Only pod types that use both real lab equipment and remote PCs are relevant to this section. Our examples use the Multipurpose Academy Pod, which contains 3 routers, 3 switches, and 3 remote PCs.
- 10. Complete the **New Pod Wizard.** Please refer to the <u>NETLAB+ Administrator</u> <u>Guide</u>, NDG website pod specific web pages, and NDG pod guides for pod specific installation instructions.

After the pod is created, you will be placed in the Pod Management page. You will notice that all virtual machines are initially ABSENT. You will add virtual machines to the pod later.



Appendix A.1.2 Determining the Base VLAN and VLAN Pool

A unique set of VLAN identifiers for each pod is automatically allocated by NETLAB+ and programmed into the control switches by NETLAB+ when the pod is created. To place virtual machines in the correct VLAN, you must know the range of VLANs used by the pod. This is shown on the Pod Management page control switch table. In the example below, the VLAN range (pool) is 110 to 117. The first (base) VLAN is 110. Each pod containing real equipment will have a unique base VLAN and pool. The size of the pool depends on the pod type.

Pod Management NETLAB+ 2011.R1V.beta.21 Admin administrator													
POD 2 - STATUS													
POD ID	POD N	IAME	TATUS	S ACTIVITY		IVITY	POD TYPE		TYPE				
2	PO	D 2 OFFLINE			INE	IDLE		MULTI-PURPOSE ACADEMY POD 3 Routers, 3 Switches					
POD 2 - ROUTERS, SWITCHES, AND FIREWALLS (click on the GO buttons to reconfigure devices)													
GO	NAME	TYPE					CCESS NES	SWITCHED OUTLETS		SOFTWARE IMAGE			
A	🔇 R1	Cisco 28	01/2811	(S0/1/)	()	AS 3 LI	NE 1	1 SOD 3 OUTLET 1		n/a			
🔍 🜔	R2	Cisco 18	41 (S0/1/	X)		AS 3 LI	NE 2	SOD 3 OUTLET 2		n/a			
🔍 🜔	R3	Cisco 28	01/2811 ((S0/1/x	()	<mark>AS 3</mark> LI	NE 3	SOD 3 OUTLET 3		n/a			
🔍 🗧	📫 S1	Cisco 35	50-24 EM	I		<mark>AS 3</mark> LI	NE 4	SOD 3 OUTLET 4		n/a			
🔍 🗄	S 2	Cisco 29	50T-24 (E	EI)		<mark>AS 3</mark> LI	NE 5	SOD 3 OUTLET 5		n/a			
a	📫 S3	Cisco 35	50-24 EM	I		<mark>AS 3</mark> LI	NE 6	6 SOD 3 OUTLET 6		n/a			
	AND SER	VERS (lick the G	C hut	ons to re	configu	re)						
GO		NAME	P	CID	D STATUS		TYPE / VM		OPERATING SYSTEM				
	Р	PC A 4173 ONL		INE	ABSENT								
٩,	Р	рс в		174	ONLINE		ABSENT						
Q	🔍 🛄 PC C		4	4175 ONLI		INE	ABSENT						
POD 2 - CONTROL SWITCH													
SWITCH ID	NGE	BASE VLAN VLAN POO			N POOL								
⊇	1-8			1	110	11	0-117						

The NDG pod specific documentation and pod specific web pages for Cisco Netacad content show the *VLAN offset* for each virtual machine. To determine the actual VLAN IDs that will be used by virtual machines, simply add the VLAN offsets to the base VLAN for the pod you are configuring.

The Multi-purpose Academy Pod now supports automatic networking. These calculations and configuration tasks described here are for example purposes.


The following table shows the VLAN offsets for a Multipurpose Academy Pod (source: NDG website).

Virtual Machine	Recommended O/S	Functions	VLAN Offset ¹
PC A	Windows XP	Student PC, client activities	+0
PC B	Windows XP	Student PC, client activities	+1
PC C	Windows XP	Student PC, client activities	+3

To determine the actual VLAN ID for each virtual machine, add the VLAN offset of each PC to the base VLAN of the pod.

Virtual Machine / Remote PC	Example Pod Base VLAN		VLAN Offset From Table Above		Actual VLAN ID
PC A	110	+	0	=	110
PC B	110	+	1	=	111
PC C	110	+	3	=	113

You will notice that only 3 of 8 VLANs in the MAP pod's VLAN pool are used by virtual machines. The other 5 are used for communication between real equipment.

When working with VLANs for remote PCs, it is important to remember:

- A set of unique VLAN IDs is created by NETLAB+ if and only if the pod type supports real lab equipment and connects to a control switch. If the pod type only contains virtual machines, no VLANs are generated for the pod. VLANs on the control switches are automatically created and maintained as needed by NETLAB+; do not manage the switch VLAN database manually.
- Each pod type may use different VLAN offsets for PCs. Use the NDG pod specific documentation or web pages to determine the VLAN offsets for the specific type of pod type you are deploying.
- The actual VLAN ID for a remote PC is computed by adding the VLAN offset for the pod type, to the base VLAN of the actual pod.
- Since each real equipment pod will always have a unique base VLAN, the remote PCs in each real equipment pod will always have unique VLAN IDs as well.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

Appendix A.1.3 Creating Port Groups for Pod VLANs on the Inside Network

Once you have determined the actual VLAN numbers used by real equipment pods for remote PCs, you can create the corresponding port groups on the inside vSwitch. In our MAP pod example we determined that VLAN 110, 111, and 113 were used for the 3 remote PCs. Therefore, 3 port groups will be created and tagged with VLAN 110, 111, and 113 respectively. Of the 8 VLANs reserved for the pod, only 3 VLANs will carry virtual machine traffic. Therefore, only 3 port groups are created.

Virtual machines are assigned to run a specific ESXi host. Therefore, it is not necessary to create the same inside port groups on every ESXi host. For example, if our MAP Pod with pod ID 2 has virtual machines that are assigned to ESXi Host A, you only need to create port groups for VLANs 110, 111, and 133 on Host A. You do not need to create port group VLANs 110, 111, and 133 on ESXi Host B because those VLANs are exclusive to MAP Pod 2 on host A.

The following procedure is used to create a single port group for a pod VLAN on the inside vSwitch. The process is repeated for each pod VLAN, but only on the ESXi host where the virtual machines will run.

- 1. Login to vCenter using the vSphere client.
- 2. Navigate to Home > Inventory > Hosts and Clusters.
- 3. Click on the ESXi host where the pod's virtual machines will run.
- 4. Click on the **Configuration tab**.
- 5. Click on **Networking** in the Hardware group box.
- 6. Click on the **Virtual Switch** view button if not already selected.
- Click Properties on the INSIDE vSwitch. The inside vSwitch is the one that is connected to the control switch (typically vSwitch0 if single homed, vSwitch1 if dual homed).





8. At the vSwitch properties page will appear, click the **Add** button.

🕗 vSwitch Properties					
Ports Network Adapters					
Configuration	Summary	vSwitch Properties			
1 vSwitch	56 Ports	Number of Ports:	56		
Management Net	VMotion and IP				
		Default Policies			
		Security			
		Promiscuous Mode:	Reject		
		MAC Address Changes:	Accept		
		Forged Transmits:	Accept		
		Traffic Shaping			
		Average Bandwidth:			
		Peak Bandwidth:			
		Burst Size:			
		Failover and Load Balancing			
		Load Balancing:	Port ID		
		Network Failure Detection:	Link Status only		
		Notify Switches:	Yes		
		Failback:	Yes		
		Active Adapters:	vmnic0		
		Standby Adapters:	None		
,	544	Unused Adapters:	None		
Add	Ealt Remove				
			Close Help		

9. Select the Virtual Machine connection type, and then click Next.

Add Network Wizard	
Connection Type Networking hardware of	an be partitioned to accommodate each service that requires connectivity.
Connection Type Network Access	Connection Types
Connection Settings Summary	Virtual Machine Add a labeled network to handle virtual machine network traffic.
	C VMkernel The VMkernel TCP/IP stack handles traffic for the following ESXi services: VMware VMotion, iSCSI, NFS, and host management.
Help	≤ Back Next ≥ Cancel



- 10. Type in a network label that identifies the pod and VLAN ID.For example, "POD 2 VLAN 110".This label will help you identify the port group's pod and VLAN.The VLAN ID is based on the calculations above.
- 11. Type in the VLAN ID that will be associated with this port group. The VLAN ID is based on the calculations above.
- 12. Click **Next** to complete the wizard.

🕜 Add Network Wizard		
Virtual Machines - Conne Use network labels to ide	ction Settings entify migration compatible connections common to two or more hosts.	
<u>Connection Type</u> Connection Settings Summary	Port Group Properties Network Label: VLAN ID (Optional): EXAMPLE ONLY: Your Values will Vary	
	Preview:	
	Virtual Machine Port Group POD 2 VLAN 110 VLAN ID: 110 VMkernel Port VMkernel vmk1: 169.254.0.241	
Help	≤ Back Ne	ext ≥ Cancel

13. Click the **Finish** button on the confirmation page.

The user interface may become unresponsive for a minute during the port group creation process and appear to "lock up". This behavior will eventually cease once the port group is created.

14. Confirm that new port group should appear in the vSwitch properties.

15. Click **Close** to complete the task.

The process for binding virtual machines network adapters to the port group will be covered later in this document.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.c

This section provides guidance on common troubleshooting issues associated with the implementation of ESXi versions 4.01 and 4.1 with vCenter with NETLAB+ and guidance on verifying connectivity after installation. Please review the material in this section prior to contacting NDG for customer support.

Objectives

- Verifying connectivity between virtual machines and lab gear.
- Reviewing and/or modifying virtual machine settings for existing virtual machines.
- Identify and resolve the most frequently encountered issues.

Appendix A.2 Verifying Connectivity Between Virtual Machines and Lab Gear

We strongly encourage verifying the connectivity between your virtual machines using the method described in this section. The troubleshooting methods shown here can also aid you in determining why a remote PC in a NETLAB+ pod is having network connectivity problems.

Verify that your pod is online (see the *Equipment Pods* section of the <u>NETLAB+</u> <u>Administrator Guide</u>) and that the pod passes the pod test (see the *Test the Pod* section).

The example below illustrates a NETLAB_{AE} BRPv2 topology installed as Pod #5 on Control Switch #4:

BRPv2 Lab Device	Device Port	Control Switch #4 Port	NETLAB+ Pod VLAN
Router 1	fa 0/0	fa 0/1	140
	fa 0/1	fa 0/2	141
PC1a	virtual NIC	fa 0/23	140
PC1b	virtual NIC	fa 0/23	140
Router 2	fa 0/0	fa 0/3	142
	fa 0/1	fa 0/4	143
PC2	virtual NIC	fa 0/23	142
Router 3	fa 0/0	fa 0/5	144
	fa 0/1	fa 0/6	145
PC3	virtual NIC	fa 0/23	144

In order to test the connectivity between remote PCs and neighboring lab devices, using the above example, you may follow these steps, using an Instructor Account (see the *Manage Accounts* section of the <u>NETLAB+ Administrator Guide</u>).



- 1. Make a lab reservation.
- 2. Configure IP addresses on the remote PCs and neighboring lab devices you will be testing.
- 3. In the example above, PC1a and PC1b should share the same VLAN adapter, so they should be able to ping each other. If they cannot ping each other, then you should review the following:
 - What VLAN adapter are PC1a and PC1b using? (refer to section 5.4.3. Is there a firewall installed or enabled on the virtual machine?
- 4. To verify the connectivity between remote PCs and neighboring lab devices, you should test the following:
 - Ping from PC1a to R1 and vice versa.
 - Ping from PC1b to R1 and vice versa.
 - Ping from PC2 to R2 and vice versa.
 - Ping from PC3 to R3 and vice versa.
- 5. If you can ping from a remote PC to a neighboring lab device, but cannot ping from the lab device to the remote PC, then you may want to determine if there is a firewall installed or enabled on the virtual machine.
- 6. If any of the tests from step 4 completely fail (you cannot ping from remote PC to neighboring lab device and vice versa), then you will need to analyze the network traffic on the control switch. Using the above example, perform the following steps:
 - Connect a PC or terminal to the console port of the control switch.
 - Type **"show vlan"** or **"show vlan brief"** to view the VLAN status on the control switch.

The control switch console password is **router**. The enable secret password is **cisco**. These passwords are used by NETLAB+ automation and technical support - please do not change them.



Remote PC Guide for VMware Implementation Using ESXi versions 4.01 and 4.1 U2 with vCenter www.netdevgroup.com

Conne Escap	Connected to 169.254.1.14. Escape character is '^]'.					
User	Access Verification					
Passu	vord:					
netla	ab-cs4>en					
Passu	vord:					
netla	ab-cs4#show vlan					
VLAN	Name	Status	Ports			
1	default	active	Fa0/14,	Fa0/15,	FaO,	
			Fa0/18,	Fa0/19,	Fa0,	
			Fa0/22,	Fa0/24,	GiO,	
3	NETLAB_3	active				
11	NETLAB_11	active				
12	NETLAB_12	active				
13	NETLAB_13	active	- • •			
140	NETLAB_140	active	FaO/1			
141	NETLAB_141	active	Fa0/2			
142	NETLAB_142	active	FaO/3			
143	NETLAB_143	active	FaO/4			
144	NETLAB_144	active	FaO/5			
145	NETLAB_145	active	FaO/6			

During a lab reservation, you will notice the active lab ports and their VLAN assignments. From the example above, Pod #5 is a BRPv2 installed on ports fa0/1 through fa0/6 on Control Switch #4. The base VLAN for this pod is 140.

• On the control switch, type **"show interfaces trunk"** to view the trunk information.

netlab-cs4#show interfaces trunk						
Port Fa0/23	Mode on	Encapsulation 802.1q	Status trunking	Native vlan 1		
Port Fa0/23	Vlans allo 140,142,	wed on trunk 144				
Port Fa0/23	Vlans al. 140,142,	lowed and active ir 144	n management	domain		
Port FaO/23	Vlans in 140,142,	spanning tree form 144	varding state	e and not pruned		



This command will reveal whether or not you have properly configured the control switch port that connects to the VMware trunking port. The following shows the proper configuration for the example above on port 23 of Control Switch #4.

On the control switch, type "show mac-address-table dynamic". Use the MAC address table to verify: 1) whether the MAC addresses of the remote PCs are in the table and 2) if these MAC addresses are in the correct VLANs.

netlab-cs4#show mac-address-table dynamic Mac Address Table				
Vlan	Mac Address	Туре	Ports	
140 140	0000.0c5d.150e	DYNAMIC	Fa0/1 Fa0/23	
140	000c.292f.57f2	DYNAMIC	Fa0/23	
142	0010.7b81.aae0 0000.0c76.bd12	DYNAMIC	Fa0/3 Fa0/5	
144	000c.29c1.1bc7 000d.60f3.1757	DYNAMIC	Fa0/23 Fa0/24	
1	0050.5000.1109	DYNAMIC	Fa0/24 Fa0/24	
1 Total	00c0.b7a3.1def Mac Addresses for	DYNAMIC this criteri	Fa0/24	

- 7. If any of the tests from step 4 completely fail (you cannot ping from the remote PC to a neighboring lab device and vice versa), and the MAC address of a remote PC is either:
 - a. Not in the correct VLAN or
 - b. Does not show up in the control switch MAC address table, then please review the VLAN and settings for your NETLAB+ pod very carefully.

Possible error conditions include:

- An incorrect VLAN ID was entered when creating a VLAN interface.
- No VLAN or an incorrect VLAN was mapped using VI Client
- The control switch port (for the Inside Connection) is not trunking or not allowing the correct VLANs.